

LSHTM Research Online

Lovell, GA; (1982) Studies on the cuticle of the human body louse, Pediculus humanus humanus. PhD thesis, London School of Hygiene & Tropical Medicine. DOI: https://doi.org/10.17037/PUBS.04655860

Downloaded from: https://researchonline.lshtm.ac.uk/id/eprint/4655860/

DOI: https://doi.org/10.17037/PUBS.04655860

Usage Guidelines:

Please refer to usage guidelines at https://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license. To note, 3rd party material is not necessarily covered under this license: http://creativecommons.org/licenses/by-nc-nd/3.0/

STUDIES ON THE CUTICLE OF THE HUMAN BODY LOUSE, PEDICULUS HUMANUS HUMANUS.

1

BY

GEOFFREY ALAN LOVELL. B. App. Sci. (Q.I.T.).

A THESIS SUBMITTED TO THE FACULTY OF SCIENCE OF THE UNIVERSITY OF LONDON FOR THE DEGREE, DOCTOR OF PHILOSOPHY.

DEPARTMENT OF ENTOMOLOGY LONDON SCHOOL OF HYGIENE AND TROPICAL MEDICINE.

JANUARY 1982.



NUMEROUS ORIGINALS IN COLOUR



BLANK IN ORIGINAL

Abstract.

The cuticular ultrastructure and surface topography of the human body louse, <u>Pediculus humanus humanus</u>, were examined. Under light microscopy, pore canals were readily observed in unstained, cryostat sections of cuticle from various locations on the imsect, particularly those areas composed of smooth, sclerotised cuticle. Transmission electron microscopy (TEM) studies revealed that, after attachment at the epidermis, the canals appeared to pass unbranched through the endocuticle (average thickness 4.5 μ) with dismeters around 1000Å. Branching occurred in the exocuticle (thickness 1.0 μ) and finally in the epicuticle (thickness 2000Å) to form war canals with diameters around 100Å. Pores or openings with diameters around 100Å were found during scanning electron microscopic (SEM) studies of the epicuticular surface and corresponded with the war canals mored in the TEM examinations.

A chemical analysis of the cuticular lipids of the louse was conducted using thin-layer chromatography (TLC), gas-liquid chromatography (GLC) and gaschromatography/mass spectroscopy (GC/MS) and the hydrocarbons were found to be the most abundant class (almost 75 per cent). Cholesteryl esters, triglycerides, 1,3-diglycerides, cholesterol and fatty acids were also present in that order of abundance. Cuticular hydrocarbons were comprised of over 43 components which contained from 21 to 35 total carbons per molecule. Four homologoum series were present and consisted of n-alkanes (51.8 per cent of total cuticular hydrocarbons), n-alkanes (26.4 per cent), internally branched monomethylalkanes (12.1 per cent) and a tentatively identified series comprising dimgthylakanes (4.7 per cent).

The effect of insecticide solvents upon the cuticular vax layer was investigated in vitro by comparing the solubility of a simulated cuticular wax mixture in a series of ten potential solvents. Results are discussed along with those from toxicity tests performed with the solvents on adult female lice and indicate a relationship between cuticular wax solubility and louse mortality. Studies with

a fluorescent dye, demonstrated that wax-dissolving solvents entered the insect via at least two routes, i.e. cuticle and spiracles.

The results are compared with the published work on other insect cuticular systems and the implications involving potential insecticide carriers discussed. List of Contents.

Page.

Abstract.	3
List of Contents.	5
List of Tables.	11
List of Figures.	12
List of Plates.	15
1. INTRODUCTION AND LITERATURE REVIEW.	18
1.1. Structure of arthropod cuticle.	18
1.2. Cuticular lipids of arthropods.	23
1.3. The penetration of arthropod cuticle.	34
1.4. Aims of work.	42
2. MATERIALS AND METHODS.	43
2.1. Examination of the structure of body louse cuticle.	43
2.1.1. Light microscope (LM) examination.	43
2.1.1.1. Examination of unstained cuticle sections.	43
2.1.1.2. Examination of stained cuticle sections.	43
2.1.2. Transmission electron microscope (TEM) examination.	44
2.1.2.1. Examination by TEM of cuticle sections.	44
2.1.2.2. Platinum/carbon (Pt/C) replication of	
cuticular surface and cross section.	45
2.1.3. Scanning electron microscope (SEM) examination.	46
2.1.3.1. Morphology and surface topography of the adult	
louse.	46
2.1.3.2. Examination of cuticular cross section.	46
2.1.3.3. Solvent effects on the cuticular wax layer.	47
2.1.3.4. Effects of alkali treatment on the cuticle.	47
2.2. Chemical composition of louse lipids.	48
2.2.1. Comparison of potential extraction solvents for	
cuticular lipida.	48
2.2.1.1. Trial thin-layer chromatography (TLC)	
separations.	48
2.2.1.2. Solubility of long chain hydrocarbon (n-C 36)	
in potential extraction solvents.	49
2.2.2. Extraction of louse lipids.	49
2.2.2.1. Cuticular lipid extraction using whole lice.	49

2.2.2.2. Cu	ticular lipid extraction using cast skins.	50
2.2.2.3. Ex	amination of cuticular lipid extraction	
te	chnique (using whole lice).	52
2.2.2.4. Ex	traction of internal lipids from whole	
ad	ult lice.	54
2.2.2.5. Ex	traction of lipids from louse faeces.	55
2.2.3. Thin-1	syer chrometography (TLC) analysis of lipid	
extrac	ta.	55
2.2.3.1. Qu	alitative TLC analysis of cuticular lipids/	
in	ternal lipids/faecal lipids.	55
2.2.3.2. Qu	antitative TLC analysis (by weight) of	
cu	ticular lipids.	57
2.2.3.3. Qu	antitative TLC analysis (by densitometer)	
of	cuticular lipids/internal lipids/fsecal	
11	pida.	58
2.2.3.4. TL	C analysis of digestion products of labelled	
fa	tty acid (oleic acid).	58
2.2.3.5. Te	st for saturation of cuticular hydrocarbons.	59
2.2.4. Gas-1i	quid chromatographic GLC) analysis of	
lipid	extracta,	59
2.2.4.1. Mo	dification of Perkin-Elmer Fll gas	
ch	romatograph (GC) for capillary column use.	59
2.2.4.2. In	itial GLC separation of cuticular	
hy	drocarbons.	63
2.2.4.3. GL	C examination for saturation of cuticular	
hy	drocarbons.	64
2.2.4.4. GL	C examination for branched/unbranched	
cu	ticular hydrocarbons.	65
2.2.4.5. Co	mparison of unknown GLC peaks with standard	
GL	C peaks.	65
2.2.4.6. Po	sition of double bond in unsaturated	
cu	ticular hydrocarbons.	65
2.2.4.7. Ga	s chromatography/Mass spectroscopy (GC/MS)	
an	alysis of cuticular hydrocarbons.	66
2.2.4.8. GL	C analysis of internal hydrocarbons/faecal	
hy	drocarbons.	67

Page,

2.2.2.2. Cuticular lipid extraction using cast s	kins. 50
2.2.2.3. Examination of cuticular lipid extracti	on
technique (using whole lice).	52
2.2.2.4. Extraction of internal lipids from whol	e
adult lice.	54
2.2.2.5. Extraction of lipids from louse faeces.	55
2.2.3. Thin-layer chromatography (TLC) analysis of	lipid
extracts.	55
2.2.3.1. Qualitative TLC analysis of cuticular 1	ípids/
internal lipids/faecal lipids.	55
2.2.3.2. Quantitative TLC analysis (by weight) of	f
cuticular lipida.	57
2.2.3.3. Quantitative TLC analysis (by densitome	ter)
of cuticular lipids/internal lipids/fae	cal
lipids.	58
2.2.3.4. TLC analysis of digestion products of 1	abelled
fatty acid (oleic acid).	58
2.2.3.5. Test for saturation of cuticular hydroc	arbona. 59
2.2.4. Gas-liquid chromatographic GLC) analysis of	
lipid extracts.	59
2.2.4.1. Modification of Perkin-Elmer F11 gas	
chromatograph (GC) for capillary column	use. 59
2.2.4.2. Initial GLC separation of cuticular	
hydrocarbons.	63
2.2.4.3. GLC examination for saturation of cutic	ular
hydrocarbons.	64
2.2.4.4. GLC examination for branched/unbranched	
cuticular hydrocarbons.	65
2.2.4.5. Comparison of unknown GLC peaks with st	andard
GLC peaks.	65
2.2.4.6. Position of double bond in unsaturated	
cuticular hydrocarbons.	65
2.2.4.7. Gas chromatography/Mass spectroscopy (G	C/MS)
analysis of cuticular hydrocarbons.	66
2.2.4.8. GLC analysis of internal hydrocarbons/f	aecal
hydrocarbons,	67

2.2.4.9. GLC analysis of cuticular cholesteryl esters.	68
2.2.4.10. GLC analysis of cuticular triglycerides.	69
2.2.4.11. GLC analysis of cuticular fatty acids.	70
2.2.4.12. GLC enalysis of cuticular 1,3-diglycerides.	70
2.2.5. Synthesis of major unsaturated cuticular	
hydrocarbon (15-hentriacontene)	71
2.2.5.1. Preparation of alkyl halide (II) -	
intermediate.	73
2.2.5.2. Preparation of aldehyde (IV) - intermediate.	73
2.2.5.3. Preparation of Grignard Reagent (V) -	
intermediate.	74
2.2.5.4. Formation of secondary alcohol (VII) -	
intermediate.	75
2.2.5.5. Formation of alkene (VIII) from alcohol (VII),	75
.3. Applied aspects.	77
2.3.1. Maintainance of lice colonies.	77
2.3.2. Physical characteristics of cuticular lipids and	
edult lice.	81
2.3.2.1. Melting-point determination of cuticular wax	81
2.3.2.2. Solubility of simulated cuticular wax in	
potential insecticide solvents.	81
2.3.2.3. Determination of upper lethal temperature of	
adult lice.	85
2.3.2.4. Comparative volatilities of potential	
insecticide solvents.	86
2.3.3. Toxicity tests with adult lice.	87
2.3.3.1. Preparation of test material.	87
2.3.3.2. Protocol for toxicity tests.	88
2.3.3.3. Topical application of potential insecticide	
solvents.	94
2.3.3.4. Topical application of aqueous insecticide	
solvent mixture.	94
2.3.3.5. Topical application of insecticide (dieldrin)	
in potential insecticide solvent.	96
2.3.4. Observation of cuticular penetration by potential	
insecticide solvents using fluorescent dye,	
Rhodamine B.	98

3,	RESULTS.	100
	3.1. Examination of the structure of body louse cuticle.	100
	3.1.1. Light microscope (LM) examination.	100
	3.1.1.1. Examination of unstained cuticle sections.	100
	3.1.1.2. Examination of stained cuticle sections.	105
	3.1.2. Transmission electron microscope (TEM) examination.	112
	3.1.2.1. Examination by TEM of cuticle sections.	112
	3.1.2.2. Platinum/carbon (Pt/C) replication of	
	cuticular surface and cross section.	126
	3.1.3. Scanning electron microscope (SEM) examination.	131
	3.1.3.1. Morphology and surface topography of the	
	adult louge.	131
	3.1.3.2. Examination of cuticle cross section,	145
	3.1.3.3. Solvent effects on the cuticular wax layer.	153
	3.1.3.4. Effects of alkali treatment on the surface.	158
	3.2. Chemical composition of louse lipids.	161
	3.2.1. Comparison of potential extraction solvents.	161
	3.2.1.1. Trial thin-layer chromatography (TLC)	
	separations.	161
	3.2.1.2. Solubility of long chain hydrocarbon $(n-C_{36})$	
	in potential extraction solvents.	161
	3.2.2. Extraction of louse lipids.	162
	3.2.2.1. Cuticular lipid yields from extractions of	
	whole lice.	162
	3.2.2.2. Examination of cuticular lipid extraction	
	technique (from whole lice).	165
	3.2.3. Thin-layer chromatographic (TLC) analysis of	
	lipid extracts.	167
	3.2.3.1. Qualizative TLC analysis and comparison of	
	cuticular lipids from whole lice and cast	
	skins.	167
	3.2.3.2. Quantitative TLC analysis (by weight) of	
	cuticular lipids.	169
	3.2.3.3. Quantitative TLC analysis (by densitometer)	
	of cuticular lipids/internal lipids/fsecal	
	lipids.	171

3.2.3.4. TLC enalysis of digestion products of labelled	
fatty acid (oleic scid).	175
3.2.3.5. Degree of saturation of cuticular	
hydrocarbona.	178
3.2.4. Gas-liquid chromatographic (GLC) analysis of	
lipid extracts.	179
3.2.4.1. GLC analysis of cuticular hydrocarbons.	179
3.2.4.2. Gas chromatography/Mass spectroscopy (GC/MS)	
analysis of cuticular hydrocarbons.	184
3,2.4,3. GLC analysis of internal hydrocarbons/faecal	
hydrocarbons .	199
3.2.4.4. GLC analysis of cuticular cholesteryl esters.	202
3.2.4.5. GLC analysis of cuticular triglycarides.	204
3.2.4.6. GLC analysis of cuticular fatty acids.	204
3.2.4.7. GLC analysis of cuticular 1,3-diglycerides.	204
3.2.5. Examination of synthesis product	
(15-hentriacontene).	208
3.3. Applied aspects.	211
3.3.1. Physical characteristics of cuticular lipids and	
adult lice.	211
3.3.1.1. Melting-point determination of cuticular wax.	211
3.3.1.2. Solubility of simulated cuticular wax in	
potential insecticide solvents.	212
3.3.1.3. Determination of upper lethal temperature of	
adult lice.	212
3.3.1.4. Comparative volatilities of potential	
insecticide solvents.	216
3.3.2. Toxicity tests with adult lice.	218
3.3.2.1. Protocol for toxicity tests.	218
3.3.2.2. Topical application of potential insecticide	
solvents.	222
3.3.2.3. Topical application of aqueous insecticide	
solvent mixture.	236
3.3.2.4. Topical application of insecticide (dieldrin)	
in potential insecticide solvent.	239
3.3.3. Observation of cuticular penetration of potential	
insecticide solvents using fluorescent dye,	

3.2.3.4. TLC analysis of digestion products of labelled	
fatty acid (oleic acid).	175
3.2.3.5. Degree of saturation of cuticular	
hydrocarbons.	178
3.2.4. Gas-liquid chromatographic (GLC) analysis of	
lípid extracts.	179
3.2.4.1. GLC analysis of cuticular hydrocarbons.	179
3.2.4.2. Gas chromatography/Mass spectroscopy (GC/MS)	
analysis of cuticular hydrocarbons.	184
3.2.4.3. GLC analysis of internal hydrocarbons/faecal	
hydrocarbons.	199
3.2.4.4. GLC analysis of cuticular cholesteryl esters.	202
3.2.4.5. GLC analysis of cuticular triglycerides.	204
3.2.4.6. GLC analysis of cuticular fatty acida.	204
3.2.4.7. GLC analysis of cuticular 1,3-diglycerides.	204
3.2.5. Examination of synthesis product	
(15-hentriscontene).	208
3.3. Applied aspects.	211
3.3.1. Physical characteristics of cuticular lipids and	
adult lice.	211
3.3.1.1. Melting-point determination of cuticular wax.	211
3.3.1.2. Solubility of simulated cuticular wax in	
potential insecticide solvents.	212
3.3.1.3. Determination of upper lethal temperature of	
adult lice.	212
3.3.1.4. Comparative volatilities of potential	
insecticide solvents.	216
3.3.2. Toxicity tests with adult lice.	218
3.3.2.1. Protocol for toxicity tests.	218
3.3.2.2. Topical application of potential insecticide	
solvents.	222
3.3.2.3. Topical application of aqueous insecticide	
solvent mixture.	236
3.3.2.4. Topical application of insecticide (dieldrin)	
in potential insecticide solvent.	2 39
3.3.3. Observation of cuticular penetration of potential	
insecticide solvents using fluorescent dye,	
Rhodemine B.	241

4. D	DISCUSSION.	253
4	.1. Physical structure.	253
	4.1.1. External features.	253
	4,1.2. The epicuticle.	255
	4.1.3. The exocuticle.	259
	4.1.4. The endocuticle.	260
	4.1.5. The epidermis.	264
	4.1.6. The pore canal system.	265
4	.2. Cuticular lipids of the body louse.	268
4	.3. Applied aspects.	278
	4.3.1. Cuticular permeability/transition temperature.	278
	4.3.2. Toxicity test procedure.	280
	4.3.3. Toxicity of potential insecticide solvents.	281
	4.3.4. Tracer studies with solutions of Rhodamine B.	283
Ackno	owledgements.	286
List	of References.	287
Apper	ndix.	305

Page.

List of Tables.

lable	- L.	Composition of synthetic cuticular wax.
Table	2.	Potential insecticide solvents.
Table	3.	Dosage levels of 10% aqueous lauryl ether mixture.
Table	4.	Dosage levels of dieldrin in 1,4-butanedol and
		lauryl ether.
Table	5.	Comparison of fresh and dried adult insect weights.
Table	6.	Comparison of cuticular wax yields.
Table	7.	Scintillation counts on labelled/unlabelled samples.
Table	8.	Composition of cuticular louse lipids (by weight).
Table	9.	Composition of louse lipids (by densitometer).
Table	10.	Metabolic products of tritiated oleic acid.
Table	11.	Composition of cuticular hydrocarbons.
Table	12.	Comparison of cuticular/internal/faecal hydrocarbons.
Table	13.	Fatty acid composition of cuticular cholesteryl
		esters.
Table	14.	Fatty acid composition of cuticular triglycerides.
Table	15.	Composition of cuticular fatty acids.
Table	16.	Fatty acid composition of cuticular 1,3-diglycerides.
Table	12.	Melting-point temperatures of cuticular waxes.
Table	18.	Solubility of synthetic cuticular wax in potential
		insecticide solvents.
Table	19.	Thermal death-points of adult lice.
Table	20.	Comparison of times for evaporation of (5ul)
		potential insecticide solvents.
Table	21.	Comparison of mortalities amongst adult female
		lice held in different container types.
Table	22.	Comparison of toxicity test results from lauryl
		ether using glass and plastic containers.
Table	23.	Comparison of toxicity test results after topical
		application of potential insecticide solvents upon
		adult female lice.
Tab le	24.	Comparison of lauryl ether toxicity from pure and
		aqueous forma.
Table	25.	Comparison of dieldrin toxicity resulting from

solutions in 1,4-butanediol and lauryl ether.

List of Figures.

r
ature,
of
the
the
ts.
221
е,
pta-
nta-
n
on
on
on
on

Figure	24.	Mass spectrum of synthesised product,
		15-hentriacontene.
Figure	25.	Recorder trace of change in electrical conductivity
		on the surface of a female sdult louse with
		temperature increase.
Figure	26.	Comparison of lauryl ether toxicities using glass
		and plastic containers.
Figure	27.	Plot of micrometer reading vs droplet mass for
		1,4-butanediol.
Figure	28.	Plot of micrometer reading vs droplet mass for
		didecyl ether.
Figure	29.	Plot of micrometer reading vs droplet mass for
		dodecanal.
Figure	30.	Plot of micrometer reading vs droplet mass for
		dodecanol.
Figure	31.	Plot of micrometer reading vs droplet mass for
		2-dodecanone.
Figure	32.	Plot of micrometer reading vs droplet mass for
		lauryl ether.
Figure	33.	Plot of micrometer reading vs droplet mass for
		1-octadecene.
Figure	34.	Plot of micrometer reading vs droplet mass for
		n-pentadecane.
Figure	35.	Plot of micrometer reading vs droplet mass for
		1,2-propanedio1.
Figure	36.	Plot of micrometer reading vs droplet mass for
		undecanol.
Figure	37.	Regression line for 1,4-butanediol.
Figure	38.	Regression line for didecyl ether.
Figure	39.	Regression line for dodecanal.
Figure	40.	Regression line for dodecanol.
Figure	41.	Regression line for 2-dodecanone.
Figure	42.	Regression line for lauryl ether.
Figure	43.	Regression line for n-octadecene.
Figure	44.	Regression line for n-pentadecane.
Figure	45.	Regression line for 1,2-propanedicl.
Figure	46.	Regression line for undecanol.

Figure	47.	Combined regression lines for the potential
		insecticide solvents.
Figure	48.	Toxicity of 10% aquecus lauryl ether (gel).
Figure	49.	Comparison of lauryl ether toxicity from pure
		and aqueous forms.

Plate	I a, b.	Feeding of lice on aneastheised rabbit.
Plate	2.	Containers tested for use in toxicity trials.
Plate	3 s,b.	Topical applicator for toxicity tests of solvents
		upon adult lice.
Plate	4.	Unstained cryostat section of sclerite cuticle.
Plate	5.	Unstained cryostat section of sclerite cuticle.
Plate	6.	Unstained cryostat section of membranous cuticle.
Plate	7.	Longitudinal section of adult louse stained with
		Hema toxylin/Phloxine.
Plate	8.	Transverse section of unfed adult louse stained
		with Mallory's triple stain.
Plate	9.	Transverse section of fed adult louse stained with
		Mallory's triple stain.
Plate	10.	Sclerite louse cuticle treated with Mallory's triple
		stain.
Plate	11.	Unsclerotised louse cuticle treated with Mallory's
		triple stain.
Plate	12.	Paratergal plate on the abdomen of female adult
		louse.
Plate	13.	Sclarite cuticle as in Plate 12 showing deposits of
		cuticular wax and/or cement layer.
Plate	14.	Sclerite cuticle showing epicuticular filamenta
		connected with cuticular surface.
Plate	15.	Sclerite cuticle showing the nonlamellate,
		homogeneous appearance of the endocuticle.
Plate	16.	Membranous cuticle with characteristic rugose
		formation.
Plate	17.	Membrancus cuticle in which the exocuticle and the
		endocuticle are macrofibrillar and a tubular network
		is evident underlying the epicuticle.
Plate	18.	Junction of sclerite cuticle and membranous cuticle.
Plate	19.	Sclerite cuticle showing the epidermis and the
		deposition zone separating the epidermis and the
		endocuticle.
Plate	20.	Membranous abdominal cuticle showing the surface

Plate	21.	Platinum/carbon replica of oblique cross section
		through membranous cuticle.
Plate	22.	Platinum/carbon replica of chitin macrofibrils in
		the endocuticle.
Plate	23.	Platinum/carbon replica of oblique cross section
		through epidermis/endocuticle.
Plate	24.	Scanning electron micrograph of female adult louse.
Plate	25.	Scanning electron micrograph of female adult louse.
Plate	26,	Head of female adult louse.
Plate	27.	Thorax of male sdult louse.
Plate	28,	Surface of membranous abdominal cuticle showing
		serrated formations on some rugosities.
Plate	29.	Surface of membranous abdominal cuticle showing
		the formation of pits in the region of 'quilted'
		cuticle.
Plate	30.	Surface of membranous abdominal cuticle showing
		disruption of the cuticular wax layer.
Plate	31.	Surface of sclerite cuticle showing the re-
		distribution of cuticular wax after preparation for
		SEM examination.
Plate	32.	Surface of membranous cuticle showing the re-
		distribution of cuticular wax after preparation for
		SEM examination.
Plate	33.	Surface of sclerite cuticle showing evidence of
		epicuticular surface pores.
Plate	34.	Sclerite cuticle on which surface pores are evident
		after extensive removal of cuticular wax.
Place	35.	Cryostat section of unfixed sclerite cuticle after
		critical point drying.
Plate	36.	Cryostat section of fixed membranous cuticle after
		critical point drying.
Plate	37.	Fracture surface of fixed and critical point dried
		sclerite cuticle.
Plate	38.	Fracture surface of fixed and critical point dried
		sclerite cuticle.
Plate	39.	Epicuticle and exocuticle of membranous cuticle
		prepared as in Plate 37.
Plate	40.	Cross section of membranous cuticle prepared as in
		Plate 37

Plate	41.	Surface of acetone-washed sclerite cuticle.
Plate	42.	Surface of hexane-washed sclerite cuticle.
Plate	43.	Surface of hexame-washed sclerite cuticle showing
		epicuticular pores.
Plate	44.	Surface of sclerite cuticle after treatment with
		alkali (KOH).
Plate	45.	Inner endocuticle surface of membranous cuticle
		after treatment with alkali (KOH).
Plate	46.	TLC separation of cuticular louse lipids.
Place	47.	TLC separation of louse lipids.
Plate	48.	UV exemination of untreated sclerite cuticle for
		autofluorescence in the Rhodsmine B region.
Plate	49.	UV examination of sclerite/membranous cuticle after
		treatment with lauryl ether only.
Plate	50.	UV examination of sclerite/membranous cuticle
		sectioned 1 hour after treatment with laury1 ether/
		Rhodamine B.
Plate	51.	UV examination of sclerite/membranous cuticle
		sectioned 4 hours after trearment with lauryl ether/
		Rhodamine B.
Plate	52.	UV examination of sclerite/membranous cuticle
		sectioned 1 hour after treatment with laury1 ether/
		Rhodamine B.
Plate	53.	UV examination of sclerite cuticle sectioned 4 hours
		after treatment with dodecanol/Rhodamine B.
Plate	54.	UV examination of sclerite/membranous cuticle
		sectioned 4 hours after treatment with lauryl ether/
		Rhodamine B.
Plate	55.	UV examination of sclerite/membranous cuticle
		sectioned 4 hours after treatment with dodecanol/
		Rhodamine B.
Plate	56.	UV examination of sclerite/membranous cuticle
		sectioned 4 hours after treatment with 1,4-butanediol
		Rhodamine B.

1. Introduction and Literature Review.

1.1. Structure of arthropod cuticle.

Arthropod cuticle is a remarkably complex, composite material which acts not only as the interface or barrier between living animal and environment but also as an exoskelston. Early investigators could not have envisaged the true extent of their task in attempting to define the nature of arthropod cuticle and its intrinsic functions. Although it is generally recognised that modern studies on outicle began with Künheit (1928 a. b, c) and Wigglesworth (1933), significant advances were made throughout the nineteenth century in the recognition and recording of basic aspects of cuticle architecture. With the continuous refinement of techniques since this period (and becoming highly intensified over the past three decades) the study of arthropod outicle has been, and remains, a rich field for scientific research, involving many disciplines. Numerous publications have been presented through the years, either as papers on specific topics related to arthropod cuticle, or as short general reviews (Wigglesworth, 1948, 1957, 1965; Hackman, 1971; Looke, 1974); but, remarkably, only two books devoted entirely to the comprehansive review of cuticle biology (Richards, 1951; Meville, 1975) have been produced. In recent years the widespread interest in the field has been reflected in the appearance of collections of monographs in book form on various aspects of arthropod cuticle (Hepburn, 1976; Gilby, 1980; Locke and Smith, 1980).

Perhaps the earliest evidence of the composite nature of arthropod outicle was provided by Odier (1823) who discovered and named chitin in the elytra of the May beetle and also recognized the presence of protein. In the same publication, he noted that the outermost layer of the cuticle was composed of a thin sheet of highly refractile material and this corresponds to the structure now known as the epicuticle. Thick outicle from arthropods such as beetles was particularly useful for examination with the microscopes of the period. Observations by Mayer (1842) on the elytra of the stag bestle Lucanus revealed the lamellate architecture of the endocuticle and the existence of lateral strands (balken) embedded within the homogeneous matrix of the layer. Although this fibrous cross-ply structure is restricted to arthropods such as beetles and crustaceans, Meyer's discovery helped to establish the concept of fibres as being the basic constructional units of cuticle in general. As microscopy improved and with the development of techniques involving fluorescence and polarised light, this concept was consolidated. The phenomenon of form birefringence observed in purified chitin (Clark and Smith, 1936) was believed to demonstrate the presence of parallel microfibres or micelles which formed the larger fibres. Much higher resolution was achieved with the development of the electron microscope and Richards and Korda (1948, 1950) were able to observe chitin microfibres which they presumed to be the basic structural units (chitin micelles). X-ray diffraction studies to determine the molecular conformation and structure of chitin (Frasnkel and

Eudall, 1940, 1947; Lotmar and Picken, 1950) were also undertaken during this period, but in many cases samples of isolated chitin were contaminated by residual protein which caused results to be misinterpreted.

Using Crustacean outicle, Bouligand (1965) demonstrated 250 A dismeter fibrous components which he regarded as the basic structural units of chitin. However, Eudall (1965, 1967) observed aven smaller components (now termed microfibrils or crystallites) with diameters from 25-60 Å in the ovipositors of adult Sirex and larval and puparial outicle of Sarcophaga. An early model (Fraenkel and Rudall, 1947) which proposed that cuticle is constructed of alternating monomolecular layers of chitin and # - extended protein was later challenged by Jensen and Weis-Fogh (1962) who concluded that such a system would be co-crystalline in structure and extremely brittle whilst the chitin concentration in the model would be uncommonly high. Instead, they regarded outicle as a composite in wiew of their results from mechanical experiments with locust cuticle. This interpretation was supported by Rudall (1967) who suggested that the basic outicle system consisted of chitin microfibrils embedded in a protein matrix. Using an improved staining technique which took advantage of the different reactions of chitin and protein with potassium permanganate and lead citrate, Heville and Luke (1969b) were able to achieve, using the electron microscope, a very high contrast between the two outicle components. Neville (1970) and Weis-Fogh (1970) subsequently demonstrated that microfibrils in

the locust prealar are ligament are composed of chitin and lie in a protein matrix. On the basis of X-ray diffraction studies, Rudall and Ks a chington (1973) estimated that the chitin molecule chains in each microfibril are arranged in 5 layers with 7 chains per layer. This was later revised to 3 layers of 6 chains by Neville (1975) who used an average microfibril diameter of 28 Å (Neville, Farry, and Woodhsad-Galloway, 1976) in calculating numbers of molecule chains per chitin microfibril. In several instances, optical diffraction patterns from electron micrographs of microfibril/matrix systems have indicated that the microfibrils are themselves arranged heragonally in the protein matrix (Neville et al. 1976).

The lamellate appearance of arthropod outicle has been the subject of considerable discussion over the years and, whilst the suggestion that lamellae allow for flexibility and stretching of the outicle by sliding over one another (Ahrens, 1930) found wide acceptance (Wigglesworth, 1965), it was strongly rejected elsewhere (Richards, 1951). Elohards (1951) interpreted the lamellae of insect procuticle as representing alternating layers of dense and less dense material. Examination of oblique sections of insect cuticle in the electron microscope revealed, however, that parabolic patterning exists between the lamellae (locke, 1960) similar to that seen in light microscope observations on Crustacean outicle (Drach, 1939, 1955; Dennell, 1960) which generally possesses thicker fibrous components. The theory that such patterning results from continuous area between the lamellae

(Drach, 1953; Dennell, 1960, Locks, 1974) was disproved by Bouligand (1965, 1972) who proposed instead, that belicoidal architecture of the outicle is responsible. Bouligand (1965, 1972) concluded that lemellate outicle is formed by successive microfibril planes which are parallel to one another and to the outicle surface. The direction of microfibril orientation in each plane differs from the preceding one by a small angle and always in the same direction. Such a formation had been suggested much earlier by Michelson (1911) whose observation that the cuticles of some arthropode reflect circularly polarised light prompted him to speculate that a screw-type of structure might be involved.

As pore canals traverse the chitin/protein complex of the outicle, their shapes (as seen from the surface) reflect the orientation of the successive microfibril layers. Fore canals were widely believed to follow a helical course (Eichards and Andersen, 1942; Wigglesworth, 1933, 1948; Locke, 1957) but later observations with both the light(Hewills <u>et al</u>, 1969) and electron microscope (Hevills and Luke, 1969a) revealed that they behave in a ribbon-like manner by rotating in helicoidal layers and remaining untwisted in layers of unidirectional microfibrils. On the basis of pore canal shapes and orientation in locust cuticle sections, Heville and Luke (1969a, b) formulated a two-system model for cuticle architecture in arthropods whereby layers consist of chitin microfibrils arranged either helicoidally in the protein matrix or unidirectionally to form a preferred layer.

1.2. Cuticular lipids of arthropods.

Most terrestrial arthropods examined to date possess a war layer on the epicuticle which is considered to be primarily responsible for minimising the transpirational water loss through the integument (Ebeling, 1974; Jackson and Eaker, 1970; Edney, 1974). The role of cuticular waxes in the waterproofing of arthropode was first reported by Ramsay (1935) when he observed that a rapid increase in water loss from the cockroach occurred above a critical temperature (about 35°C for the cockroach). Later, Wigglesworth (1945) obtained similar results using a wider range of insects: whilst Beament (1945) showed that there was general agreement between the critical temperatures recorded by Wigglesworth from whole insects and those produced by waxes extracted from the same insects and deposited on artificial membranes. Chemical analyses of cuticular waxes during this period were, at best, rudimentary; however, the advent of gas-liquid chromatography (GLC) in the early 1950s followed by thin-layer chromatography(TLC) at a later stage revolutionised organic analytical chemistry in that it became possible to reacive lipid mixtures into their individual components using small (submilligram) amounts of sample. Currently, the highest levels of performance in terms of resolution and identification of lipids is achieved by using a capillary column GLC and a mass spectrometer (MS) in an integrated system (GC-MS) linked to a data processing unit.

The earliest reports on arthropod outicular lipids, which involved the use of chromatographic techniques, were published in the 1960s and include descriptions of lipids from the Mormon cricket Anabrus simplex (Baker et al., 1960), the cookroach Periplaneta americana (Baker et al. 1963; Gilby and Cox, 1963) and larvas of the silkworm Samia cynthia ricini (Bowers and Thomson, 1965). Since then the cuticular lipid compositions of many arthropod species have been examined and the results comprehensively reviewed by various authors (Jackson and Baker, 1970; Hackman, 1974; Neville, 1975; Jackson and Blomquist, 1976; Blomquist and Jackson, 1979; Hadley, 1981). In addition, specific groups of lipids such as the cuticular hydrocarbons (Lockey, 1980a) and long-chain methyl-branched hydrocarbons (Helson, 1978) have also formed the basis of detailed reviews. Extensive accounts of recent techniques in ingect lipid analysis have been provided by Jackson and Armold (1977) and Gilby (1960).

Among the most common components of arthropod outicular lipids are hydrocarbons, free fatty acids, triglycerides, and sterols whilst wax esters, alochols and aldehydes are also occasionally present. Because hydrocarbons have been identified in all outicular extracts investigated to date and often comprise the most abundant lipid class (Jackson and Elonquist, 1976; Lockey, 1976), they are assumed to play an essential role in the waterproofing of arthropods (Hadley, 1978). Cuticular hydrocarbon

content in each of five species of desert tenebrionid bastles exceeds 90% of total lipids (Hadley, 1978) and similar levels have been reported in seven occkroach species (Jackson, 1970, 1972; Tartivita and Jackson, 1970). Consequently, many of the investigations carried out on cutioular lipids have been concerned only with the composition of the hydrocarbon fraction. There are numerous exceptions, however, where the hydrocarbon content of cuticular lipids is substantially lower than the levels (50 - 7%) suggested by Nelson <u>et al</u>, (1971) and include six species of temebrionid bastles (< 10%) which infest stored products (Lock ey, 1978a, b, 1979).

Although arthropod cuticular hydrocarbons represent a vast and complex series of compounds, they may be classified according to the following categories: n - alkanes. alkenes. monomethylalkanes with terminal branching, monomethylalkanes with internal branching, dimethylalkanes and trimethylalkanes. n - Alkanes, particularly those with an odd number of carbon atoms, are always present in arthropod cuticular hydrocarbons and n - tricosane, n - pentacosanen - heptacosane and n - nonacosane are often the most abundant(Lockey, 1980a). Exceptions include the termite <u>Zootermonfis</u><u>angusticollis</u> and the larval cigarette beetle <u>Lasioderma gerricorme</u>in which the major <math>n - alkanes are n - heneicosane (Blomquist <u>et al.</u> 1979a) and n - tritriacontane (Baker <u>et al.</u>, 1979b) respectively. Hadley (1978, 1981) has proposed that maximum cuticular impermethility is most likely to be achieved with a wax containing

a high proportion of saturated, long-chain compounds. Since it appears that arthropods in general are unable to synthesise n alkanes greater than 33-35 carbons (Jackson and Elomquist, 1976), Toolson and Hadley (1977) have suggested th that an increase in the concentration of branched methylalkanes containing more than 35 carbons might produce an overall decrease in permeability; albeit less effectively than that expected from comparable n - alkanes.

Research into the biosynthesis of outicular hydrocarbone has been rather limited when compared with studies on composition of cuticular hydrocarbons in arthropods. Nevertheless, Melson et al. (1971) noted a possible correlation between distary n - alkanes and integumentary n - alkanes of the tobacco hornworm Manduca sexts which was later supported by the findings of Blosquist and Jackson (1973) who included labelled n - alkanes in the diet of the grasshopper Melanopus sanguinipes to demonstrate that a significant proportion of cuticular n - alkanes could be derived from the dist, There is also evidence indicating that n - alkanes are products of biosynthesis. Labelled acetate was readily incorporated into all cuticular hydrocarbons of the cockroach Periplaneta fuliginosa (Blomquist and Kearney, 1976) and the termite Zootermopia angusticollis (Blomquist at al. 1979b) whilst labelled palmitate was preferentially incorporated into the n - alkanes of Z. angusticollis (Blomquist et al., 1979b). Results of experiments with cockroaches

a high proportion of saturated, long-chain compounds. Since it appears that arthropods in general are unable to synthesise n alkanes greater than 33-35 carbons (Jackson and Elomquist, 1976), Toolson and Hadley (1977) have suggested th that an increase in the concentration of branched methylalkanes containing more than 35 carbons might produce an overall decrease in permeability; albeit less effectively than that expected from comparable n - sikanes.

Research into the biosynthesis of cuticular hydrocarbons has been rather limited when compared with studies on composition of cuticular hydrocarbons in arthropods. Nevertheless, Belson et al. (1971) noted a possible correlation between distary n - alkanes and integumentary n - alkanes of the tobacco hornworm Manduca sexta which was later supported by the findings of Blomquist and Jackson (1973) who included labelled n - alkanes in the dist of the grasshopper Melanopus sanguinives to demonstrate that a significant proportion of cuticular n - alkanes could be derived from the dist. There is also evidence indicating that n - alkanes are products of biosynthesis. Labelled acetate was readily incorporated into all cuticular hydrocarbons of the cockroach Periplaneta fuliginosa (Blomquist and Kearney, 1976) and the termite Zootermopis angusticollis (Blomquist et al, 1979b) whilst labelled palmitate was preferentially incorporated into the n - alkanes of Z. angusticollis (Blomquist et al, 1979b). Results of experiments with cookroaches

(Major and Biomquist, 1978) indicate that biosynthesis of cuticular hydrocarbons occurs via an elongation - decarboxylation pathway whereby 16 and 18 carbon fatty acids are the precursors of longer chain fatty scids which are converted to alkanes by a decarboxylation step.

Although unsaturated hydrocarbons, mainly n - alkenes and n - alkadienes, occur extensively throughout arthropod cuticular lipids, they are usually only minor constituents (Jackson and Blomquist, 1976). Unsaturated cuticular hydrocarbons feature predominantly in four of the seven cockroach species investigated to date (Jackson, 1972). Cis - 9 - nonaccesene and cis, cis-6,9nonaccessdiene are the major cuticular hydrocarbons of Periplaneta japonica (Jackson, 1972) whereas cim-6, 9-heptacosadiene accounts for half the total cuticular lipids of P. americana (Gilby and Cor. 1963; Beatty and Gilby, 1969; Jackson, 1972). Cis - 9 - tricosene is the major cutioular hydrocarbon of P. australasiae and P. fuliginoss adult males, but is only a very minor component of adult female cuticular hydrocarbons (Jackson, 1970) and a similar disparity occurs in the case of cis-9heptacosene for adults of P. japonica (Jackson, 1972). Few reports on biosynthesis of unsaturated cuticular hydrocarbons have been published; however, cis, cis-6, 9-heptacosadiene, the major cuticular lipid of P. americana. is thought to be produced in post-ecdysial adults via a condensation - decarboxylation pathway in which linoleate is a precursor (Jackson and Baker, 1970; Conrad and Jackson, 1971).

Long-chain methyl-branched hydrocarbons have been detected only comparatively recently in arthropod cuticular lipids since their identification usually requires the use of a GC-MS system. Monomethylalkanes may be divided into two groups; terminally branched monomethylalkanes comprising 2 - methyl, 5 - methyl and 4 - methylalkanes and internally branched monomethylalkanes in which the methyl branched is positioned centrally on the carbon chain. As a class of hydrocarbons, monomethylalkanes occur almost universally throughout the outicular lipids of arthropods and in some instances, such as the adult cigarette beetle Lasioderma serricorne (Baker et al, 1979b) and the desert tenebrionid beetle Elecdes armata (Hadley, 1977), account for as much as 87% and 92.7% respectively of the total cuticular hydrocarbons. Overall, 5-methylalkanes have been reported in over twice as many cuticular lipid mixtures as 2 - methylalkanes and 4 - methylalkanes have been identified only in the adult cigarette beetle Lasioderma serricorne (Baker et al. 1979b). Although 2 - methylalkanes are usually only minor hydrocarbon components when present, the crickets Acheta domesticus. Gryllus pennsylvanicus and Nembius fasciatus (Blomquist at al, 1976) and the bestle Phyllobius maculicornia (Jacob, 1977) are uncommon in each having a 2 - methylalkane series which accounts for 20-30% of their respective cuticular hydrocarbons. When present in cuticular lipids, 3 - methylalkanes usually comprise a reasonable percentage of the hydrocarbons. The fire ants Solenopeis invicta

and <u>S</u>. <u>richteri</u> have levels of 3 - methylalkanes in their cuticular hydrocarbons which, at 39% and 34% respectively (Lok <u>at al</u>. 1975), are among the highest reported for arthropods.

Internally branched monomethylalkanes have been identified in the majority of cuticular hydrocarbon mixtures investigated and usually form an appreciable proportion of the total. The cockroaches Periplaneta brunnes. P. fuliginosa (Jackson, 1970), Leucophase maderas and Blatta orientalis (Tartivita and Jackson, 1970) have particularly high concentrations of internally branched monomethylalkanes comprising over half of each cuticular hydrocarbon mixture whereas these monomethylalkanes are absent in P. americana (Jackson, 1972). In general, internally branched monomethylalkanes containing even numbers of carbon atoms are more prevalent than those with odd carbon numbers and the methyl branch is usually located on an odd-numbered carbon atom (Melson, 1978). They often occur as members of isomeric mixtures in which carbon numbers are identical and the methyl branch is located on different oddnumbered carbon atoms. Adults of the cigarette bestle Lasicderma serricorne (Baker et al. 1979b) have unusual cuticular hydrocarbons which contain internally branched monomethylalkanes of odd and even carbon numbers as well as mixtures of isomers with methyl branches on odd- and even - numbered carbon atoms as in mixtures of 10-, 11-, 12- and 13-methylhexacosanes. Similar situations have been observed in surface hydrocarbons from several species of tensbrionid bestles (Lockey, 1978b, 1979, 1980b).

Although dimethylalkanes occur less frequently in cuticular lipids than monomethylalkanes, they generally have the longer chain lengths (Hadley, 1981). Isomeric mixtures of internally branched dimethylalkanes are common and are usually composed of molecules with methyl branches on odd-numbered carbon atoms separated by three carbon atoms referred to as isoprenoid spacing (Lockey, 1980a). The cricket Gryllus penneylvanicus has a relatively high proportion of internally branched dimethylalkanes (34%) with typical structures containing odd carbon numbers from 35 to 39 inclusive (Blomquist et al. 1976). A number of unusual dimethylalkanes have been identified including a series of 11, 12- isomers from the larval cigarette beetle Lasioderma serricorne which contain 55, 34, 35 and 37 carbon atoms (Baker et al. 1979b). Blomquist et al. (1979a) detected a unique outioular dimethylalkane, 5, 17- dimethylheneicosane, in all castes of the termits Zootermopis angusticollis which comprised about one quarter of the hydrocarbons.

In comparison with other cuticular hydrocarbons, trimethylalkanes are extremely rare. All of those identified so far have shown isoprenoid spacing of the methyl branches such as 11, 15, 19 - and 13, 17, 21 - trimethylpentatriacontanes from the grasshopper <u>Schiatocerce vage</u> (Nelson and Sukkestad, 1975). Martin and MacConnell (1970) identified two homologous series based on 3, 7, 11- and 4, 8, 12 - trimethylalkanes in the whole body hydrocarbons of the ant <u>Atta colombics</u>. Collectively, the trimethylalkanes formed the largest fraction within the extracted hydrocarbons which

were considered to be mostly cuticular in origin.

A number of studies on the biosynthesis of branched cuticular hydrocarbons have been conducted based on the incorporation of labelled substrates in surface hydrocarbons. Results from experiments using the oricket Nambius fasciatus indicate that valine is incorporated into 2 - methylalkanes with odd numbers of carbon atoms after conversion to isobutyric acid (Blailock et al, 1976). Similarly, leucine is incorporated into the outicular 2-methylalkanes with even carbon numbers of the cricket Gryllus pennsulvanicus following conversion to isovaleric acid (Blailock at al. 1976). Labelled propionate was preferentially incorporated into the 3 - methylalkanes of the cockroaches Feriplanets americana and P. fuliginoss as well as the internally branched monomethylalkanes of the latter (Blomquist et al. 1976; Blomquist and Kearney, 1976). Data from trials on the termite Zootermonis angusticollis indicate that the methyl branches of mono - and dimethylalkanes are derived from incorporation of propionate in place of acetate during chain elongation. (Blomquist et al. 1979b).

Although triglycerides, free fatty moids, and sterols are seldom major components of, arthropod outicular lipids, their presence in the majority of those surface lipids investigated so far suggests that they (along with hydrocarbons) are in some way involved in the waterproofing process.
The surface lipids of the naiad big stonefly <u>Pteronarcys</u> <u>californics</u> contain an exceptionally high proportion of triglycerides (76%) resulting in a lower melting mixture than that of the adult which has a greatly reduced triglyceride content (7%) (Armold <u>et al</u>. 1969). Qualitative and quantitative differences in the surface lipids of the equatic naiad and terrestrial adult forms of <u>P</u>. <u>californics</u> appear to vary with environmental need for water concervation (Armold <u>et al</u>. 1969).

Falmitio acid (16:0) and oleic acid(18:1) are usually the predominant free fatty acids in arthropod outicular lipids (Jackson and Blomquist, 1976) but changes in free fatty acid composition with stages in insect development have been reported. Whilst free fatty acid content of the cuticular lipids from the larval and adult black carpet bestle <u>Attacenus momentum</u> remained fairly constant during development, a significant increase in unsaturation accompanied by a general reduction in chain length was observed in adult fatty acids (Baker <u>at al.</u> 1979a). In comparing outicular permeability and cuticular lipid composition of two scorpions <u>Hadrurum a. arizoneneis</u> and <u>Droctonus apachesnus</u>. Toolson and Hadley (1977) recognised among free fatty acids inverse correlations between permeability and extent of saturation and between permeability and molecule size.

Cholesterol is usually the single most abundant component (>7%) in cuticular sterols although its concentration in total

outioular lipids is generally low. The steroid nucleus cannot be synthesised in arthropods (Gilbert, 1967) and minos cholesterol is sesential for development in many insect species (Vroman at al. 1964), it (or pressureors) must be acquired through the dist.

Involvement of cuticular lipids in various functions apart from maintaining the water balance in arthropods has been reported in numerous species (Heville, 1975; Jackson and Elomquist, 1976; Nelson, 1978). On the basis of their effects on cultures of <u>Aspergillus flavus</u>, surface lipids from larvae of the silkworm <u>Bombyr mori</u> have significant entifungal properties, with esturated, medium-chain free fatty acids being the most potent components (Koidsumi, 1957). Examples of surface lipid components from insects acting as kairomones and sex attractants are videspread and include many long-chain alkanes which are either unsaturated or have methyl branches (Melson, 1978).

1.5. The penetration of arthropod outicle.

Studies on the permeability of arthropod outicle to chemicals have become increasingly significant in recent years in view of the widespread use of chemical toxicants in the control of insects which pose a threat to man. Despite intensive research during the past few decades however, the mechanics of cuticular penetration are still not clearly defined. Early researchers such as Richards (1951) recognized the complexity of the field and the need for unambiguous quantitative data in attempting to understand the phenomenon, but many investigations have been qualitative in approach and conflicting observations have been published. In addition, experimental data from studies in which the existence of a direct relationship between rates of intoxication and penetration was assumed are of limited value since various physiological interactions have been found to occur when chemical mixtures are used (Brattsten and Wilkinson, 1977). In spite of the considerable interest in the field, only two major reviews have appeared to date. Ebeling (1974) published a comprehensive review of work carried out up until 1974 whilst Lewis (1980) has compiled a detailed account of the various techniques which may be used in studying cuticular penetration with an emphasis on producing quantitative data.

While the effectiveness of the epicuticular war in minimising water losses from arthropods is well established (Edney, 1974), its hydrophobic or lipophilic character is also responsible for the resistance of the outicular surface to wetting by external water.

In contrast, non-polar liquids, which are not usually encountered by insects in their natural environment, readily wet the cuticular surface and cause extensive disruption of the spicuticular wax layer (Ebeling, 1976). Insecticidal form ulations are usually prepared with this in mind and the toxicant is presented to the insect as a solution in a non-polar solvent. Some controvermy exists regarding the main route of entry for contact insecticides, but most evidence suggests that penetration occurs via the pore canals and hasmolymph (Vigglesworth, 1942; Levis, 1965, 1980; Burt and Lord, 1968) rather than the trachese (Gerolt, 1969, 1970).

Histological studies by Wigglesworth (1942) on the fifth stage nymphs and adults of Rhodnius prolinus demonstrated that the pore canals and dermal gland ducts could facilitate the passage of oils through the outicle and this was the earliest concrete evidence for penetration via the cuticle. More recently, Bruck and Komnick (1971) observed, using the electron microscope, penetration of the epicuticular filaments and pore canals of the cockroach Periplaneta americana by an inorganic phosphate; while Lewis (1980), using autoradiographic techniques, has demonstrated absorption of []E]cleic acid through the pore canals in the antenna of the locust Schistccerca gregaria. Much of the experimental evidence in support of insecticide entry via the pore canals/haemolymph route is indirect. Using the blowfly Phornia terraenovae. Lewis (1965) compared the quantities of DDT absorbed at various parts of the body which were representative of a broad spectrum of outicular porosity. After a period of 48 hours, the antennas, which possess the most porcus outicle, had absorbed the greatest quantity of DDT per unit area

followed by the cores and sternal-coral membranes and finally the lens of the compound eye which possesses no pore canals. This demonstrated that not only did penetration proceed much faster in cuticle where pore canals were present, but also that penetration could still occur in cuticle in which pore canals were absent.

In the same publication, Lewis reported that absorption of DDT from spheres of lanoline applied to the head of the blowfly occurred initially at a rapid rate which declined steeply thereafter. Although the rate of absorption into dead flies became negligible after 30 hours, the absorption rate into living flies eventually reached a steady value. Lewis suggested that the high initial absorption rate for DDT was caused by a high concentration gradient across the outicls and the steady declins in absorption rate resulted as the concentration gradient was diminished by the accumulation of insecticide in cells adjacent to the treated area of cuticle. In living flies, circulating hasmolymph was partly responsible for the final steady rate of absorption; whereas, in dead flies, a state of equilibrium was reached as the concentration gradient of DDT across the cuticle approached zero. Bi-phasic absorption in insects has also been observed with several other insecticides including malathion (Matsumura, 1963), famphur and dimethoate (Buerger and O'Brien, 1965) and isomers of benzene hexachloride (Kurihara, 1970), and a number of investigators (Hoffman et al. 1952; Le Roux and Morrison, 1954; Matsumura, 1963) have also found that entry rates of insecticides for dead insects are significantly lower than those for living insects.

In studies using the radioactively labelled insecticide, [¹⁴c]-DDT on houseflies <u>Musca domestica</u>. Le Roux and Morrison (1954)

detected radioactivity in the hasmolymph of the cervical region within 30 seconds of topical application of the labelled DDT in bensol to the tibiofemoral membrane of a metathoracic leg. Benezet and Forgash (1972) detected radioactivity in the hasmolymph of houseflies within 15 seconds of topically applying thoraces with solutions of $[^{14}c]$ -malathion in acetone. A further indication of the involvement of hasmolymph in the transportation of toxicants was demonstrated by Burt <u>et al.</u> (1971) using the cockrosoh <u>P. metricana</u>. Exposed abdominal nerve cords in untreated insects were irrigated with hasmolymph from insects poisoned with doese (LD_{95}) of pyrethrin I. Electrophysiclogical monitoring of the cords revealed that after 10 minutes, spontaneous activity in sixth abdominal ganglis increased tonfold.

On the basis of the following experimental observations, Gerolt (1969, 1970) proposed an alternative theory to the widely held view that insecticides traverse the outicle via the pore canals and are transported to the site of action by the haemolymph. When he deposited dieldrin in beeswax at various loci on the body of the housefly <u>M. domentica</u>, the effect was greatest at sites near the central nervous system (CNS) and least at the more distant loci. Gerolt maintains that if the insecticide is transported to the CNS by the haemolymph, distance between CNS and site of topical application would have little effect.

Although a reduction was noted in the speed of action of various insecticides upon houseflies covered in a thin film of olive oil, no reduction was observed for flies containing injections of the oil. Insecticides used in the experiment were highly soluble

in olive cil and would be expected to be readily absorbed by the internal deposits.

Bouseflies which were exposed to a surface deposit of dieldrin (0.2 µg/fly) suffered complete knockdown in 65 minutes whereas the same mass of crystalline dieldrin introduced into the abdominal cavity produced no obvious effects in the flies during the initial 7 hours. The organophosphate insecticides, chlorphenvinphos, methyl parathion and dichlorvos produced similar results.

When [¹⁴C]-dieldrin in benzene was topically applied to isolated insect integument mounted in a diffusion cell, no accumulation of radioactivity was detected in the underlying saline solution over a 24 hour period. Integuments from various insects were tested in the cell.

In houseflies, a band of beesvar around the 'vaist' of an insect appeared to prevent insecticide (dieldrin) applied to the abdomen from migrating to the thorax by the outicular route. Paralysis was observed among flies without war barriers soon after treatment with an equivalent dose. Cockroaches with sealed abdominal spiracles and war barriers displayed no adverse effects for 3 days after application of dieldrin crystals to their abdomens; but those without any protection were paralysed within 6 hours and those with war barriers only were affected within 17 hours.

Autoradiographic investigations in which [¹⁴G]-dieldrin was applied to the abdomene of locusts <u>Schistocerca</u> <u>gregaria</u> showed that the insecticide penetrated into the deeper layers of the integrment. Gerolt suggested that the dieldrin was concentrated in the endocuticle-hypodermis region and proposed that migration of the insecticide proceeded laterally in the integument into the trachese.

After housefly larvas were exposed to a dry deposit of $[^{14}C]$ -dieldrin for various periods of time, the tracheal trunks and other organs were dissected out and subjected to autoradiography. Although the distribution was uneven, some labelled insecticide was detected in the tracheal trunks and showed an increase with time of exposure. The hasmolymph gave no detectable image.

On the basis of all these results, Gerolt concluded that contact insecticides accumulate in the integument, probably the endocuticle, and spread laterally to reach the site of action by way of the integument in the tracheal system.

In earlier experiments by Hoy and co-workers (1943) with cookroaches, paralysis was caused by the introduction of powdered pyrethrum into the trachese via the spiracles. They considered that the toxicant reached the ganglis by diffusing through the trachesl walls into the hasmolymph rather than following the direct route through the trachesl system.

Burt and co-workers (1971) conducted experiments which were designed to test Gerolts hypothesis. In one study, three different methods of insecticide presentation were compared. Doses (IL_{95}) of pyrethrin I were applied to cockroaches either, (a) in a solution of acetone on the metathoracic sterna or, (b) in a solvent-free emulsion with seline on the metathoracic sterna or, (c) in a solventfree emulsion with saline into the tracheal system through a meta-

thoracic spiracle. The tracheal route of pyrethrin I entry was no more effective than the topical route and, in fact, appeared to be slightly less efficient over a 48 hour period.

In another experiments the sixth abdominal gaughton and the associated trachesi systems were exposed in isolated abdomens of cookroaches. Fyrethrin I was introduced into one of the longitudinal wentral traches close to the gaughts in 0.15 pl aliquote in a solvent-free emulsion. Only large doses of pyrethrin I produced any change in spontaneous nerve activity when introduced into the trachese. Activity was greatly decreased but without the initial burst of activity observed when gaughts are irrigated externally with high concentrations of pyrethrin I.

Burt at al. (1971) have cast further doubts on Gerolt's theory on the basis that even after reaching the ends of the tracheoles within the CHS, an insecticide still faces a barrier of glial cells before reaching the neurones. They have suggested that if insecticides can traverse these, it should be possible for them to reach the baseolymph by diffusion through the tracheal walls.

Although the phenomenon of outioular permeability is considered to be an essential occurrence in the model generally accepted for insecticide penstration, that is, passage through the outicle into the hasmolymph and thence to the CNS, Ebeling (1974) recognized the importance of three physico-chemical factors; liposolubility, surface migration and phase partition, which contribute to the

overall permeability of outicle to insecticides. The active compound is usually presented to the outicle as a solution in cil or in a more volatile organic solvent and, in many cases, the choice of solvent can be crucial. For example, Levis (1963) has demonstrated that red flour bestles <u>Tribolium castaneum</u> absorb dieldrin from hydrocarbon cil solutions at rates which vary inversely with the viscosity of the oil and the size of the cil molecules. Levis (1962) has also provided evidence of surface migration on insect outicle by showing that when insects walk over fine deposits of oil, the whole of the external surface becomes covered within minutes with at least a monomolecular film.

Phase partition effects assume significance in outicle permeability in that an insecticide must be able to migrate from the wax of the epicuticle to the underlying hydrophilic layers of the procuticle and beyond. It has been found that pyrethrin I, although much more liposoluble than diazoxon, is absorbed by the cockroach <u>P. americana</u> at a slower rate and this has been attributed to its lesser solubility in the squeous phase within the insect (Burt and Lord, 1968; Burt et al. 1971).

1.4. Aims of the work.

A dramatic escalation has been observed in recent years in the levels of human louse infestation in many countries and, as a result, much attention has been focussed on the need for new and effective insecticides. The development of improved insecticides for the control of lice is extremely important in view of the potential for the spread of disease associated with widespread infestations, and also the fact that insecticide registance has been observed amongst head lice.

The colony of human body lice maintained at the London School of Hygiene and Tropical Medicine is one of perhaps, anly three or four laboratory colonies in the world and provided the ideal opportunity to carry out several lines of research on this ectoparasite. Although the body louse has been the subject of numerous investigations much earlier in this contury, interest was centred mainly on physiological, taxonomic and behavioural characteristics.

In an attempt to bring the information on this insect up to date and in line with what is currently known of other species, I have endeavoured to make use of as many advanced techniques as possible. The overall approach, however, has been to investigate those areas of research which might yield the most worthwhile data for use in formulating insecticidal preparations in general.

2. Materials and Methods.

2.1. Examination of the structure of body louse outicle.
2.1.1. Light microscope (1M) examination.
2.1.1.1. Examination of unstained outicle sections.

Male and female adult lice (2-3 days old) were smap-frozen by submerging in isopentane (melting-point - 159.9°C) cooled in liquid nitrogen. The lice were then embedded in Tissue - Tek II 0.C.T. Compound (G.T. Gurr, Ltd., London, S.W.6) on separate outting blocks and placed in the refrigerated compartment of a Slee orycetat (Slee Medical Equipment Ltd., Londen, S.E. 15) maintained at - 25°C. Sections ranging in thickness from 6 to 8 μ were out from various locations on the lice and examined under oil using a Ziess Photomioroscope II.

2.1.1.2. Examination of stained cuticle sections.

Mature male and female adult lice, fed and unfed, were fixed separately in Deboscq Brasil (G.T. Gurr Ltd., London, S.W.6) for 46 hours at room temperature prior to dehydration in Supercedrol (G.T. Gurr Ltd.) for 72 hours with two changes of the solvent during this period. The fixed lice were placed in molten Paraplast (G.T. Gurr Ltd.) for 96 hours and then solidified in freeh Paraplast to form blocks for outting. Sections (5-10 μ) were out from the blocks with a rotary microtome (A. Gallenkamp and Co. Ltd.,London, E.C. 2P), and the most satisfactory results were obtained when the blocks had been previously ohilled at 4°C for 1 hour.

Material produced in this way was stained with either Mallory's triple stain (G.T. Gurr Ltd.) or Hematoxylin and Phloxins (G.T. Gurr Ltd.) and examined as before (see section 2,1,1,1.).

2.1.2. Transmission electron microscope (TEN) examination.2.1.2.1. Examination by TEM of outicle sections.

Unfed male and female adult lice (2-3 days old) were out into quarters under an ice-cold solution of 5% glutaraldehyde in 0. IM cacodylate buffer at pH 7.2. After fixation in the solution for 2 hours at 4°C, the specimens were washed with two changes of 0.1M cacedylate containing 0.3M sucross for a further 20 minutes at the same temperature. Post-fixation was carried out in 1% osmium tetroxide in the cacodylate buffer at 4°C for 1 hour. The specimens were then washed as above at room temperature and dehydrated in an alcohol series before being subjected to two 5 minute washes in propylene oxide. Embedding took place in Araldite at 60°C over 12 hours after the specimens had been brought through a propylene oxide - Araldite series in which the Araldite concentration was progressively increased. Sections of outicle approximately 50 nm and 100 nm thick were out on an LEB ultramicrotome (Ultrotome III - LKB Instruments Ltd., South Croydon, Surrey) and floated onto 400 - mesh, formwar coated copper grids and post stained with uranyl acetats and lead citrate. Examination of the sections under low and high resolution was

performed using transmission electron microscopes, Ziess EM9, AEI 6B and AEI 801.

All chemicals and grids used throughout the procedure were obtained from EM scope Laboratories Ltd., London, S.W.S

Platinum/carbon (Pt/C) replication of outioular surface and cross section.

Unfed male and female adult lice (2-3 days old) were dissected at the junction of the thorax and abdomen and as much of the contents as possible removed from the abdominal cuticle under distilled water. After a further wash in distilled water. specimens of abdominal cuticle were allowed to stand overnight in a solution of 20% glycerol in distilled water prior to freezefracturing. The impregnated specimens were positioned in metal stubs which were then placed into a small container of Freon 22 held in liquid nitrogen (- 196°C). These rapidly frozen specimens and stubs were then installed in the vacuum chamber of a freeze-fracture unit (BAF 301 High Vacuum Freeze-Etch Unit, Balzers High Vacuum Ltd., Berkhamsted, Herts.) where temperature and pressure were reduced to - 110°C and 2 x 10⁻⁶ torr respectively. A machanically operated razor was used to out some frozen specimens while others were left intact. All specimens remained frozen and under vacuum during shadow-costing (at angle 45°) with Pt/C then support-coating (at angle 90°) with carbon. Replicas formed in this way were floated off in distilled water and recovered on TEM grids for examination in Ziegs EM9 and AEI 6E transmission electron microscopes.

2.1.3. Scanning electron microscope (SEM) examination.
2.1.3.1. Morphology and surface topography of the adult louge.

Whole, unfed male and female adult lice (2-3 days old) were fixed and post-fixed as described (see section 2.1.2.1.) and then oritical point dried in acetone in a Polaron E3000 dryer (Polaran Equipment Ltd., Holywell Industrial Estate, Watford, Herts.) attached to a Churchill thermocirculator (Churchill Instrument Co. Ltd., Uxbridge, Middleser). Individual spacimens were mounted on SEM stube and coated with a thin layer of gold in an Edwards S150 sputter coater (Edwards High Vacuum, Crawley, West Sussex) in preparation for examination at low resolution in a JNOL JSM-T20 scanning electron microscope (JEOL UK. Ltd., Colindale, London N.W.9) and high resolution in a JNOL JEM -100 CX TEMESCAN.

2.1.3.2. Examination of cuticular cross section.

Cryostat sections were prepared as described (see section 2.1.1.1.) using unfixed adults and fixed (see section 2.1.2.1.) oritical point dried adult lice. The specimens were mounted and sputter coated prior to SEM examination of the cross sectional faces (see section 2.1.3.1.). Fractured specimens were prepared by gently tearing the carcasses of fixed, dried adults and the resultant fragments were then mounted on SEM stubs. Examination was carried out after sputter coating.

2.1.3.3. Solvent effects on the outicular way layer.

Whole, unfed adult lice (male and female, 2-3 days old) were agitated for 10 minutes in either acetons or hexane prior to fixation and critical point drying as described (see section 2.1.3.1.). Specimens were prepared and examined by SEM also as described (see section 2.1.3.1.).

2.1.3.4. Effects of alkali treatment on the cuticle.

Specimens of abdominal outiols from unfed male and female adult lice (2 - 3 days old) were dissected as described (see section 2.1.2.2.) and submerged in 37% potassium hydroxide (KOE) at room temperature for 24 hours. The treated specimens were than dried in air and prepared and examined by SEM as described (see section 2.1.3.1.). 2.2. Chemical Composition of Louge Lipids.

2.2.1. Comparison of potential extraction solvents for outicular lipids.

2.2.1.1. Trial thin-layer chromatography (TLC) separations.

The following organic solvents were selected as potentially suitable for extraction of the cuticular lipids from the louse.

- i acetone
- 11 benzene
- iii chloroform
- iv chloroform/methanol (2 : 1)
- v disthyl ether
- vi dimethorymethane
- vii ethyl acetate
- viii hexane.

A 10ml aliquet of each solvent was placed into separate 25ml beakers followed by 200 adult lice (unsered) and the mixtures gently agitated. After a 10 minute extraction, the contents of the beakers were filtered through separate small, sintered glass (size 3Å) funnels and the extracts evaporated almost to dryness in a stream of nitrogen. Five microlitre samples of each extract were then spotted onto separate lance of a prevashed TLC plate (20 om x 20 om Marck, Kieselgel HF 254 + 366, EDE Chemicals Ltd., Middlessex, U.K.) which was subjected to a 2-step, 1 dimensional development socording to Skipski at al (1965). This comprised development first with isopropyl ether/acetio acid (96:4) until the solvent front ascended 13 om., drying in nitrogen and development again in the same direction with petroleum ether (40° - $60°_0$)/

disthyl ether/acetic acid (90 : 10 : 1). All TLC solvents had been previously double-distilled in glass. On completion of the second development (solvent front ascended to 19 cm.), the plate was dried in nitrogen and placed in an atmosphere of iodine vapour. The resultant spots were noted and the plate then sprayed with 40% sulphuric acid before heating in an oven for 3 houre.

2.2.1.2. Solubility of long chain hydrocarbon $(n - C_{36})$ in potential extraction solvents.

A 1 ml aliquot of each test solvent (numbers i to viii) was placed into separate stoppered 5 ml tubes. Five milligrams of hexatriacontane (n - 0_{36}) was added to each tube and the mixtures shaken. After 10 minutes at room temperature (25^o C) the solvents in which the hexatriacontane had dissolved were noted.

2.2.2. Extraction of louse lipids.

2.2.2.1. Cuticular lipid extraction using whole lice.

Hexane proved the most satisfactory solvent for extraction of cuticular lipids (see section 3.2.1.) and was used exclusively for this purpose in all subsequent experimental work. Without prior purification however, commercially pure hexane was found unsuitable in preparing lipid extracts for eventual gas chromatographic analysis. Minor contaminants were concentrated along with the outicular lipids during evaporation of the barane in the nitrogen stream. Hexans used throughout all experimental work was obtained as 'pure' hexane (Nanograde, Mallinckrodt) from Camlab Ltd., (Cambridge, U.K.) and then double-distilled using a spinning band column.

Pasteur pipettes (short form) were prepared as illustrated in Fig. 1 . Lice for solvent extraction were weighed collectively (e.g. 17.75 g, approx. 22,000 lice), then placed in the pipettes filling them to the three-quarters level. Rubber bulbs were connected to the pipettes (Fig. 1) and enough pure because to cover the lice drawn from a 25 ml reservoir of the solvent. To minimise contamination of the lipid extract, glassware used throughout all procedures had been previously cleaned in chromic acid and rimsed finally in mostone.

After 10 minutes, the because was expelled from the pipettes into the reservoir and the procedure repeated rapidly twice to rinse the contents. The combined solvent extracts were then evaporated to dryness in a nitrogen stream and the mass of remaining lipids determined using a Cahn Electrobalance.

On a number of occasions, adult lice were separated according to sex, counted and weighed prior to havane extraction as above. After extraction, the lice were thoroughly dried and weighed.

2.2.2.2. Cuticular lipid extraction using cast skins.

Cast skine were collected from large colonies of lice (in excess of 100,000 insects) over a period of one week. Removal of extraneous matter (e.g. fasces and dead bodies) was accompliabed by placing the orude mixture in a vertical glass column





connected to a gentle stream of nitrogen entering at the base (Fig. 2). Being the lightest of the mixture, the skins were readily carried to the top of the column for collection. When enough skins had been collected, they were removed and washed in distilled water, dried and refluxed in obloroform for 6 hours in a soxhlet apparatus. The chloroform extract was evaporated to dryness in a nitrogen stream and the recovered lipids dissolved in several microlitres of hexame in preparation for TLC analysis. The quantity of lipids extracted was minute and prevented accurate weighing. TLC analysis of the lipids produced spurious results (see section 5.2.3.1.) and all further analyses of cuticular lipids were performed on samples from whole insects.

2.2.2.3. Examination of cuticular lipid extraction technique (using whole lice).

A colony of lice (8.66 11 g, approximately 10,000 lice) of both series and at various stages of development were collected for feeding on a live rabbit. The rabbit (adult female New Zealand white) was anaesthetized with an injection of Saffan (alpharalone 0.% (w/w), alphadolone acetate 0.% (w/w); Glaxowst Ltd., Greenford, U.K.) and all fur removed from the stomach area with electric clippers. Blood (5.5 ml) was collected in a defibrinated syrings from vessels of both ears. Tritisted oleic acid (5 m Ci; n9, 10 B^3) (The Endlochemical Contre, Amersham, U.K.) dissolved in toluene was added to % (rabbit) albumin in 0.5 ml of Kreb's Ringer solution (Ca ^{+ +} omitted)(see Appendix). The toluene solution, which floats

on the albumin solution, was evaporated in a gantle stream of nitrogen and 5 ml of the blood added with thorough mixing. A small volume (0.5 ml) was retained for scintillation counting. The labelled blood was re-injected into the rabbit via vessels in the ears and allowed to circulate for 10 minutes. Feeding of the lice colony was then permitted for 10 minutes upon the exposed skin of the rabhit's stomach. Blood samples were taken from shallow incisions in the area of lice feeding. immediately before and after feeding. The rabbit was sacrificed by injection immediately after removal of the lice. Extraction of the lice with hexane followed (see section 2.2.2.1.) and the dead lice retained. Lipids extracted by the hazane were recovered after evaporation of the solvent and weighed before adding to a vial containing 15 al of scintillation solution (0.6% w/w PPO and 0.02 % w/w POPO in toluene). The dead lice were homogenised in a glass tissue grinder with a total volume of 50 ml. of distilled water and the homogenate sonicated (Rapidis sonicator; Ultrasonic Ltd., Shipley, Yorks, U.K.) for 5 minutes at 100 watts. A fine suspension was formed which was extracted three times for lipids using chloroform/methanol (2 : 1). These extracts were combined, filtered and the solvent evaporated to recover the lipids which were finally dissolved in scintillation solution as above. Blood samples were centrifuged at 5000 rpm using a Beckman TJ-6 benchtop centrifuge with TH4 rotor and 40ul aliquota of supernatant sers spotted and dried on separate scintillation

discs (Whatman 2.5 cm circles, Whatman Labsales Ltd., Maidstone, Kant) before adding to scintillation vials as above. Scintillation counts were performed at 4⁰C (Packard 3003 scintillation spectrometer) for 1 minute periods for the following specimens:

- i 40 µl blood serum before radio-labelling
- ii 40 pl blood serum before lice feeding
- iii 40 µl blood serum after lice feeding
- iv extracted outicular lipids
- v extracted whole body lipids
- vi blank (15 ml sointillation solution only for background).

2.2.2.4. Extraction of internal lipids from whole adult lice,

Twenty female adult lice (2 - 3 days old) were fed simultaneously with the large louse colony on radio-labelled rabbit blood (see section 2.2.2.3.). After 10 minutes, the lice were removed and held alive in an incubator for 24 hours (tamperature 30° C and relative humidity 75°). The fascal matter was then removed by washing the insects in 10 ml of distilled water and the washings retained for lipid extraction (see section 2.2.2.5). Cuticular lipids were collected as described (see section 2.2.2.1.) and the insect bodies homogenised with 15 ml of distilled water, sonicated and extracted, also as described (see section 2.2.2.5.).

The procedure described above, was repeated using two larger colonies of lice consisting of 1000 male and 1000 female adults respectively; however, feeding took place on unlabelled rabbit blood (i.e. separate rabbit). In both instances, lipid extracts were evaporated to dryness in a nitrogen stream and dissolved in approximately 5 μ l of hexane before TLC analysis (see section 2.2.5.).

2.2.2.5. Extraction of lipids from louse fasces.

The faecal washings (10 ml) collected from the adult lice fed on radio-labelled rabbit blood (see section 2.2.2.4.) were sonicated and extracted for lipids as described (see section 2.2.2.3.). Extracted lipids were obtained by evaporating the solvent extract to dryness in a nitrogen stream, then prepared for TLC analysis (see section 2.2.3.) by dissolving in 5 µl of hexane.

Fancal vashings (20 ml each) were also collected from the two colonies of adult lice (1000 males and 1000 females respectively) described (see section 2.2.2.4.) and combined prior to monication and lipid extraction as above. Recovery and of the extracted lipids from the solvent their preparation for TLC analysis also followed the above procedure.

2.2.3. Thin-layer chromatographic (TLC) analysis of lipid extracts.
2.2.3.1. Qualitative TLC analysis of outicular lipids/internal lipids/fascal lipids.

TLC plates (20 om x 20 cm) were prepared using Marck Kieselgel $HF_{254 + 366}$ spread in layers of 0.3 mm thickness. The plates

were dried, guide lines drawn and subsequently given a chromatographic run in chloroform as an effective means of washing. After drying and activation at 110° C for 1 hour, cleaned plates were spotted with 1 pl aliquots of the following lipids in because:

- i cuticular lipids from unsexed whole lice (see section 2.2.2.1.)
- ii cuticular lipids from cast skins (see section 2,2,2,2,)
- iii internal lipids from adult male lice fed on normal rabbit blood (see section 2.2.2.4.)
- iv internal lipids from adult female lice fed on normal rabbit blood (see section 2.2.2.4.)
- v fascal lipids from adult (male and female) lice fed on normal rabbit blood (see section 2,2,2,5,).

Lipid standards (see below) were dissolved in hexane and 1 µl of each solution spotted on the plates prior to TLC development, thus enabling comparison with the unknowns to be made.

8.	hydrocarbon	-	hexatriacontane
ь.	cholesteryl ester	-	cholesteryl stearate
c.	methyl ester	-	methyl stearate
d.	triglyceride	-	triclein
е.	free fatty moid	-	cleic acid
f.	1, 3 - diglyceride	-	1, 3 - diolein
g.	1, 2 - diglyceride	-	1, 2 - diolein
ь.	cholesterol	-	obolesterol
1.	1 - and 2 - monoglycerides	-	mono olein (90%)

The plates were developed under the same conditions as those described in 2.2.1.1. After completion of the TLC run the plates were thoroughly dried under nitrogen and exposed to iodine vapour. When the spots became obvious, the plates were observed under UV radiation (366 nm) and the results recorded by photographing with a Folaroid CU-5 Land Camera (Folaroid Ltd., St. Albans, Harts. U.K.) fitted with a Wratten UV 2E filter (Kodak Ltd., Holborn, London). Detection with 40% sulphuric acid and heat was carried out and the resultant spots photographed under both natural light and UV radiation (366 cm).

2.2.3.2. Quantitative TLC analysis (by weight) of cuticular lipids.

Preparative TLC plates (20 cm x 20 cm) were formed using Merck Eisselgel 60 PF 254 + 366 spread in 1.5 mm layers. Cutioular lipids from the extraction of approximately 15,000 unsegregated lice were spotted in a band formation along the origin and subjected to a one dimensional, two step development as described (see section 2.2.1.1.). After completion of the second development, the plates were dried in an atmosphere of nitrogen and exposed to icdime vapour to locate the lipid classes. Each lipid class was collected separately by carefully scraping the silica gel from the plates and placing it in Pasteur pipettes prepared with glass wool instead of filter paper as described (see section 2.2.2.1.). Lipids were eluted from each pipette with pure toluene and collected in separate 20 ml flasks. The toluene was then removed by evaporation in a stream of nitrogen and each lipid class dissolved in 20 µl of hexane to allow transference to small vials previously weighed on a Cahn Electrobalance. After evaporation of the solvent. the vials were reweighed to determine the lipid weights. Two

replicates were performed.

2.2.3.3. Quantitative TLC analysis (by densitometer) of outioular lipids/internal lipids/fascal lipids.

The plates containing TLC separations of the following lipids (see section 2.2.3.1.) were used in the quantitative estimations;

i cuticular lipids from unsered whole lice

- ii internal lipids from adult male lice fed on normal rabbit blood
- iii internal lipids from adult female lice fed on normal rabbit blood
- iv fascal lipids from adult (male and femals) lice fed on normal rabbit blood,

After visualisation of the lipid classes with 40% sulphuric moid and heating, the plates were examined using a comming densitometer (E.C. densitometer; Camlab Ltd., Cambridge, U.K.) to produce a linear representation of the TLC separations. Peaks from the same separations were then out out and weighed and percent composition calculated from the weights.

2.2.3.4. TLC analysis of digestion products of labelled fatty acid (Oleic acid).

The twenty adult female lies fed on rabbit blood labelled with tritiated cleic acid (see section 2.2.2.3.) were extracted after 24 hours for cuticular and internal lipids (see section 2.2.2.4.). Fascal matter expelled during this period was also extracted for lipids (see section 2.2.2.5.) and the total of each extract separated into lipid classes and recovered as described (see section 2.2.3.2.). The various lipid classes were counted for radioactivity as described (see section 2.2.2.3.) over 10 minute periods.

2.2.3.5. Test for saturation of outicular hydrocarbons.

Cutioular hydrocarbons were examined for maturation using the method of Jackson et al (1974). TLC plates were prepared as described (see section 2.2.3.1.) and sprayed with 9% ethanol (in water) saturated with silver nitrate. The plates were then beated to 100⁰C for several minutes to remove the ethanol and allowed to cool. Those not used immediately were stored in darkness in a desicoator. Cutioular hydrocarbons and reference standards were spotted onto plates for development in hexage followed by visualisation with 40% sulphuric acid spray and oharring with beat.

- 2.2.4. Gas-liquid chromatographic (GLC) analysis of lipid extracts.
- 2.2.4.1. Modification of Perkin-Elmer F11 gas obromatograph (GC) for capillary column use.

Initially, GC equipment consisted of a single channel flame ionisation detector (FID) gas chromatograph (Perkin-Elmer F11) coupled to a temperature programmer. The instrument was not specific-dly designed for use with glass columns; however, glass columns were prepared with dimensions allowing exact alignment with the existing inlet and outlet column couplings inside the GC oven. Two columns (10ft x 3/32 inch and 5 ft x 3/32 inch) were packed with 3% OV 101 (Chrompack U.K. Ltd., London SW16) on Chromosorb W-EP (80 - 100 mesh) (Chrompack U.K. Ltd.,) and 10% butanediol succinate (EDS) (Chrompack U.K. Ltd.,) on Chromosorb W-HP (80 - 100 mesh) respectively.

An homologous series of straight chain alkanes (carbon numbers 15 to 25) were eluted from both columns with temperature programming from 70° C to 190° C and 270° C for EDS and OV101 phases respectively, separation of alkanes containing more carbons was found increasingly difficult and higher final oven temperatures appeared necessary. Methyl silicone, OV101, proved a more suitable stationary phase than EDS under these conditions because of its higher stability at the increased temperatures (maximum temperature limit, OV101 (150° C), EDS (200° C)).

The small quantities of lipids available for analyses proved a limiting factor in the use of the packed glass columns. Injection of volumes less than 1 µl required operation of the FID smplifier at maximum sensitivity; however, baseline increase under these conditions, due to phase bleed and FID sensitivity change with temperature, was excessive. In response to the problems encountered, particularly that of using minute samples, it was decided to install a capillary column in the GC.

Fig. 3 shows the final arrangement with the capillary (10m x 0.5msWCOT 0V101 - Phase separations Ltd., Clwyd, U.K.) attached to the FID Analyser Unit. The restricted space within



Figure 3. Installation of capillary column in Perkin-Elmer F11 GLC.

the small GC oven governed the placement of the column (coiled dismeter 11 cm.). It was suspended diagonally in the front half of the oven, avoiding all obstacles. All gas connections to the column were made with pre-shrunk PTFE tubing (internal dismeter 0.5 mm.) capable of enduring temperature up to 260° without deformation. Prior to connection of the PTFE tubing, it was important for the sharp glass edges of the capillary column to be gently abraded with fine emery paper and the ends of the PTFE tubing opened out with a pencil point.

The injection system of the F11 was unsuitable for use with the capillary column and, instead, nitrogen carrier gas was supplied directly to the column via PTFE tubing attached to nickel tubing through the insulated oven lid. A straight length of glass capillary tubing (approximately 5 cm.) was inserted between the nickel supply line and the column to facilitate introduction of samples for analysis. The carrier gas supply was controlled by a toggle valve positioned close to the column and was switched off while the straight glass tubing was gently removed from the system. Samples for analysis were deposited on the internal surface by capillary action, ensuring that excessive volumes of solvent were not ultimately introduced into the column itself, leading to removal of the stationary phase. This was avoided by allowing most of the solvent to evaporate before replacing the straight tubing in the system. After re-connection, the gas supply was snapped on simultaneously with the temperature programmer which was set from room temperature to 260°C. The carrier gas

pressure (8 pai) was pre-set for the system to give a flow rate of 25 cm. mec-1 for methane. Stability and mensitivity of the FID were optimized by employing two modifications. Firstly, the FID was insulated from the main oven and maintained at a steady, controllable temperature between 300°C and 350°C using a length of insulated heating cord coiled around the exterior of the detector (Fig. 5). Secondly, the flame-drawn end of a straight length of glass capillary tubing was inserted through the metal tubing connected to the FID to stop a few millimetres short of the FID flame (Fig. 3). The swagelock connection towards the other end was fitted with a graphite ferrule to secure the capillary tubing in place and provide a gas-tight seal. With the hydrogen supply also connected at this junction, an effective sweep of the volume preceeding the FID was established. Using the system described above, it was possible to obtain excellent chromatograms from sample volumes less than 1 pl.

2.2.4.2. Initial GLC separation of cuticular hydrocarbons.

Cuticular lipid samples from segragated male and female adult lice (see section 2.2.2.4.) were prepared by TLC as described (see section 2.2.3.2.) and subjected to GLC analysis with the system described (see section 2.2.4.1.). The column used for the GLC separations was an OV 101 WCOT glass capillary column (10 mm x 0.5 mm) temperature programmed from room temperature to 270° C at 10° C minute.⁼¹ Bitrogen (oxygen-free) was the carrier gas used at a pressure of Apri at the column inlet. As comparison showed the chromatograms to be identical, further analyses

were conducted using cuticular hydrocarbons from unsered adults for quantitative purposes. Feak areas in subsequent chromatograms were estimated using an electronic integrator (Venture Digital Integrator NR II; Smiths Industries, London 5W2.) and in turn, percent composition of the component hydrocarbons was calculated.

2.2.4.3. GLC examination for saturation of outicular hydrocarbons.

Cuticular hydrocarbons from unsexed adult lice were separated according to saturation on a silver nitrate impregnated TLC plate (see section 2.2.3.5.). The regions of silics gel containing the saturated and unsaturated hydrocarbons were located under UV radiation (366 nm), scraped from the plate and the hydrocarbons eluted separately from the material with pure hexane (see section 2.2.3.2.). Both solutions were evaporated to dryness in a stream of nitrogen and redissolved in 10 µl of hexane prior to GLC analysis under the conditions described (see section 2.2.4.2.).

Unsaturated hydrocarbons were removed from a second outicular hydrocarbons sample using bromination. A solution of bromine in tolusne was added to the hydrocarbon sample also dissolved in tolusne and allowed to stand overnight at - 10° C. The brominated hydrocarbons were recovered after evaporation of the tolusne and bromine and analysed under the GLC conditions described (see section 2.2.4.2.). Chromatograms produced by brominated and unbrominated cuticular hydrocarbons were then compared and the differences noted.

 2.2.4.4. GLC examination for branched/unbranched cuticular hydrocarbons.

A cuticular hydrocarbon sample was added to 10 ml of 2,2,4 trimethylpentane containing activated Linde molecular sieve 5A as 1/16 inch pellets. The mixture was allowed to stand for 24 hours and then filtered (O'Connor et al, 1962). Normal hydrocarbons were taken up by the molecular sieve while the branched hydrocarbons remained in solution and were recovered after evaporation of the solvent. GLC analysis under conditions as described (see section 2.2.4.2.) identified the chromatogram peake produced by branched hydrocarbons.

2.2.4.5. Comparison of unknown GLC peaks with standard GLC peaks.

An homologous series of normal hydrocarbons (carbon numbers 15 to 32) was prepared and chromatographed as described (see section 2.2.4.2.). Comparison of the retention times produced by the homologous series with those of the cuticular hydrocarbons indicated the carbon chain lengths of the unknowns. The hydrocarbons series was then co-chromatographed with the cuticular hydrocarbons and the resulting chromatographed with the cuticular hydrocarbons prease which indicated hydrocarbons of equivalent carbon number.

2.2.4.6. Position of double bond in unsaturated outicular hydrocarbons.

Unsaturated cuticular hydrocarbons (see section 2.2.4.5.) were subjected to a periodate - permanganate oxidation (Downing and Greene, 1968) which caused cleavage at the double bonds. From the products

of the cleavage, it was possible to deduce the structure of the major alkenes.

Briefly, 5 mg of unsaturated outioular hydrocarbons was dissolved in 10 ml. of pure t-butyl slochol and added to 10 ml. of distilled water containing 120 mg of sodium periodate and 10 mg of petassium permanganate. The solution was brought to pH 8-9 with the addition of aqueous potassium carbonate, stirred for four days then extracted with methylene chloride to remove neutral lipids. After neutralisation with aqueous sodium bisulphite, the remaining aqueous solution was extracted with chloroform which was dried, filtered and evaporated to dryness. The fatty acids which were formed by the reactions, were methylated by dissolving in 2% sulphuric acid in anhydrous mathanol and heating to 70°C in a closed tube. After 1 hour, the methyl estors formed were recovered by other extraction and analyzed by GLC as described (see section 2.2.4.2.).

2.2.4.7. Gas chromatography/Mass spectroscopy (GC/NS) analysis of outioular hydrocarbons.

Cuticular hydrocarbons from unsered adult lice were analysed by GC/MS using both Electron Ionisation (E.I.) and Chemical Ionisation (C.I.).

The first analyses were performed using E.I. on a Kratos MS 25 mass spectrometer coupled to a Data General DS-50 computer system. The glass capillary column (10 m x 0.3 mm 07101 WCOT)

used in previous GLC analyses (see Materials and Methods) section 2.2.4.2.) was installed in the GC unit (Sigma 3) and connected directly to the spectrometer. Helium at 5 psi (flow, 2 ml minute⁻¹) was used as the carrier gas and the GC oven programmed from 100°C to 260°C at 10°C minute⁻¹ during the GC run. Injection (splitless) temperature was 300°C while the spectrometer inlet was held at 275°C. The epectrometer was operated with Electron Energy at 70 eV, Ionising Current 100 uA, Accelerating Voltage 1.35 KV and source temperature at 200°C.

An analysis was performed with the system described above using the mass spectrometer in the C.I. mode and ultrapure methans as the ioniming gas. Operating conditions were duplicated with the exception of Electron Energy which increased to 100 eV and Ioniming Current became Emission Current 5 mA.

The outicular hydrocarbons were also examined using a Finnigan 1020 GC/HS system. A 50 m glass WCOT SE 30 capillary column was used for the GC separation with temperature programming from 150° C to 270° C at 10° C minute ⁻¹. Mass spectral data were processed by a computer incorporated in the system,

2.2.4.8. GLC analysis of internal hydrocarbons/fascal hydrocarbons.

Internal lipids from 1000 male and 1000 female adult lice respectively were collected (see section 2.2.2.4.) and the hydrocarbons obtained using column chromatography with alumina
(Merck, neutral, activity grade 1; EDH Chemicals Ltd., Middx. U.K.) and hexane as the elutant.

Vascal lipids were extracted from the combined fascal vashings of the colonies described above (see section 2.2.2,5.). The hydrocarbons were separated and collected as above and the three extracts concentrated by evaporation in a stream of nitrogen. GLC analyses were performed using a Perkin-Elmer Sigma 3 gas chromatograph fitted with a 2m x 3mm packed (% 07 - 1 on Chromosorb W-HP, 100-120 mmsh) glass column and temperature programming from 150°C to 300°C at 10°C minuts ⁻¹. Injection and detector blocks were heated to 350°C and nitrogen (oxygenfree) flowing at 32 cm³ minuts⁻¹ was the carrier gas. n-Alkanes were identified by comparison of retention times and cochromatography with authentic standards.

Fascal hydrocarbons were also analysed by GC/MS using a Kratos MS25 mass spectrometer. The preliminary GC/MS separation was performed on a 50 m OV 101 WCOT glass capillary column with temperature programming from 150° C to 300° C at 10° C minute ⁻¹. All other operating conditions were identical to those described (see section 2.2.4.5.).

2.2.4.9. GLC analysis of cuticular cholesteryl esters.

Cutioular lipids were separated into lipid classes by TLC and each collected (see section 2.2.3.2.). A 0.5 mg sample of cholesteryl esters was added to 5 ml. of 2% sulphuric acid in anhydrous methanol in a glass tube. The mixture was sealed and heated to 70°C for 1 hour then extracted with ether to recover the methyl esters produced. After evaporation of the ether, the esters were redissolved in 5 pl of hexane and analysed using a Pye Unicam Series 204 gas chromatograph fitted with a 10m x 0.3 mm OVICI WCOT glass capillary column. An injection splitter (S.G.E. Pty. Ltd., London NW2) was used at the injection end of the column while the straight end at the detector stopped several millimetres short of the FID flame as described (see section 2.2.4.1.) to enhance sensitivity. Capillary column connections were PTFE tubing as described (see section 2.2.4.1.) which limited the upper temperature to 260 - 270°C. Nitrogen (axygen-free) carrier gas was supplied at 7-8 psi and the GC separation temperature programmed from 100°C to 265°C at 10°C minute⁻¹. Comparison of retention times and co-chromatography with standards assisted in identifying the main methyl/ester peaks, while electronic integration (Venture Digital Integrator MK II; Smithe Industries, London NW2) was used to calculate their percent composition.

2.2.4.10. GLC analysis of cuticular triglycerides.

The triglyceride fraction from a TLC separation of outicular lipids (see section 2.2.3.2.) was treated in the same manner as the cholesteryl esters (see section 2.2.4.9.) converting each triglyceride fatty acid component into a fatty acid methyl ester (F.A.M.E.). GLC analysis of the methyl esters was performed on the Pye Unican Series 204 gas chromatograph under conditions as described (see section 2.2.4.9.). The saturation of the methyl esters was not investigated. 2.2.4.11. GLC analysis of cuticular fatty acids.

The fatty acid fraction was collected from a TLC separation of cuticular lipids (see section 2.2.3.2.) and mothylated as described (see section 2.2.4.9.). Conditions for GLC analysis of the fatty acid mothyl esters were as described (see section 2.2.4.9.) using the same equipment. He examination of fatty acid methyl ester saturation was carried out.

2.2.4.12. GLC analysis of cuticular 1, 3-diglycerides.

The 1, 3-diglyceride fraction was collected from a TLC separation of outioular lipids (see section 2.2.3.2.) and treated with 2% sulphuric acid in anhydrous methanol as described (see section 2.2.4.9.) to form fatty acid methyl esters. CLC analysis of the methyl esters was performed with equipment and conditions as described (see section 2.2.4.9.).

No examination of methyl ester saturation was carried out.

2.2.5. Synthesis of major unsaturated outicular hydrocarbon (15 - hentriacontene).

71

The cuticular hydrocarbon 15 - hentriacontene, was prepared via the route indicated below.

(2.2.5.1.)

CH_OH τ

1 - pentadecanol

partadectator

(2.2.5.2.) CH OH

PEr.

CH2Br

II

1 - bromopentadecane(alkyl halide

CH

0



VI

heradecanal (aldehyde)

111

1 - hexadecanol

(2.2.5.3.)

CH_Br



.

pentadecylmagnesiumbromide (Grignard Reagent)

II



2.2.5.1. Pr-paration of alkyl halide (II) -intermediate .

 Pentadecyl bromide was prepared in a substitution reaction where the halogen replaced the hydroxyl group of 1 - pentadecanol (Noller and Dinemore, 1943).

An excess of phosphorus tribromide (5 ml.) was added with stirring to a solution of anhydrous disthyl ether containing 25 g of anhydrous 1-pentadecanol (I) while at -10° C. The temperature of the mixture was kept below 0° C during the addition and was allowed to reach room temperature overnight. Crude halide was obtained by distillation of the mixture (160° C at 10 mm pressure), washed twice with 5 ml. of concentrated sulphuric acid and shaken with dry potassium carbonate until the hydrogen bromide odour had disappeared. The balide was redistilled at atmospheric pressure (approximate temperature 320° C) and the distillate (II) recrystallised from chloroform and weighed (16.65 g). Overall yield for the 1-pentadecyl bromide was 52.3% and the melting point found to be 20.2° C.

2.2.5.2. Preparation of aldehyde (IV) - intermediate .

The primary alcohol, 1-bewadecanol, was oxidised to form the aldehyde (IV), 1-bewadecanal, following the method of Partch (1964).

A solution containing 15 g. of 1-hexadecancl in 150 ml, of pyridine was stirred while an equal quantity of powdered lead tetraacetate was added at room temperature. The solution turned

yellow after several hours indicating that exhaustion of the oxidant had occurred. Addition of lead tetrancetate was repeated as above until the solution finally remained red. The reduced lead tetrancetate was precipitated on cooling and removed by filtration, and after evaporation in vacuo of the pyridine, the crude hexadecanal was distilled at $200^{\circ}C$ (30 mm pressure). The examination of the aldehyde (see section 2.2.3.1.) revealed the presence of hexadecanol as a minor impurity which was removed by column chromatography of the distillate on a silica gel column (1 cm x 60 cm). Hexadecanal was eluted first using hexame/disthyl ether (90:10 V/V) as the sluant and after recovery by evaporation of the solvent, its weight (6.26 g) represented a 57.2% yield. The melting point of the hexadecanal (IV) was 32.5°C.

2.2.5.3. Preparation of Grigmard Reagent (V) - intermediate.

A spherical reaction flack containing 2 g of pure magnesium turnings covered with 30 ml of anhydrous disthyl sther was fitted with two Quickfit dropping funnels, which effectively sealed the flack. Enough halide from a solution containing 16.65 g of 1-bromopentadecame in 20 ml of anhydrous disthyl ather was added from one dropping funnel to initiate the reaction. Addition of the solution was maintained at a slow rate until exhaustion to avoid violent boiling and the final mixture warmed alightly to ensure completion of the reaction. The freshly made Grigmard Reagent was used immediately.

2.2.5.4. Formation of secondary alcohol (VII) - intermediate.

A solution containing 8.26 g of hexadecanal (IV) in 20 ml of anhydrous disthyl ether was added dropwise to the Grignard Reagent (V) via the second dropping funnel of the apparatus as described (see section 2.2.5.3.). After addition of the whole aldehyde solution, 10 ml of ice-cold aqueous chlorids solution was added to the mixture and the sther and aqueous layers allowed to separate before extracting the latter three times with diethyl ether. The united volumes of ether extracts were dried over anhydrous sodium sulphate and the crude alcohol recovered by evaporation of the ether and distillation at atmospheric pressure (approximate temperature 390° C). Recrystallisation of the distillate (VII) from hexane produced 6.64 g of 16 hentrimecontanol which represented a combined yield of 43.2% for procedures 2.2.5.3. and 2.2.5.4. A melting point of 87°C was recorded for the product.

2.2.5.5. Formation of alkene (VIII) from alcohol (VII).

The secondary alcohol, 16-bentriacontanol, (see section 2.2.5.4.) was dehydrated using orthophosphoric acid to form the alkene (VIII), 15-bentriacontene following the method of Corson and Ipatieff (1955). One double bond per molecule was formed, which, in this case was possible on either eide of the central carbon atom (carbon no. 16). The final products, however, were identical since both possible molecular structures consisted of 15 and 16 carbon chains on either side of the double bond

respectively (see diagram in section 2.2.5.). It was expected that both cis - and trans - forms would be produced in the reaction.

A distillation flamk containing 6.64 g of 16 - bentriscontanol in 10 ml of 8% orthophosphoric acid was gradually heated at atmospheric pressure and water distilled off. Crude alkene was collected around 380° C and purified by column chromatography with alumins (Merck, nentral, activity grade 1) and herane as the eluant. The product (VIII) weighted 3.59 g which represented a yield of 55.1% for the reaction and a melting point of 46° C recorded.

A GLC examination of the 15 - hertriacontene was carried out using a Perkin-Elmer Sigma 3 gas chromatograph under conditions as described (see section 2.2.4.8.) while a GC/MS examination was also performed using a Kratos MS25 mass spectrometer operating under conditions as described (see section 2.2.4.6.).

2.5. Applied aspects

2.3.1. Maintenance of lice colonies

All lice used in experiments throughout these studies were obtained from four large colonies which were originally established in 1972 using eggs from rabbit-adapted lice (Orisodo strain) supplied by the U.S. Department of Agriculture in Gainesville, Florids.

After feeding once daily, each colony of lice was placed in a separate glass beaker (diameter 15 cm) containing 3 cm squares of black corduory fabric and kept in the dark in an incubator at 30° C and 75% relative humidity. Eggs were laid on the fabric squares in the colonies and removed with the material after two week periods when clean material was introduced. In general, eggs laid on the squares required 9 days for hatching and all nymphs had emerged by the eleventh day.

A group of 12 rabbits, mostly New Zealand white, were kept exclusively for the maintenance of the lice colonies. Preparation of the rabbits involved anaesthetisation with an intraemscular injection of Saffan (0.% v/v alphaxalone, 0.% v/v alphadolone; Glaxovet Ltd., Greenford, U.K.), then the stomach area shaved as closely as possible with electric olippers (Oster

Professional model A = 5) and loose fur removed with a vacuumcleaner. The animal was then transferred to a plastic trough in which it lay on its back supported by the structure (Plate ia). Nymphal feeding was commenced by placing the egg-laden fabric squares on the shaved stomach of the rabbit for 5 minutes allowing newly emerged nymphs to migrate from the squares. All squares were then removed and returned to the incubator while the nymphs continued feeding for a further 5 minutes until being collected for addition later to the main colonies.

Feeding of the main colonies followed a similar format whereby fabric squares and lice from a colony were placed on a rabbit for 5 minutes before returning the squares to the original container. At this point, the lice colory was cleaned of cast skins, fasces and other accumulated debris by gently vacuuming over the feeding lice while they remained securely attached to the skin of the rabbit. Time of feeding totalled 15 minutes which ensured that virtually all lice were fully engorged before their collection by hand from the surface of the rabbit. The small numbers of lice which escaped collection in this manner were removed by wiping the shaved stomach of the rabbit with cotton wool and destroyed. In practice, the populations of the colonies were allowed to reach 15,000 - 20,000 adults before reducing the numbers to around 5,000 adults per colony.

Plates 1a and 1b. Li

Lice feeding on anaesthetised rabbit supported in trough.



 2.3.2. Physical characteristics of cuticular lipids and adult lice.

2.3.2.1. Melting-point determination of cuticular war.

Approximately 5000 unsexed adult lice were extracted for outioular lipids as described (see section 2.2.2.1.) and the wax deposited in a melting-point tube. The melting-point temperature was determined using a Gellenkamp melting-point apparatus (A. Gallenkamp and Co. Ltd., London BC2P), programmed for a slow temperature increase. A total of 5 extractions were carried out and melting-point temperature determined for each extract.

2.3.2.2. Solubility of simulated outicular wax in potential insecticide solvents.

A simulated war mixture was prepared containing n - alkanes and n - alkene (see Table 1) in the same proportions as the corresponding naturally occurring compounds determined in section 3.2.4.2. Homogeneity was achieved by agitation of the moltan mixture (60° c). The n - alkene (15 - hentriacontene) was prepared as described (see section 2.2.5.).

2 ml. vials were set up, each containing 1 mg of the simulated wax mixture. Ten potential insecticide solvents listed below

Table 1

Composition of synthetic outicular war

Hydrocarbon	Carbon No.	% Composition	Mass mg
n - tricosane	23	3.0	0.66
n - tetracosane	24	1.3	0.30
n - pentacosane	25	20.8	4.63
n - hexacosane	26	1.5	0.33
n - heptacosane	27	22.5	5.00
n - octacosane	28	2.5	0.56
n - nonacosane	29	23.3	5.19
n - triacontane	30	0.7	0.16
n - hentriacontane	31	1.5	0.33
15-hentriacontene	31	22.9	5.10

(Table 2) were then added in 100 μ l aliquots to the separate numbered vials and stoppered before placing in an incubator at 30°C and 75% relative humidity.

After 24 hours, the solvents in which the war mixture had dissolved were recorded. Where the war mixture had not dissolved, 100 µl aliquots of the corresponding solvents were added to the vials which were then returned to the incubator for a further 24 hours period. The addition of solvents in 100 µl increments was repeated until each 1 mg sample of war mixture had dissolved and the volume of solvent necessary for dissolution in each case recorded.

Compound	Melting-point (°C)	Empirical formula	Structural formula
1,4-butenedicl	20	^C 4 ^H 10 ⁰ 2	но
didecyl ether	16,5	с _{20 В₄₂ 0}	
dodecanal	13	с ₁₂ В ₂₄ о	رe.
dodecanol	22	с _{12 ^д26} 0	ОН
2-dodecanone	17	с _{12 В₂₄ о}	` <u></u>
lauryl ether	3-5	с ₂₄ В ₅₀ о	
1-octadecene	17.5	С ₁₈ н ₃₆	
n-pentadecane	10	с ₁₅ ш ₃₂	
1, 2-propanedicl	-60	с ₃ н ₈ о ₂	Он
undecanol	19	C ₁₁ H ₂₄ 0	Он Он

-

Table 2. Potential insecticide solvents.

2.3.2.3. Determination of upper lethal temperature of adult lice.

Two male and 10 female adult lice were collected 2-3 days after moulting (see section 2.3.3.1.) and carefully placed into separate melting-point tubes. Each louse was subjected to a slow increase in temperature (around 1.66° C min⁻¹) in a Gallenkamp melting-point apparatus and the temperature at which the insect ceased all movement recorded.

In a second series of experiments, lice were placed in melting-point tubes and set up in turn with an electrode in contact with the louse as illustrated (Fig. 4). The electrode was a simple construction, consisting of two wires insulated from each other by means of a glass capillary tube and was connected in series to a 1.5 volt battery and pen recorder (Fig. 4).



Figure 4. System for determination of upper lethal temperature.

As the temperature of the melting-point tube was slowly elevated using the melting-point apparatus, the recorder measured the potential of the circuit and recorded its variation with time and temperature.

2.3.2.4. Comparative volatilities of potential insecticide solvents.

A solution was prepared containing 3 mg of simulated was mixture in 250 ml. of hexans. Using a syringe, 1 ml aliquots of the solution were carefully deposited upon the horizontal surface of 10 clean microscope slides (26 mm x 75 mm) so as to cover the entire uppermost surface of the slides without spilling over the edges. Evaporation was allowed to take place at room temperature (22°C) resulting in the formation of a wax layer (approximately 0.63 g cm⁻²) with an average thickness of 0.66 micron (Lookey, 1960).

Each of the potential insecticide solvent described earlier (Table 2) were then deposited as single 5 µl droplets on the prepared microscope slides. The slides were held in an incubator at 30° C and 75% relative humidity and examined twice daily to establish the time taken for each solvent to completely evaporate in order to calculate rates of evaporation.

2.5.3. Toxicity tests with adult lice.

2.3.3.1. Preparation of test material.

A degree of standardisation among the test insects was achieved by selecting only females which had moulted 2-3 days earlier and using them in the tests approximately 6 hours after last feeding. Initially, 250 nymphs at the pre-adult stage were collected from the main colonies while under anaesthetization in a gentle flow of carbon dioxids (Fig. 5) and returned separately to the incubator (see section 2.3.1.).



Figure 5. Equipment for collection of numphs and adult female lice.

Following each 24 hour period, successive batches of 250 nymphs were set up to form a total of 4 batches whilst all adult males were removed from preceding batches. The lice were fed daily as described (see section 2.3.1) in conjunction with the main colonies.

At any time, the most mature sub-colony of lice contained in excess of 120 adult females, 2-3 days old for use in tests.

2.3.5.2. Protocol for toxicity tests.

Toxicity tests followed a format whereby 120 adult female lice (100 test, 20 control), 2-5 days old (see section 2.3.5.1.) were placed in separate small containers. One hundred of these lice were topically applied with the potential insectioide solvent under a microscope.

In standardising the procedure, 3 types of containers (Flate 2) were used to hold the separate lice in a series of preliminary toxicity tests. One type of container (1 cm dism x 0.5 cm height) was composed of glass, and produced from out down vials while the other two were plastic and normally used as caps for bottles (Flastic A - Packard scintillation vial cap, 1.5 cm ID x 0.5 cm height, Flastic E - glass vial cap, 1.7 cm ID x 0.7 cm height). All containers were prepared for use in the tests by scaking for several hours (overnight) in a 3% solution of Decon 90 (Decon Laboratories Ltd.,



Plate 2. The 3 types of containers used to hold separate lice during preliminary toxicity tests. (Details in the text) Portalade, Brighton, U.K.), rinsed twice with distilled water and dried while inverted at 60°C. The preliminary toxicity tests involved 4 potential insecticide solvents (dodecanol, 2-dodecanone, lauryl ether and undecanol) in conjunction with the 3 types of containers. Substantial differences were noted between the mortality rates obtained with the various containers (see section 3.3.2.1.) and consequently, the glass containers were chosen for use throughout all subsequent tests on the basis of their relative inertness in contact with the solvents.

A topical applicator (after Busvine) (Flate 3) was used in the procedure and consisted of a 10 µl microsyrings (SGE (UE) Ltd., London HW2 7AT) coupled to a micrometer. Capillary glass tubing, 4 cm in length and flame-drawn to 0.1 mm internal dismeter, was attached with Araldite to the end of the steel syrings needle to facilitate application of 0.01 - 0.2µi volumes.

A relationship between the actual mass of solvent deposited by topical application and the corresponding number of micrometer divisions was established. At various dosage levels for the solvent, 20 droplets of equal volume were deposited collectively in pre-weighed glass containers produced by sealing one end of thin-walled glass tubing (internal diameter 1 mm) and cutting approximately 2 mm from the sealed end. Bolvent and container were weighed on a Cahn Electrobalance and the mass of the 20 droplets found after subtraction of the container

Plate 5a. Topical applicator for toxicity tests of solvents upon adult lice. (After Busvine).

Plate 3b. Close-up of topical applicator showing fine glass tubing secured to syringe needle with Araldite.



weight. A graph was constructed for the solvent plotting mass of the topically applied droplet against the number of micrometer divisions. Volume of topically applied solvent was calculated using the following equation.

After topical application, treated and control groups of lice were placed in an uncubator at 30°C and 75% relative humidity for 24 hours and then examined individually on the skin of the wrist under a microscope. These lice which failed to take blood from the skin were classed as dead and mortality rates were calculated using these results.

Complete test results were analysed by computer (CDC 7600 computer) using a Genstat package to fit data with a probit model (designed for bicassays) by using regression characteristics. Probit-log graph paper was used to plot dosage (mass/louse on logarithmic axis) versus corresponding mortality rate (percent mortality on probit axis) for the solvent under test and the computer analysis provided information to construct a regression line through the points to give the lowest possible chi² value (i.e. goodness of fit). The Genstat programme also estimated the LD₅₀ and LD₉₀ (i.e. dosage causing 50% and 90% mortality respectively) with associated fiducial limits, slope of the regression line with standard error and chi² value with number of degrees of freedom to obtain the P value. Bignificance was taken at the 5% lavel, which meant that dosage-mortality data resulting in P values lower than 0.05 were considered unreliable.

2.3.3.3. Topical application of potential insecticids solvents.

The 10 selected solvents (see Table 2) were topically applied in the pure form and at four different desage levels (ug/louse) for each solvent to give mortalities ranging from the low percentages (20%-50%) to almost 100%. Three replicate tests were performed with each chosen dose of solvent. Lauryl ether and dodecanol were supplied by Reed and Carnrick Pharmaceuticals of Kenilvorth, New Jerwey, U.S.A. for testing while the remaining solvents were obtained from Koob-Light Laboratories Ltd., of Colnbrook, Bucks, U.K.

 2.3.3.4. Topical application of aqueous insecticide solvant mixture.

The most toxic potential insecticide solvent, lauryl ather (see section 3.3.2.2.) was thoroughly mixed with distilled water in the ratio 1:9 (V/V) to form a 10% hosegenous suspension. Toxicity tests were performed with the suspension as described (see section 2.3.3.2.) except that a 50 pl Hamilton glass syringe

Table 3

Domage levels for topical application of 10% aqueous lawryl ether mixture.

Number of mioro- meter divisions	Estimated volume of 10% lauryl other mixture (µ1)
5	0.0416
10	0.0833
15	0.1250

It was not possible to accurately weigh droplets of the test mixture due to the rapid evaporation of the distilled water.

(Phase Separation Ltd., Queeneferry, Clwyd, U.K.) fitted with a glass meedle was used in the topical applicator to supply a larger volume range.

Three replicate tests were performed at such of the following dosage levels (Table 3) using the oriteria for death as described (see section 2.3.5.2.) after the 24 hour incubation period.

2.3.3.5. Topical application of insecticide (dieldrin) in potential insecticide solvent.

Using the most toxic and the least toxic potential insecticide molvents, lauryl other and 1, 4-butanediol respectively (see meetion 3.3.2.2.) solutions were prepared containing 0.04 g of dieldrin per ml (Shell Research Ltd., Sittingbourns, Kant, U.K.) and 0.004 g of dieldrin per ml. Topical application of these 4 solutions at a constant dosage volume of 0.005 µl (i.e. 2.5 micrometer divisions) gave calculated amounts of dieldrin as shown in the following table.

Three replicate tests were performed with both solvents at either dieldrin concentration and mortality rates determined as described (see section 2.3.5.2.).

Table 4

Dossge levels for topical application of disidrin in potential insecticide solvents

Insecticide	Applied volume	Dieldrin per louse	Dieldrin per louse
solvent	per louge	(at conc ⁿ 40mg ml ⁻¹)	(at conc ^h 4mg ml ⁻¹)
lauryl ether	0.005 ml	0.2 µg	0.02 pg
1,4-butanediol	0.005 ml	0.2 µg	0.02 µg

Table 4

Dosage levels for topical application of dieldrin in potential insecticide solvents

Insecticide	Applied volume	Dieldrin per louse	Dieldrin per louse
solvent	per louse	(at conc ⁿ 40mg ml ⁻¹)	(at conc ^b 4mg ml ⁻¹)
lauryl ether 1,4-butanedicl	0.005 ml	عبر 0.2 بير 0.2	0.02 µg

 2.3.4. Observation of outicular penetration by potential insecticide solvents using fluorescent dys (Ehodamine B).

Solutions were prepared with the most toxic potential insecticids solvents (lauryl ether and dodscanol) and the least toxic potential insecticide solvent (1, 4-butanedic)), each containing 0.5% (w/w) Ebodamine B (G.T. Gurr Ltd., London S.W.6.). Each solution was then used in the topical application of three separate batches of 10 femals adult lice, 2-5 days old (see section 2.3.3.1.) respectively at a dose rate of 0.011 µl per louse. Control insects were set up in an identical manner without the presence of Ebodamine B in the insecticide solvents. After 30 minutes in an incubator at 30° C and 7% relative humidity, three lice from each batch were vashed in a stream of acctone and map-frozen in isopentane cooled in liquid nitrogen in preparation for outling cryostat sections as described (see section 2.1.1.1.).

At 1 and 4 hours, dead lice were removed from the batches of lice treated with lauryl ether and dodecanol respectively, as well as 5 lice each from the untreated and 1,4-butanedicl treated batches and cryostat sections produced as above. The sections of control material, Rhodamine B treated material and the untreated material were examined using a Leits Dialux

microscope fitted with Rhodamine filter system (UV-blue incident illumination, 610 nm barrier filter) (E. Leitz, Luton, Beds., U.K.). Exposure for photomicrographs was set automatically with a Wild exposure unit coupled to the fluorescent microscope. 3. Results.

Examination of the structure of body louse outicle.
Light microscope (IM) examination.
Li. Examination of unstained outicle sections.

During these studies, two distinct types of outiols were identified at various locations on the body of the louse. Smooth, solerotised outiole (Flates 4 and 5) covered the head, thorax, appendages and paratergal plates whilet the abdominal integument was found to be rugose and membranous (Flate 6). Except for regions of modification where the integument was somewhat thicker much as various points on thorax and legs, outicals thickness was essentially uniform throughout (approximately 7 μ) irrespective of type. Tanning in the exocuticals and epicuticals of modification (Flate 4) which cocurred even more extensively in the thicker integument (Flate 5). Minimal tanning was observed in the membranous integument and was restricted to the region at the peaks of rugosities (Flate 6).

Fore canals were readily distinguished as strictions in solerotised outicle, particularly in areas of greater thickness, and followed relatively direct routes between the epidemis and epiouticle. In sections of membrancus outicle however, pore canals appeared to be discontinuous and such less ordered than those described above which suggested that they followed rather circuitous paths through the proouticle.

Flate 4. Unstained cryostat section of sclerite outicle with pore canals (arrows) appearing as continuous strations from epidermal layer (ed) to excouticle(ar) Note the tanning in the spiouticle(ar) and excouticle. Trapped air in some pore canals has caused them to appear darkened. en, endouticle. x 2300.

Flate 5. Unstained cryostat section of dorsal thoracic outicle showing extensive tanning and numerous pore canals following relatively direct paths. ep, epicuticle; sr, exceuticle; en, endouticle, x 2300.






3.1.1.2. Examination of stained outicle sections.

Longitudinal whole body sections stained with the nuclear staining dyes Hematorylin/Phloxine were particularly informative, revealing the locations within the body of the important structures (Flate 7). The head capsule and thorax of both seres were occupied primarily by musculature which was associated either with movement of mouthparts and antennae in the case of the head or of legs in the case of the thorax. The alignetary canal and reproductive organs were the dominant internal features in the abdomen, with the former becoming considerably distended during feeding to fully occupy the anterior half of this section of the body.

An examination of transverse body sections stained with Mallory's triple stain (Flates 6 and 9) confirmed these anatomical observations and provided histochemical information on the types of outicle present. In addition to the usual areas of head, thorax, legs and paratergites, solerotised outicle was also located in distinct strips across abdominal segments. These dorsal and ventral solerites, known as tergites and sternites respectively (Flates 8 and 9), were present in adults of both sames but were more extensive in males. In Plate 10 (paratergite outicle), the executicle and spicuticle were distinguished by their colouration due to tanning whilst the endocuticle, trawarsed by pore camals, remained unstained.

Unscience that accentuated the laminated character of the

- Plate 7. Whole body section of (fed) female adult louse stained with Hematoxylin/Phloxine. The advanced development of the eggs (ovarioles) gives some indication of the maturity of the insect.
 - 1 oibarial muscles
 - 2 lateral pharyngeal muscle
 - 3 antennal muscles
 - 4 coular lobe
 - 5 brain
 - 6 spirmole
 - 7 doreal lateral muscle
 - 8 vertical muscle
 - 9 notal pit
 - 10 salivary glands
 - 11 stomach (mid-gut)
 - 12 ovariole
 - 13 oviduct
 - 14 uterus
 - 15 accessory reproductive glands
 - 16 pylorus
 - 17 vagina
 - 18 gonopod



108 Plate 8. Transverse section through abdomsn of unfed female adult louse. Note the sclerotised cuticle (t, tergite) and ovariales (ov) at various stages of development. s. seta; sp. spiracle; m. musculature; g. gut. (Mallery's triple stain) x 100. Plate 9. Transverse section through abdoman (anterior) of engorged female louse. Membranous outicle has expanded whilst the solerctised cuticle (st, sterrite) remains rigid. bm, blood-meal; d vm, dorsoventral muscle; p p, paratergal plate; lm, lateral muscle. (Mallory's triple stain) x 100.



Plats 10. Solerotised outicle from an unfed male adult louse. Tenning is evident in epiouticle (ep) and executicle (ex) layers. The endoouticle is traversed by numerous pore canals but remains unstained by Malloxy's triple stain. ed. spidermis. x 100.

Flats 11. Unsclerotised outiols from abdomen of an unfed male adult louse. After staling with Mallory's triple staln, the exocutiols (ex, rendered ormage in the photomicrograph) is separated from new endowticle (en, blue) by an intermediate layer containing both colours.



structure. The red staining exocutiols (appearing orange-red in Plate 11) was separated from the blue-staining endocuticle by a mauve layer (mesocuticle) apparently containing both colours and denoting an intermediate region. Epicuticular tanning, if present, could not be distinguished because of the general level of staining which also comouflaged the pore canals.

Transmission electron microscope (TEM) examination.
1.2.1. Examination by TEM of cuticle sections.

Cuticle sections showed the multilaminate construction characteristic of arthropods in general. The sequence of layers in both types of cuticle, namely solerite and membranous, was identical with a thin epicuticle cutermost, overlying a relatively thicker exocuticle and endocuticle respectively, with an epidermal layer at the base.

The epiquticle

At least four layers were identified within the epiouticle of solerite and membranous outicle and the combined thickness of the layers varied between 0.1 and 0.2 μ . An electrondense membrane with an average thickness of 30 Å lay outermost and followed strictly the contours of the outicle surface. A 50 Å thick layer, the outer epicuticle, separated the outer membrane from the cuticulin and was completely electron-lucent (Plates 13 and 14). In the epicuticle of solerite cuticle, the cuticulin was rather ill-defined and, although extremely electron-dense at the boundary with the cuter epicuticle, it became less so on approaching the underlying dense homogeneous layer (Plate 14). However, the average thickness of the cuticulin in both cuticle types was estimated to be approximately 100 $\frac{1}{4}$ whilst the dense layer formed the remainder of the inner epicuticle thickness. Differentiation between dense layer and underlying excouticle was more marked in the membranous cuticle (Plate 17) than in scierite cuticle (Plate 13).

In some instances where the outicle surface had been relatively protected during embedding in Araldite, electron-dense deposits were observed in areas adjacent to the surface (Flates 13 and 16) and were probably the resmants of the surface wax layer or the comment layer or both.

A series of tubules running parallel to the surface of the outicle was located at the interface of the epicuticle and proouticle (Flate 17). In vertical sections the tubular network was quite distinct from the epicuticular filaments and appeared as a single row of electron-lucent holes which, in most instances, possessed electron-dense circumferences. Diameters ranged between 250 and 400 \hat{A} (Flate 17).

The exocuticle

The exocuticle thickness in solerites of adults varied between 1.0 and 1.4 μ , being composed of two distinct layers of approximately equal thickness (Flate 15). Both regions possessed a granular appearance; however, the innermost was the less electrondense of the two and was, in fact, the least electron-dense of all cuticle layers. The outer region of the sclerite exocuticle was dominated by a profusion of electron-dense filaments extending in both horizontal and vertical directions. In passing from the outer to the inner exocuticle, successive attachment among filaments caused a noticeable reduction in numbers until only single pore canals remained (Flate 15).

In the rugoes, membraneous cuticle, the executicle occupied a rather irregular volume bounded loosely by the epicuticle and the base of each rugosity (Flate 18). As with the solerite ercouticle, maximum thickness of the layer averaged 1.0 to 1.4 μ and vas also divided into two different regions (Plate 17). Both regions contained unidirectional, closely-packed macrofibrils arranged parallel to the longitudinal axis of each rugosity or corrugation. Macrofibrils in the outer region (average thickness 0.5 μ) were of medium electron-density and clearly defined by more dense cutlines which aided the measurement of macrofibril dismeters (250 - 300). This layer followed closely the contours of the outicle surface whilst the underlying inner excouticle occupied the remaining space above the outer endocuticle. Macrofibrils in the inner exocuticle were somewhat larger in diameter (i.e. 400 - 1000) and less densely packed than those described above (Flate 17).

The endocuticle

In scierite outicle, the endocuticle $(5.0 - 4.5 \mu \text{ thick})$ possessed a reasonably homogeneous granular appearance but did not exhibit the lamellate structure commonly encountered in arthropod outicle (Flates 15 and 19). Individual chitin crystallites were not resolved in the micrographs, but the uniform appearance of the endocuticle was consistent with that of unidirectionally oriented crystallites frequently observed in the endocuticle layer of many other arthropods.

The endocuticle of membranous outicle however, consisted of a well-defined immeliate structure (Plates 16, 17 and 18) formed by the helicoidal arrangement of 250-1000 Å diameter macrofibrils. Outer immeliae were generally much wider than those more recently daposited but all were composed of relatively few macrofibril layers which suggested that the angles between successive macrofibril planes were large. Deposition by the underlying spidermal layer often resulted in the lameliae being thrown into quite convoluted patterns (Plates 17 and 18) but the basic helicoidal arrangement was still preserved. The fusion of solerits (tergite) with membranous outicle shown in Plate 18 illustrated the gradual transition from the unidirectionally oriented chitin crystallites in the former to the macrofibrils with helicoidal arrangement in the latter. Endocuticle structure in membranous cuticle was also noted to be less closely-packed than in sclerite cuticle.

The spidermis

The epidermis underlying both solerite and membraneous cuticle consisted of a single layer of epidermal cells and was limited basally by the basement lamella. Dimensions of the epidermal layer were dependent upon developmental state and the location on the louse body such as a smooth layer under sclerite cuticle or a rugose layer under membranous cuticle; but the average thickness lay between 0.5 and 3.0 p (Plates 15, 16, 18 and 19). In sclerite integument the spical cellular surface of the epidermis appeared to be separated from the procuticle by a granular layer and a plasma membrane respectively (Plate 19) though this was not so readily apparent in membranous integument because of its convoluted nature (Plate 18). Each spidermal cell contained a large nucleus with a prominent nucleolus whilst the remaining cytoplass consisted primarily of densely-packed free ribosomes and rough and smooth endoplasmic reticulum in addition to numerous mitochondria (Plates 15, 16 and 19). Desmoscones were observed forming the lateral attachments between adjacent cells as well as hemidesmosomes performing a similar function between cells and the basement lamella (Plate 19).

The pore canal system

Pore canals with diameters from 0.1 to 0.2 μ were observed traversing the outicle from the spical epidermal surface to the inner exocuticle (Flates 15, 16 and 19). Earlier observations which indicated that the paths of the pore canals through solerite

endocuticle were more direct in comparison to those through membranous endocuticle (see section 3.1.1.) were confirmed conclusively at the ultrastructural level (Flates 15 and 16). In both types of integument the half of the pore canals proximal to the epidermis appeared as electron-lucent structures bounded by a cytoplasmic membrane approximately 100 Å thick and containing one or more filaments of about 200 Å diameter (Flates 15, 18 and 19). It was noted that the filaments entered the epidermal layer (Flate 19) and this was particularly evident in integument from immature adults (Flate 16). It could not be ascertained whether or not they actually entered the epidermal cells.

The distal half of the pore canals showed heavy staining by ognium tetroxide, including both the cytoplasmic membrane and the internal contents (Plates 15 and 18). Some minor branching was observed in the outer endocuticle but most occurred in the exocuticle and with such frequency that the system in this layer was more filamentous than canal-like (Plates 15, 17 and 18). The exocuticle of solerites contained significantly greater numbers of filaments than that of membranous outicle (Plates 15 and 17) and this disparity also existed in the epicuticular layers of both types of cuticle. Below the layer containing the tubular network, the filaments followed both vertical and lateral routes through the exocuticle; however, all the epicuticular filaments above the tubular network traversed the epicutice normal to the surface (Flates 14 and 17). Each epicuticular filament was

Plate 12. Paratergal plate (pu) on abdomen of female adult iouse (2-5 days old) showing junction of solarite outiols with membranous outiols. Arrow indicates location of outicle in Flate 13. x 350.

118

Plate 15. Sciencie cuticle from area shown in Flate 12. Deposite (d) of wax and/or coment layer are seen in a state of gradual removal adjacent to the outicle surface. sp. spicuticle; ex, excouticle. x 32000.



Plats 14. Solerits outlos from abdomen of male adult louse (2-5 days old). Epicuticular filaments (ef) can be seen connected with the surface vis pores. OH, outer membrane; ce, outer epicuticle; c, cuticulin; dl, dense layer; ex, excouticle. x 126000.

Plate 15. Scierite cuticle from abdomen of female adult louse (2-3 days old). The homogeneous granular endocuticle (en) appears to be non lameliate and is travermed by pore scale (pc) passing directly through the matrix. oar, outer exocuticle; isr, innor excouticle; en, andocuticle; gam, apical plasma membrane; ed, epidermis; n. nucleus; bm, basemont lamella. x 10600.



Plate 16. Membranous outicle from the abdomen of postecdysial female adult. The excouticle is large in comparison with the endoouticle (composed of only 3 immellas). Pore canals appear to penetrate the spidermis. sp; epicutole; ex, excouticle; en, endoouticle; ed, epidermis; hm, basement lamella; m, muncle. x 1500.

Plate 17. Membranous outicle from abdomen of female adult (2-5 days). The tubular network is visible underlying the epiouticle and the macrofibrillar nature of the exnouticle/ endocuticle layers is also evident. ep, epiouticle; in, tubular network; ce, outer exocuticle; i einner exocuticle; en, endocuticle; i einner exocuticle; en, endocuticle; po, pore canal. x 25000.



Plate 18. Junction of sclerite/membranous outicle on abdomen of male adult louse (2-3 days) showing fusion of helicoidal macrofibrils in membranous endocuticle with unidirectional microfibrils in sclerite endocuticle. ce, outer exocuticle; i e inner excoutiole; pc, pore canal; en, endocuticle; ed, epidermis; bm, basement lamella. x 8400.

Plate 19. Sclerite abdominal integument from female adult louse (2-3 days) showing outicle deposition some louse (2-) days) moving outloss comparison to adjacent to apical planma membranes of epidemis. po, pore canal, agm, apical planma membrane; fr, free ribecomes (and rough/mmooth KPR); cj, call junction; bm, basement lambla; m, mitochondrion; hd, hemideamonome. x 42000.



about 100 Å in diameter and linked with the surface of the outiols via a small depression or pore which was approximately $100 \overset{\circ}{}$ at its base (Plate 14).

3.1.2.2. Platinum/Carbon (Pt/C) replication of outicular surface and cross section.

The outicular surface was found to have a covering of fairly evenly spaced small mounds, each approximately 0.1 p in diameter at the base (Flate 20). On membranous outicle, the mounds occurred at the rate of 80-90 per square micron which was consistent with the occurrence of the epicuticular filements in the corresponding TEM sections (i.e. 8-10 epicuticular filements per micron). A similar situation was observed for solerits outicle except the mounds were found to be approximately three times as numerous which resulted in a more uniform appearance. That the mounds were part of the surface was bloom was demonstrated by the fact that whole sections of the layer were stripped from the outicular surface during freeze-fracturing (Flate 21).

Examination of oblique cross sections of membranous integument (Plate 21) confirmed that the macrofibrils of the exocuticle were oriented unidirectionally along the length of the rugosities. In the underlying endocuticle, the macrofibrils appeared to form a lamellate structure which was consistent with the observations made during the TEM studies (see section 3.1.1.1.). The angular difference in orientation between two successive layers of smoorofibril

Plate 20. Surface replica of membranous abdominal outicle from female adult louse (2-5 days) demonstrating the wax bloom formed by extruded outicular war, x 22000.

Plate 21. Raplice of surface and oblique oross section of membranous outiols from male adult louse. Outioniar was has been stripped away in sheets to rewail a smooth surface. Pore canals spear as boles (arrows) throughout the andoutiols. os, outiols surface; ex, exocutiols; en, endooutiols. x 4000.



Flate 22. Esplica of an oblique cross section through endocutiols of membranous outicle from female adult louse (2-3 days). Two adjacent places of macrofibrils can be seen (arrows) separated by approximately 45". x 40 000.

Plate 25. Esplica of oblique cross section through endocutiols/spidermis region of sembranous outlide from famals adult louss (2-3 days). Circular holes indicate pore canals following vertical routes while elongated holes or chancels result from canals following lateral routes. en, endocutiols; ed, epidermis. x 10 000.



planes in the endocuticle was about 45⁰ (Plate 22). Many pore canals, particularly in the outer half of the endocuticle, were out obliquely during the fracturing procedure and produced elongated holes (Plate 21) whilet those produced by pore canals in the endocuticle adjacent to the epidermis were irregular (Plate 23).

3.1.3. Scanning electron microscope (SEM) examination.

3.1.3.1. Morphology and surface topography of the adult louse.

The anterior extremity of the head was formed by a membranous protuberance (haustellum) surrounding the mouth opening (Plate 26) through which the internal mouthparts were retracted when not feeding. Two large areas of solerite integument (clypeus and coular-antennal segment) separated by arthrodial integument (clypeofrontal suture) completely covered the dormal surface of the head and were extended ventrally to some degree (Plate 26). A single, well-developed eye was located on the ocular lobe posterior to each antenna (Plate 26) which, in turn contained five segments. Feg organs or sensilla basiconice (Torre-Euneo, 1962) were observed on the extremity of the fifth segment of each antenna whilst tuft organs or sensilla coeloconice (Torre-Euneo, 1962) were restricted to the fourth and fifth segments (Flate 26). Taotile hairs were observed on each antennal segment (Plate 26) and in fact, occurred extensively on other parts of the body. The dormal surface of the thorax displayed only slight segmentation as a result of the fusion of the pleurites or subcoxes (Flates 24 and 27). Whilst the metathorax had almost entirely disappeared, the notum was reduced to a narrow, membranous axisistrip encompassing the notal pit. The one pair of spiracles present on the thorax were located on the mesothoracic subcoxes and were totally isolated by sclerotised outicle (Flates 25 and 27). Ventrally, the thorax consisted essentially of an enlarged, sclerotised sternal plate which showed no apparent segmentation. Lags were highly modified orab-like structures especially adapted for clasping hairs and fibres etc. Each leg possessed a tible with extended thumb coupled to a singlejointed tareous bearing a tareal claw (Flate 25).

Some reduction in abdominal segments was observed at the anterior and posterior ends of the adult louss which resulted in only 7 segments being readily apparent (plate 24). The membranous integument which covered most of the abdomen was of a rugose nature and existed in two forms (Plates 27, 28 and 29). One type (Plate 28) consisted of long corrugations whilst the other (Plate 29) seemed to be merely an attenuated version of the first and axhibited a 'quilted' appearance. Solerite cuticle on the abdomen was restricted to a series of plates (paratergal plates) at the lateral margins of each of the first 6 segments (Plates 24 and 25) as well as much smaller strips on the dormal and ventral surfaces of each segment. Each paratergal plate bore a single





135 Plate 26. Close-up of female adult louse (Plate 24). The mouthparts have been retracted into the head. h, haustellum; 1, labrum; c, clypeus, cs, clypeo-frontal suture; th, tactile hair; sb, sansilum basiconica; so, sensillum cosloconica; as, antennal segment; e, eye; ol, ocular lobe. x 100. Plate 27. Thorax of male adult louse (2-3 days) showing close fusion of the three segments. The notal pit is surrounded by membranous outicle of the notum, itself almost entirely surrounded by solerite cuticle. n, notum; np, notal pit; sp, spiracle; ap, abdominal pits. x 100.





Plate 28. Nambranous abdominal outicle from male adult louse (2-5 days). Distinct patches occur in which rugositise possess serrations on their posterior edges and are found on the dormal and ventral surfaces of each segment. x 600.

Plate 29. Membranous abdominal outicle from female adult louse (2-5 days) showing individual pits which form the series found on dormal and ventral surfaces of both series. x 650.


Plate 30. Membranous outicle from abdomen of male adult louse, 2-3 days old. The surface war layer has been severely disrupted as a result of the SEM preparation procedure and the war bloom is not evident. x 3500.

Plate 31. Solerite outicle from thorax of female adult louse (2-3 days) showing the distribution of the surface wax which has occurred as a result of preparation for SDM examination, x 2000.



Plate 32. Mambranous cuticle from abdomen of male adult louse, 2-3 days old. Surface war has been dissolved during the SSM preparation procedure and deposited at a later stage as an irregular film. x 60 000.

141

Plate 33. Solarite cuticle from thorax of female adult louse (2-3 days) on which surface pores (arrows) can be seen through the irregular wax film. x 60 000.







spirable (Flate 25) displaying the distinctive muchroom shape characteristic of <u>Pediculus</u>. Dormal and ventral segments of the abdomen possessed two elongated regions where the posterior edges of the rugosities were serrated (Flate 28) and pits (Flate 29) were also observed along the outer boundaries of the segments towards the paratergal plates forming two continuous errise on both abdominal surfaces (Flates 24, 25 and 27).

The outicular wax layer on both membranous (Flates 30 and 32) and solerite (Flates 31 and 33) integument experienced severe disruption during the SEM preparation procedure despite initial omnium fixation. Unlike the regular appearance exhibited by the extruded wax mounds in platinum/carbon replicas (Flate 20), the cuticular wax was spread over the surfaces as an even film (Flates 31, 32 and 33) as a result of dissolution during critical point drying. On close examination of solerite surfaces, there appeared to be epicuticular porces concealed beneath the dispersed wax film (Flate 33) and were more clearly seen in areas where the wax had been completely removed (Flate 34).

3.1.3.2. Examination of cuticle oross section.

Unfixed cryostat sections examined by the SIGM after critical point drying showed severe structural alteration throughout the whole area of cross section (Flate 35) and was probably caused by dehydration immediately after sectioning. Consequently, the

camples prepared in this manner proved unsatisfactory for use in the study of outicular ultrastructure. Although prior fixation of cuticle samples successfully preserved basic structural integrity, it was found that deformation occurred during sectioning (Flate 36) and these specimens were also unsuitable for further study.

It was possible however, to examine cross sectional formations using whole lies which were simply torm apart after prior essim tetroxids fixation and oritical point drying. The fracture surface produced in this way indicated that the solarite cuticle failure occourted in a brittle manner (Plates 57 and 38). Fore canals were readily distinguished throughout the solarite outicle orcess section and structures consistent with those of pore canal filaments were observed within them (Plate 38). In contrast with the results from the TRM studies (see section 3.1.2.1.), the fracture surfaces of solarite cuticle provided evidence that it did, in fact, possess a layered structure which showed signe of being composed of fibrils (Plate 37).

Membranous outicle fractured in an entirely different fashion. The outer layers (epicuticle and exocuticle) failed in a brittle manner whilst the underlying endocuticle, composed of macrofibrils, gave some indication of plastic deformation (Plate 39) despite the hardening effects of the combium fixation. Fracture surfaces through spicuticle and exocuticle layers of membranous cuticle (Plate 39) and all layers of sclerite cuticle (Plate 38) were

Plate 35. Cryostat section of unfixed sclerits which was then oritical point dried. Severe structural deformation is evident throughout the area of cross section probably due to uncontrolled dehydration prior to critical point drying. x 10 000.

147

Flate 36. Cryostat section of fixed (casium tetroxids) membranous outfole which was then oritical point dried. Structural deformation has been caused during sectioning by the microteme blade. x 8000.



Plate 37. Fracture surface of sclerite outicle produced by tearing outicle previously fixed and oritical point dried. The endouticle has a distinct lamellate appearance and there is some indication that it is fibrous (arrow). x 15 000.

Plate 38. Practure surface of scientic outicle prepared as in Plate 37. Pore canals and their filaments are visible and follow relatively direct routes through the endocuticle. x 20 000.



Plate 39. Cross saction of membranous outicle prepared as in Plate 37. The macrofibrillar structure of the endocuticle contrasts with the closely-packed exocuticle and epicuticle. ep. epicuticle; ex. encouticle; en. saccuticle. x 10 000.

151

Plate 40. Cross section of membranous outicle prepared as in Plate 37. The series of holes parallel to the surface and underlying the spicuticle corresponds to the tubular network seen in TEM studies. The strictions or channels above the tubular network correspond to epicuticular filaments. x 50 000.



quite similar in appearance and this was consistent with the closely-packed structure of the material involved. The individual macrofibrils comprising the endocuticle of membranous integument were discernible (Plates 39 and 40) and their dimensions (400 - 1000 Å) coincided with those estimated from the TEM examination (see section 3.1.2.1.). As observed during the platinum/carbon replication studies (see section 3.1.2.2.), the numbers of macrofibril layers between each lamella was relatively small as a result of successive layers twisting approximately 45° from the next to form a helicoidal arrangement (Plate 39). The tubules located by TEM (see section 3.1.2.1.) at the interface of epicuticle and produticle were also observed during the SEM examination of sclerite and membranous integument and appeared as a series of holes forming a line parallel to the cuticular surface (Plate 40). Narrow canals were observed traversing the epicuticle prependicularly to the surface (Flate 40) and corresponded to the spicutioular filaments noted previously (see section 3.1.2.1.).

3.1.3.3. Solvent effects on the cuticular wax layer.

Cuticle which was subjected to acetons washing prior to preparation for SEM examination exhibited an extensive loss of cuticular war mounds from the surface (Flate 41) when compared to the appearance of the untreated outicular surface (platinum/ carbon replication - see section 3.1.2.2.). Despite the removal of many war mounds, the epicuticular pores remained Plate 41. The structure of sclerite outicle which was acetomewashed (10 minutes) prior to fixation (OsO₄) and oritical point drying. Epicuticular pores are concealed by wax mounds and a web-like film of wax deposits. x 50 000.

Plate 42. Surface of solerite outicle which was became-washed (10 minutes) prior to firstion (Sed4) and oritical point drying. Was mounds have been resourced and epicutioular pores (arrows) are visible through the remaining debris. x 30 000.



156 Flate 43. Surface of hexane-washed solerite outicls as in Flate 42. Bplouticular pores or depressions have been revealed by the complete removal of outicular way by the solvent. x 60 000.



concealed by a web-like wax film deposited on the surface (Flate 41).

Herane acted much more efficiently in dissolving and removing cuticular war (Plates 42 and 43) so that only insoluble debris remained after washing for 10 minutes (Plate 42). In areas af complete war removal, surface pores or depressions approximately 100 \mathring{A} in diameter were clearly seen (Plate 43) which corresponded both in appearance and numerically with the epicuticular structures observed previously in TEM studies (see section 5,1,2,1,).

3.1.3.4. Effects of alkali treatment on the surface.

Although some cuticular was had undoubtedly been removed from the surface, the greater percentage had collected together in large mounds (Plate 44) which were substantially larger than those noted earlier (see section 3.1.2.2.). Removal of wax during alkali treatment was brought about by saponification, but an analysis of the cuticular was revealed that it was composed of over 75% hydrocarbons which do not respond to such action (see section 3.2.3.2.).

Although the oblight structure of the cuticle was unaffected by the strong alkali, the epidermal layer was completely removed by digestion and the basal openings of the pore canals were revealed (Plate 45). Flate 44. Outer surface of selerite outicle on which the unseponified cuticular wax has collected into large mounds after treatment with alkali (KON). The electron micrograph also shows the surface duct of a dermal gland. x 3500.

159

Plate 45. Inner sndocuticle surface of sembranous outicle after alkali (KNE) treatment during which the epidermal layer was digested. Large basal openings to pore canals are visible throughout the snitre surface. x 5000.



3.2. Chamical Composition of Louse Lipida.

3.2.1. Comparison of potential extraction solvents.

3.2.1.1. Trial thin-layer chromatography (TLC) separations.

Visualisation with iodine vapour of the TLC separations of the solvent outioular extracts produced positive results for the hexame extract only. These appeared as two faint spots near the solvent front. After sulphuric acid and heat treatment, the two spots became more intense and two more spots (fainter) appeared at distances closer to the origin. Observation of the plate under UV radiation revealed corresponding but much fainter spots produced by the benzene and chloroform extracts while the dimethoxymethame extract produced faint spots at and near the origin which were tentatively identified as phospholipids and cerebroatides.

3.2.1.2. Solubility of long chain hydrocarbon $(n - C_{\frac{3}{5}6})$ in potential extraction solvents.

After a period of 10 minutes at room temperature, the beaatriscontane had dissolved in the beame only. With mild heating of the remaining tubes, the benatriacontane dissolved in both the benzene and chloroform.

5.2. Chemical Composition of Louse Lipids.

3.2.1. Comparison of potential extraction solvents.

3.2.1.1. Trial thin-layer chromatography (TLC) separations.

Visualisation with indime vapour of the TLC separations of the solvent outicular extracts produced positive results for the hexame extract only. These appeared as two faint spots near the solvent front. After sulphuric acid and heat treatment, the two spots became more intense and two more spots (fainter) appeared at distances closer to the origin. Observation of the plate under UV radiation revealed corresponding but much fainter spots produced by the benzene and obloroform extracts while the dimethoxymethane extract produced faint spots at and near the origin which were tentatively identified as phospholipids and cerebrosides.

3.2.1.2. Solubility of long chain hydrocarbon $(n - C_{36})$ in potential extraction solvents.

After a period of 10 minutes at room temperature, the hexatriacontane had dissolved in the hexane only. With mild heating of the remaining tubes, the hexatriacontane dissolved in both the benzene and chloroform.

3.2.2. Extraction of louse lipids.

3.2.2.1. Cuticular lipid yields from extractions of whole lice.

Table 5 shows a comparison of both fresh and dry body weights of male and female adult lice. According to the results, the adult female was found to be significantly heavier than the corresponding male (46.96% heavier) and this relationship was maintained when dry weights were compared (48.56% heavier).

The weight of the female adult louse decreased by an average of 66.12% upon drying whilst that of the male adult decreased by an average of 66.46%, illustrating the high water content of the adult lice.

A quantitative comparison of the wax yields from hexane extractions is shown in Table 6 . On average, the femals adult yielded approximately 20% more wax through hexans extraction than the male adult. However, when compared as percentages of body weights (both fresh weights and dry weights), the average yield of cuticular wax from the male adult exceeded that of the female adult by approximately 20%.

Table 5

Comparison of fresh and dried adult weights

Female lice (adult, 2 - 3 days)			Mals lice (adult, 2 - 3 days)		
numbers of lice	fresh weight mg/louse	dried weight mg/louse	numbers of lice	fresh weight mg/louse	dried weight mg/louse
200	1.779	0.569	200	1.125	0.365
200	1.659	0.550	200	1.157	0.322
540	1.299	0.445	759	0.908	0.307
435	1.411	0.519	416	0.993	0.406
Average weight mg/louse	1.537	0.521	Average weight mg/louse	1.046	0.350

Table 6

Comparison of cuticular wax yields

numbers and sex of lice	mass of extracted wax mg	per louse	wax as % of fresh weight	wan as % of dry weight
435 💡	0.10	0.23	0.0163	0.0488
540 9	0.11	0.20	0.0157	0.0458
416 đ	0.08	0.19	0.0193	0.0580
759 đ	0.14	0.17	0.0188	0.0556

3.2.2.2. Examination of outicular lipid extraction technique (from whole lice).

Sampling of blood from the site of lice feeding on the rabbit indicated a maximum level of 275 counts/40 µl blood specimen (Table 7). The ratio of sointillations from the extracted outicular vax and labelled cleic acid to the scintillations from the total labelled cleic acid consumed by the lice represented the fraction of internal lipids removed simultaneously with the cuticular lipids during a 10 minute extraction period.

i.e. internal lipids removed during outicular lipid extraction

total internal lipids

= cpm (cuticular lipids + internal labelled cleic acid)

cpm (remaining internal labelled oleic moid + labelled oleic acid extracted with cuticular lipids)

= 1272 - background

46270 - background + 1272 - background

= <u>1247</u> 47492

= 0.0262

= 2.62%

The 10 minute extraotion of the lice colony (approximately 10,000 lice) produced 1.5 mg of cuticular wax which was equivalent to 0.15 µg per insect (see section 3.2.2.1.).

Table 7

Scintillation counts on labelled/unlabelled samples

Sample	Scintillations cpm ⁸
blood ^b (unlabelled)	21
blood (after lice feeding, 20 minutes)	275
cuticular lipids + internal labelled cleic acid	1272
remaining internal labelled cleic acid	46270
background	25

- average scintillation counts per minute from 5 observations.
- b. blood serum 40 µl.

- Thin-layer chromatographic (TLC) analysis of lipid extracts.
- 3.2.3.1. Qualitative TLC analysis and comparison of cuticular lipids from whole lice and cast skins.

Detection with iodine vapour of the outioular lipids from whole lice revealed the presence of six lipid classes (Skipski <u>et al.</u> 1965) and the speed and intensity with which the hydrocarbon spot appeared, indicated some unsaturation. Bulphuric acid and heat treatment did not reveal further spots (Flate 46). During heating, two of the spots first appeared red before turning black which was characteristic of the behaviour of oholesterol and cholesteryl esters (Mangold, 1965). The six lipid classes were tentatively identified by comparison with authentic standards chromatographed simultaneously on the plate and are listed below in decreasing order of intensity.

- i hydrocarbona
- ii cholesteryl esters
- iii triglycerides
- iv 1. 3 diglycerides
- v cholesterol
- vi fatty acids.

Eight lipid classes were detected with iodine vapour in the outicular lipid extract from cast skins and further treatment with sulphuric acid and heat did not reveal any additional classes. The spots appeared in the following decreasing order of intensity.



Plate 46. TLC separation of cuticular louse lipids.

- a,b louse lipids
- c reference standards
 - 1. hexatriacontane
 - 2. cholesteryl stearate
 - 3. methyl stearate
 - 4. triolein
 - 5. cleic acid
 - 6. 1,3-diolein
 - 7. 1,2-diolein
 - 8, cholesterol
 - 9. 1- and 2-monoolein

- i phospholipids/cerebrosides
- ii fatty acids
- iii cholesteryl esters
- iv cholesterol
- v hydrocarbons
- vi 1 and 2 monoglycerides
- vii triglycerides
- viii 1, 3 diglycarides.

5.2.3.2. Quantitative TLC analysis (by weight) of cuticular lipids.

Eydrocarbons were found to constitute almost threequarters by weight of the louse outioular lipids (Table 6). Cholesteryl esters formed the second most abundant lipid class but at approximately one-tenth the weight of the hydrocarbons. The occurrence of the remaining lipids i.e. triglycerides, 1, 3 - diglycerides, cholesterol and fatty acids was substantially lower, between 5.79% (triglycerides) and 5.87% (fatty acids).

Table 8

Composition of outicular louse lipids (by weight)

Lipid	Mass ng		Percentage of Total Lipida		Average	
Clame	Replicate 1	Replicate 2	Replicate 1	Replicate 2	Fercentage	
hydrocarbons	2.700	4.210	72.835	74.712	73.77	
cholesteryl esters	0.259	0.463	6.987	8,216	7.60	
triglyceridea	0.237	0.292	6.393	5.182	5.79	
fatty acids	0.162	0.198	4.370	3.372	3.87	
1, 3-diglycerides	0.201	0.223	5.422	3.957	4.69	
cholesterol	0.148	0.249	3.992	4.419	4.21	
Total	3.707	5.635	99.999	99.858	99-93	

3.2.3.3. Quantitative TLC analysis (by densitometer) of outicular lipids/internal lipids/fascal lipids.

The TLC separations of the various louse lipids (Plate 47) were scanned with a densitometer and the percent composition estimated by comparing weights of the chromatogram peaks cut from the recorder trace (Table 9). The concentration of hydrocarbons was highest in the cuticular lipids (55.2%) and relatively high in the whols body (17.1%). Cholesteryl esters constituted the most abundant lipid class in fascal lipids (59.2%) while the levels in cuticular lipids (13.7%) and whole body (17.7%) were quite similar.

The concentrations of triglycerides and fatty acids in all three samples were relatively low except for the fatty acids of the whole body lipids (40. %) which comprised the predominant lipid class of that mixture.

Levels of 1, 3 - diglycerides and monoglycerides throughout were low while cholesterol levels were comparatively high, i.e. cuticular wax (13.2%), fasces (19.1%) and whole body (13.2%).

PAGINATION ERROR









a,e	- o internal lipids
ъ	- o + 🗣 faecal lipids
с	- J + 9 cuticular lipida
d	- 💡 internal lipids
f	- 0+9 faecal lipids
Composition of louse lipids (using TLC - densitometer)

Lipid source	Lipid class	% composition of total
extracted outioular wax	hydrocarbons cholesteryl estera triglycerides fatty acids 1,3-diglycerides cholesterol	55-2 13-7 11.6 1.9 4.4 13.2
faeces	hydrocarhons cholesteryl esters triglycerides fatty solds cholesterol monoglycerides	6.4 59.2 3.7 9.9 19.1 1.7
whole body	hydrocarbons cholestaryl satars triglycerides fatty acids 1,5-diglycerides cholesterol monoglycerides	17.1 17.7 7.2 40.2 1.5 13.2 3.1

3.2.3.4. TLC analysis of digestion products of labelled fatty acid (olsic moid).

The hydrocarbons (73.0%) possessed the highest level of radioactivity of the lipid classes of the cuticular wax (Table 10) and also that of the female whole body (36.5%) but the level was much lower in the fascal matter (6.2%). The value for the cholesteryl esters (73.6%) in the fasces was similar to that of the hydrocarbons in the cuticular wax, while it was also relatively high in the whole body (19.8%) but low in the cuticular wax (5.42%). The radioactive levels for other lipid classes (i.e. triglycerides, fatty acids, diglycerides, cholesterol and monoglycerides) were not significantly high (Table 10) with the exception of fatty acids in the whole body (29.6%).

Using the sointillation data it was possible to calculate the mean volume of blood taken per insect during the 10 minute feeding period.

Assuming virtually all the radioactivity was recovered from the 20 fed sdult lice, then

sointillation (op10m) from outicular lipids = 162.4
sointillation (op10m) from fascal lipids = 245.2
sointillation (op10m) from whole body lipids = 458.6
Total sointillation (op10m) = 866.2

scintillation (cpm) from blood (40 µl) during feeding = 238(mean) scintillation (cp10m) from blood (1 µl) during feeding = <u>2380</u> 40

Total blood volume (µ1) taken by 20 insects = $\frac{866.2 \times 40}{2380}$

Average blood volume per insect = $\frac{866.2 \times 40}{20 \times 2360} = 0.7278 \ \mu l$

This result was compared with that obtained from weighing a known number of lice (120) before and after feeding for 10 minutes. The density of the blood (1.0698 g ml⁻¹) was calculated by weighing a known volume of blood (50 µl) from the rabbit on which the lice were fed.

mass of container with 120 unfed lice = 13.2448 g mass of container with 120 fed lice = 13.3416 g mass of blood ingested by 120 lice = 0.0968 g mass of blood ingested by 1 louse(mean)= 0.000806g

v

Prom the scintillation data it was also possible to estimate more accurately the numbers of lice used in the original experiment to examine the extraction technique (see section 2.2.6.1.).

Table 10

Lipid source	Ligid class	Sointillation c p 10m [®]	% composition of total
	hydrocarbons	193.6	73.03
extracted	cholesteryl esters	83.8	5.42
outicular	triglycerides	92.6	10.83
VAX	fatty acids	86.6	7.14
	cholesterol	80.8	3.57
	hydrocarbons	90.4	6,28
	cholesteryl esters	256.0	73.82
facos	triglycerides	86.6	4.73
	fatty acids	97.8	9.30
	cholesterol	89-4	5.87
	hydrocarbons	242.8	36.59
	cholesteryl esters	166.2	19.68
whole	triglycerides	91.2	3.53
body	fatty acids	211.0	29.65
(female)	1,3 diglycerides	89.0	3.05
	cholesterol	94.0	4.14
	monoglycerides	89.4	3.14
			1

Metabolic products of tritiated oleic aoid

average scintillation counts per 10 minutes from 5 observations .

calculated after subtraction of background (75 cp10m) ъ

From above:

Total sointillation (op10m) from 20 lice = 866.2 From 2.2.6.1. Total scintillation (opm) from x lice = 47492 Total scintillation (op10m) from x lice = 474920 x (numbers of lice used) = 20 x 47492 1 x 866.2

= 10,965 lice

3.2.3.5. Degree of saturation of cuticular hydrocarbons.

The hydrocarbons separated into two distinct spots or corresponding with the saturated and mono-saturated reference spots, hexatriacontans and 1 - octadecens respectively. From these results, both saturated and unsaturated hydrocarbons appeared to be present in the cuticular lipids and the unsaturated hydrocarbons were mono-unsaturated.

 Gas-liquid chromatographic (GLC) analysis of lipid extracts.

3.2.4.1. GLC analysis of cuticular hydrocarbons.

The chromatograms produced by the suticular hydrocarbons from adult lice separated according to sex were identical (GLC peak numbers 1 to 36) and contained four major peaks which gave the appearance of forming a series (Fig. 6). Proportions and retention times of the corresponding minor GLC peaks from both samples coincided exactly. Separation of saturated from unsaturated hydrocarbons (see section 2.2.4.3.) indicated that the cuticular louse mixture contained eight mono-unsaturated hydrocarbons (peak nos. 4, 9, 15, 18, 21, 24, 27 and 33 all absent in Fig. 7). Removal of straight chain hydrocarbons from the total hydrocarbons with molecular sieve 54 (Fig. 8) enabled the structural character of the hydrocarbons to be determined i.e. whether branched or unbranched, while co-chromatography with n - alkane standards confirmed many of the identifications. A linear plot of retention time versus carbon numbers revealed the presence of three series of hydrocarbons; & - normal alkanes, B - normal alkanes and C tentatively identified as mono methylalkanes (Fig. 9).

As the subsequent tests proved, one major component (peak no. 27) was mono-unsaturated (series B) and did not form a







components_







series with the three other major peaks (nos. 10, 16, 22) which were identified as n - alkanes (series Å). GLC analysis following the periodate-permanganate oxidation (see section 2.2.4.6.) of the unsaturated hydrocarbone produced two main fatty acid methyl ester peaks which contained 15 and 16 carbons in the fatty acid portion respectively. This result agreed with the reaction products expected of peak 27 which appeared to contain 31 carbons in a straight chain. Unfortunately, no other peaks of significance appeared so it was impossible to deduce the structure of the second most abundant alkene (peak no. 21).

Table 11 (page 186) shows the completed analysis of the hydrocarbons in which the methylalkane structures were determined from their mass spectra as well as retention times.

3.2.4.2. Gas chromatography/Mass spectroscopy (GC/MS) analysis of cuticular hydrocarbons.

In each case, the identity of the n - alkanes, as datarmined by GLC techniques (see section 2.2.4.2.) was confirmed by the results of the mass spectral analysis. The mass spectra obtained with Electron Ionisation(E.I.) was more readily interpreted than that obtained with Chemical Ionisation (C.I.). This was due, in part, to the lower sensitivity at which the system was operated for the C.I. mode.





	1	T			
Peak No.	E.C.L.	Series	Mean % Composition		Hydrocarbons
			Hydrocarbons	Vax	
1	21.0		Tp	т	n - hen sicosane
2	22.0		т	т	n - docosane
3	22.7	в	T	т	unsaturated n-tricosane
4	23.0		2.0	1.8	n - tricosane
5	23.3	с	0.5	0.2	11 - methyltricosane
6	23.7	-	т	т	Unknown
7	24.0		0.9	0.6	n - tetracosane
6	24.3	-	Т	T	Unknown
9	24.6	в	0.2	0.1	Unsaturated n-pentacosane
10	25.0		14.0	10.3	n - pentacosane
11	25.3	С	1.2	0.9	11 - methylpentacosane + 13 - methylpentacosane
12	25,6	D	0.3	0.2	11, 15-dimethylpentacosane
13	26.0		1.0	0.7	n - hexacosane
14	26.3	-	T	т	Unknown
15	26.6	в	0.8	0.6	Unsaturated n - heptacosan
16	27.0	A	15.1	11.1	n - heptaccaane
17	27.3	c	1-7	1.2	11-methylheptscomme + 13-methylheptscomme
18	27.6	в	0.7	0.5	Unsaturated n-octacosane
19	28.0	A	1.7	1.2	n - octacosane
20	28.3	-	0.5	0.4	Unknown
21	28.7	в	7.8	5.7	Unsaturated n-nonacceane
22	29.0	A	15.6	11.5	n - nonacosane

Peak No.	E.C.L.	Series Mean % Composition		ition	Hydrocarbons
_			Hydrocarbons	Vax	
23	29.3	с	4.9	3.6	7 - methylnonacosane + 9 - methylnonacosane + 11 - methylnonacosane
24	29.7	в	2.4	1.8	Unsaturated n-triacontane + unknown
25	30.0		0.5	0.4	n - triacontane
26	30.3	-	0.4	0.3	Unknown
27	30.8	в	15-4	11.3	15 - hentriacontene
28	31.0		1.0	0.8	n - hentriacontane
29	31.3	c	3.3	2.4	9 - methylhentriacontane + 11-methylhentriacontane + 13-methylhentriacontane
30	31.6	D	3.7	2.7	9,19-dimethylhentriacontane + 11,21-dimethylhentria- contane
31	32.0		т	т	n - dotriacontane
32	32.3	-	0.5	0.4	Unknown
33	32.7	в	0.8	0.6	Unsaturated n -tritriacon- tane
34	33.0		т	т	n - tritriacontane
35	33.3	с	0.7	0.5	15 - methyltritriacontane
36	33.6	D	0.7	0.5	Possible dimethylalkane

a-R.C.L. Equivalent Chain Length. b-T \leq 0.1% of total hydrocarbons.

Table 11. Composition of cuticular hydrocarbons

Hence, those hydrocarbons which were present at lower levels produced seriously depleted spectra. Several of the n alkanes, however, were quite abundant and their spectra were significantly enhanced by the use of C.I. (Fig. 11 a/b). The data system was particularly useful in allowing a search to be conducted through the MS scans whereby n - alkanes $(C_n H_{2n + \hat{c}})$ and mono-unsaturated n - alkenes $(C_n H_{2n})$ were detected on the basis of their molecular ions (Fig. 12). Table 11 shows the completed analysis of the outicular hydrocarbons n - alkanes (51.8%) comprised over half of the cuticular hydrocarbons and formed an homologous series (Series A, see 3.2.4.1.) with carbon numbers ranging from 21 to 33. Odd-numbered components (47.7%) predominated and, together, n - pentacosane (14.0%), n - heptacosane (15.1%) and n - nonacosane (15.6%) accounted for almost half of the cuticular hydrocarbons (44.7%). The mass spectrum (Fig. 13) of n - nonacceane (Peak no. 22) illustrates the fragmentation pattern typical of long chain n - alkanes. The largest peak in each cluster is at m/s ($C_n \mathbb{I}_{2n+1}$), indicating all fragmentions to be straight chain primary ions. The molecular ion (C, \mathbb{H}_{2m} , 2) was present (m/e 408) but weak while the most intense fragment ions occurred at C3, C4 and C5 with intensities decreasing along a smooth curve formation down to M - C₂H₅. Fragment ion, M - CH₂ was characteristically missing from the spectrum.





Eight n - alkanes (Series B, 26.4%) comprised over a quarter of the cuticular hydrocarbone with 15 - hentriacontene (15.4%) the most abundant of the series. Upon GLC separation of the hydrocarbons, each n - alkene was sluted immediately prior to the n - alkane of the same carbon number and the mass spectrum of each n - alkene contained a molecular ion at \mathbf{m}/\mathbf{e} (C_n H_{2n}). Fragmentations patterns throughout all n - alkene mass spectra were identical and the mass spectrum (Fig. 14) of 15 - hentriacontens (Peak no. 27) displays the fragmentation typical of this series. The base peak was at m/e 57 followed by two peaks of equal intensity at m/e 65 and m/e 97; then another, of lesser intensity, at m/e 69. After m/e 97, the fragment ion intensities decreased along a smooth curve to $M = C_{2}H_{4}$. As the n - alkene chains were long, it was impossible to locate the position of the double bond from the MS data due to the facile migration in the fragments of the double bond. However, the position had already been established for 15 - hentriacontene using GLC techniques (see section 3.2.4.1.) but not for other n - alkenes.

Series C compounds (12.1% of cuticular hydrocarbons) were not removed by treatment with molecular sieve 54 (see Materials and Methods section 2.2.4.2.3.) thus revealing the branched configuration of their structure. For each compound, a difference of 0.7 was observed between the equivalent chain length (E.C.L.) and the carbon number, indicating that the

compounds might be monomethykalkanes, each containing the methyl branch in a central position in the molecule (Nelson et al. 197). Further evidence was found from examination of the mass spectral data in which molecular ions (M) were missing but peaks invariably found at M-15 and M-27 (Fig. 15 Peak no. 17). Where fragmentation ion doublets occurred, the even mass peaks (C, H2n) equalled or exceeded the corresponding odd mass peaks (Cn H2n + 1) in intensity. This inferred that the alkane contained a single methyl branch attached at a position which allowed the formation, upon fragmentation, of secondary ions on which the straight chain tails comprised at least 7 carbon atoms (McCarthy et al. 1968). The mass spectrum of GLC Peak no. 17 (Fig. 15) contained ion doublets at m/e 168/169 (C12), m/e 196/197 (C14), m/e 224/225 (C_{16}) and m/e 252/253 (C_{18}) with predominant even mass peaks and low intensity peaks at m/e 365 (M = 29) and m/e 379 (M = 15). This fragmentation pattern indicated that the GLC peak consisted of two monomethylalkanes of the same carbon number (28), and both with a central methyl branch (i.e. at carbons 11 and 13 for the respective mathylalkanes). The compounds were identified as 11 - and 13 - methylheptacosane. Other members of the series were examined similarly and identifications made (see Table 11).

Evidence was found to suggest the existence of a fourth series (series D) of hydrocarbons consisting of dimethylalkanes.



GLG peaks 12, 30 and 36 had ECL values of 25.6, 31.6 and 33.6 respectively which, together with their mass spectra and nonremoval with molecular sieve 54, indicated that they were dior tri-methylalkanes. The mass spectrum of GLC peak 12 (Fig. 16) contained neither the molecular ion peak nor the more usual peaks at (M - 15) and (M - 29) due to the low concentration of the compound. Fragmentation ion doublets occurred m/e 168/169 (C₁₂), m/e 182/183 (C₁₃), m/e 210/211 (C₁₅) and m/e 238/239 (C₁₇) with the C₁₂ and C₁₇ ion doublets of almost equal intensity. The C₁₂ ion was considered to be a secondary fragment ion since its even mass peak(m/e 168) intensity exceeded that of its odd mass peak (m/e 169), while the C₁₇ ion was the corresponding primary fragment ion with its odd mass peak (m/e 239) intensity predominating due to the methyl branch at carbon 15 (Fig. 17).

The significance of ion doublets m/e 162/185 ($C_{1,5}$) and m/e 210/211 ($C_{1,5}$) in this depleted mass spectrum was difficult to establish. Their presence was possibly due to fragmentation within the central region of the molecule as shown in Fig. 17 but it would have been expected that m/e 211 (odd mass peak) exceed m/e 210 (even mass peak) in that case. Since, in fact, the opposite occurred, the ion attributed to m/e 210/211 might have been a secondary fragment ion produced in conjunction with ion doublet m/e 182/185 from the fragmentation of the monomethylalkape (12 methylpentacceane) shown in Fig. 18.







Figure 17. Structure for GLC peak 12.



Figure 18. Alternative structure for GLC peak 12.

The structure of this compound satisfied the mass spectral information; however, the GLC peak was eluted shortly after GLC peak 11 which comprised the unresolved monomethylalkanes 11 - and 13-methylpentacosane. If 12-methylpentacosane was present, it was unlikely to be sluted meparately. GLC peak 12 was therefore tentatively identified as 11, 15-dimethylpentacosane.

From the chromatogram of the hydrocarbons, the appearance of the second peak (GLC peak 30) of the Series (D) suggested that the peak contained more than one compound. The complexity of its mass spectrum (Fig. 19) certainly supported this view, and the spectrum was interpreted in the same manner as that of the previous dimethylalkane.

Tentative identifications (Fig. 20) were made which suggested that at least two dimethylalkanes (9, 11- and 11,21dimethylhentriacontane) were present; however, neither of these compounds could account for the branched primary fragment ion at π/e 210/211.

Fig. 21 shows the incomplete mass spectrum produced from another position in GLC peak no. 30. The ion doublet at m/e210/211 may be the result of a primary fragment ion of the same fragmentation which produced a secondary fragment ion at m/e280/281 (Fig. 22).



Figure 19. Mass spectrum of GLC peak 30 (First position on GLC peak).



9,19-dimethylhentriacontane



11,21-dimethylheutriacontane

Figure 20. Tentative structures for GLC peak 30.



9,13-dimethylhentriacontane

°0

×[0-0

280

C - C.

Figure 22. Tentative structure for GLC peak 30.

The third GLC peak in Series D (GLC peak 36) produced a grossly depleted mass spectrum due to low concentration and extended retention time and further identification was prevented.

3.2.4.3. GLC analysis of internal hydrocarbons/fascal hydrocarbons.

With the exception of the 15-hentriacontens levels, cuticular and fascal hydrocarbons were closely comparable; however, differences were noted when both were compared with internal hydrocarbons (Table 12). Differences were also observed within the internal hydrocarbon components from male and fasale adult lice. While n-pentacosane was the most abundant hydrocarbons in both adult sexes (Table 12), the level in the adult female (25.6%) far exceeded that in the adult male (15.6%). Conversely, the levels in the adult males of n- heptacosane (10.6%), n - octacosane (9.6%) and 15-hentriacontens (11.0%) were much higher than those in the adult females, (1.6% and 0.1% and 1.5% respectively).

Comparison of internal and cuticular hydrocarbons revealed that n - heneicosane and n - docosane, present at trace levels in the cuticular wax, were appreciably more abundant in the internal tissues of adult females (2.% and 2.0% respectively) and males (0.6% and 1.1% respectively). Major cuticular

Peak	Evdrocarbon	Percentage Composition of Eydrocarbons				
No.	Component(m)	Cuticular ^a	Cuticular ^b O' + Q	Internal ^b	Internal ^b Ş	Fancal ^b Of + Q
1	n - heneicosane	T	т	0.6	2.3	т
2	n - docosane	т	т	1.1	2.0	т
3	n - tricosene	т	Ť	т	т	т
4	n - tricosane	2.0	2.6	10.1	12.3	2.9
5	11-methyltricosans	0.3	0.6	1.4	2.0	0.6
6	Unknown	т	т	T	т	т
7	n - tetracosane	0.9	1.0	2.7	4.6	0.9
8	Unknown	т	т	0.2	0.2	т
9	n - pentacosene	0.2	0.2	0.5	0.4	0.2
10	n - pentacceane	14.0	14.5	15.8	25.8	17.5
11	11 + 13-methylpenta- cosane	1.2	1.7	3.1	5-4	1.8
12	11, 15-dimethylpenta- cosane	0.3	0.5	0.6	0.8	0.3
13	n - hexacosane	1.0	1.5	3.7	1.7	1.5
14	Unknown	т	т	0.3	0.5	0.5
15	n - heptacosene	0.8	1.0	2,0	2.5	0.7
16	n - heptacosane	15.1	15.6	10.4	1.6	19.3
17	11 + 13-methylhepta- cosane	1.7	1.7	3.0	7.5	2.0
18	n - cotacosene	0.7	0.6	0,6	0.7	0.3
19	n - octacosane	1.7	2.1	9.8	т	3.1
20	Unknown	0.5	0.3	0.3	1.0	0.2

Peak	Hydrocarbon	Percentage Composition of Hydrocarbons				
No.	Component(s)	Cuticular ^a	Cuticular ^b C + ¥	Internal ^b o	Internal ^b Ç	Faecal
21	n - nonacosens	7.8	8.0	15.1	10.2	7.3
22	n - nonacosans	15.6	15.8	T	т	20.7
23	7 + 9 + 11-methylnona- cosane	4.9	3.4	3.8	9.4	4.7
24	n - triacontene + unknown	2.4	1.4	1.1	2.0	1.8
25	n - triscontane	0.5	0.5	т	т	0.4
26	Unkno vn	0.4	0.3	т	т	0.3
27	15 - hentriacontene	15.4	15.3	11.0	1.5	7.0
28	n - hentriacontane	1.0	1.5	т	т	0.3
29	9 + 11 + 15-methyl- hentriacontane	3.3	3.5	0,6	0.3	2.7
30	9, 19 + 11, 21 + 9,13- dimethylhentriscontane	3.7	2.9	1.2	3.9	1.8
31	n - dotriacontane	т	т	т	т	т
32	Unknown	0.5	0.2	T	Т	T
33	n - tritriacontene	0.8	1.0	т	т	т
34	n - tritriacontane	т	т	т	т	т
35	13-methyltritria- contane	0.7	1.0	т	т	0.3
36	Possible dimethyl- alkanes	0.7	0.8	0.2	0.3	0.2

a - Estimations from areas given by electronic integration

b - Estimations using peak heights

c - T le 0.1 % of hydrocarbons.

Table 12. Comparison of outioular/internal/faecal hydrocarbons bydrocarbons, n - heptacosane (15.6%), n - nonacosane (15.6%)and 15 - hentriacontene (15.5%) were present at much lower levels in adult females (1.6%, 0.1% and 1.5% respectively) and this trend was repeated for n - nonacosane in the adult males (0.1%). In contrast with its low occurrence in cuticular wax, n - octacosane (2.1%) was found to be more abundant in adult males (9.5%).

3.2.4.4. GLC analysis of cuticular cholesteryl esters.

Table 15 lists the main fatty acids derived from the cholesteryl esters of the cuticular wax after identifications had been carried out by comparison of retention times of standards and unknowns and by co-chromatography. Although the degree of saturation of the fatty acids was not established, the fractional values for the equivalent chain lengths can probably be attributed to unsaturated carbon-carbon bonds within the chains.

The fatty acid chains ranged from 14 carbons (myristic acid, 1.0%) to 25 carbons (pentacosanoic acid 30.7%) with the latter being most predominant. Falmitic acid (16.0%) and oleic acid (9.6%) followed respectively in order of abundance; however, with cholesteryl esters accounting for less than 0% of the total wax, cholesteryl pentacosanoate was the only cholesteryl ester present at a significant level (2.3% of wax).

Table 13

Fatty acid composition of outicular cholesteryl esters

Fatty acid	Mean % Composition		
B.C.L.ª	of cholesteryl esters ^b	of vax	
14.0(C ₁₄₁₀)	1.8	0.1	
16.0(C ₁₆₊₀)	16.0	1.2	
17.6(C _{18:2})	1.5	0.1	
17.7(C _{18:1})	9.6	0.7	
18.0(C _{18:0})	4.3	0.3	
18.5	1.6	0.1	
19.6	2,1	0,2	
22.6	5-7	0.4	
23.8	5-4	0.2	
24.5	4-1	0.3	
24.6	7-5	0.6	
25.0(c ₂₅₁₀)	30.3	2.3	

a Equivalent chain length

b Fatty acids < 1.0% not listed

C where x is carbon number xiy

y is number of unsaturated bonds

3.2.4.5. GLC analysis of cuticular triglycerides.

Fractional equivalent chain length values suggested that some triglycerids fatty moids were unsaturated (Table 14). Palmitic acid (16 carbons) was the most abundant fatty acid (33.1%) followed by oleio acid (18 carbons, mono-unsaturated, 20.0%) and a fatty acid thought to be unsaturated and contain $\frac{2}{23}$ carbons (ECL 22.6, 12.6%).

3.2.4.6. GLC analysis of cuticular fatty acids.

Falmitic acid (16 carbons) and stearic acid (18 carbons) at 26.1% and 18.2% respectively were found to be the major fatty acids. Fractional equivalent chain lengths for many of the minor components indicated the likelihood of their being unsaturated. The fatty acids comprise the least abundant class of the cuticular lipids (3.87% by weight), hence the overall concentration of the predominant fatty acid (palmitic acid) is extremely low (approximately 1%). The main fatty acids are listed in Table 15. Many minor fatty acids were present (i.e. less than 2%); however, their levels of concentration in the total war might be regarded as insignificant (i.e. less than 0.00%).

3.2.4.7. GLC analysis of cuticular 1, 3 - diglycerides.

Falmitic acid (17.6%) was found to be the most abundant diglyceride fatty acid. Table 16 shows many of the fatty acids possessed fractional equivalent chain length values, due possibly to a high occurrence of unsaturation amongst the fatty acid chains.

Table 14

Fatty sold composition of outioular

triglycerides

Fatty acid	Hean % Comp	osition
R.C.L.	of total b triglyceride fatty soids	of wax
14.0(C ₁₄₂₀)	1,8	0.1
15.6	1.9	0.1
16.0(C16:0)	53.1	1.9
17.8(C1811)	20.8	1.2
18.0(C1810)	7.6	0.4
18.5	2.5	0.1
19.6	3.5	0.2
20.5	1.4	0.1
22.2	3.5	0.2
22.6	12.6	0.7

a Equivalent chain length

b Fatty solds < 1.0% not listed

C where x is carbon number

y is number of unsaturated bonds

Table 15

Composition of cuticular fatty acids

Fatty acid	Mean % Composition			
E.C.L. A	of total fatty acids	of wax		
14.0 (C _{14:0})	2.5	0.1		
16.0 (0 _{16:0})	26.1	1.0		
17.6 (C _{18:1})	5.0	0.2		
18.0 (C _{18:0})	18.2	0.7		
19.6	7.8	0.3		
20.0 (C _{20:0})	2.4	0.1		

a. equivalent chain length

b. fatty acids 🛭 🗲 2.0 % not listed

Table 10	5
----------	---

Fatty acid composition of cuticular 1, 3-diglycerides

Fatty acid	Mean % Composition			
E.C.L.ª	of total diglyceride fatty acids	of war		
13.8	2.4	0.1		
14.0(C _{14:0})	2.0	0.1		
14.8	6.4	0.3		
15.8	8.9	0.4		
16.0(C1620)	17.6	0.8		
16.8	8.3	0.4		
17.9	7.6	0.3		
18.0(C ₁₈₁₀)	8.4	0.4		
18.9	10.5	0.5		
19.8	10.0	0.5		
20.8	7.7	0.4		
21.8	4.9	0.2		

a Equivalent chain length

b Fatty acids < 2.0% not listed.

3.2.5. Examination of synthesis product (15-hentriacontene).

The following products were prepared as intermediate reagents for the synthesis of 15 - bentriacontene (see section 2,2,5, for synthesis obart).

I → II 1 - pentadecane (m.pt 20.2°C).

III - IV 1 - bexadecanal hexadecanal (m.pt. 34°C) II - V

pentadecyl magnesium bromide

IV + V ---- VII 16 - bentriacontanol (m.pt. 87°C)

Fig. 23 shows the chromatogram obtained when the synthesis endproduct (VIII) was analysed by GLC. The impurities were minor by comparison with the major peak (VIII) and appeared to consist of straight alkanee and mono-unsaturated alkenes.

The mass spectrum (Fig. 24) of the major peak was identical with that from peak number 27 (see section 2.2.4.7.). Both
spectra contained molecular ions at m/e 434 and fragment ion patterns. Ions were observed at m/e 55/57, 69/71, 83, 97. 111, 125, 139, 153, 167, 181, 195 etc. (at m/e 14 intervale) and the relative abundance of each ion in one spectrum corresponded with that of its counterpart in the other. Mass spectroscopic examination also revealed that the major GC peak attributed to the synthesis product comprised two unresolved peaks each possessing molecular ions at m/e 434. These peaks corresponded to the cis - and trans - isomers of the alkene, 15 - bentriacontere.



- 3.3. Applied aspects
- 3.3.1. Physical characteristics of cuticular lipids and adult lice

3.3.1.1. Helting-point determination of outicular war.

When subjected to a slow temperature increase, outioular louse wax began softening around $41-42^{\circ}C$ (Table 17) and gave the appearance that the melting was restricted to a minor component of the material. The condition remained constant until the temperature reached 49.5 - 50.5°C (Table 17) when the solid wax, which formed at least 90% of the total, melted sharply. By allowing the waxes to cool in the melting-point apparatus, the temperatures of solidification were observed and found to coincide with the corresponding melting-point temperatures.

Table 17

Melting-point temperatures of cuticular waxes

Wax Sample	Temperature of initial melting	Temperature of complete melting
1	42 °C	50.0 - 51.0
2	41 °C	49.0 - 50.5
3	42 °C	49.5 - 50.5

3.5.1.2. Belubility of simulated outicular wax in potential insecticide solvents.

The synthetic cuticular wax dissolved readily in potential insecticids solvents, didecyl ether, 1 - octadecene and n pentadecenes giving concentrations greater than 1.0 g/100 ml of solvent (Table 18). Solutions of the wax were less readily formed in dodeceneal, dodecenol, 2 - dodecenone and undecenol and the maximum possible concentrations of the wax fell in the range 0.5 - 1.0 g/100 ml of solvent. Lauryl ether dissolved the wax at concentrations not exceeding 0.25 g/100 ml of solvent while 1, 4 - butanedical and 1, 2 - propenedical failed to dissolve the wax to any appreciable extent.

3.3.1.3. Determination of upper lethal temperature of adult lice.

When subjected to an increase in temperature, both male and female lice died in the 50 - 52° C range (Table 19) after a layer of moisture had been noted on the cuticular surface of the insects at temperature immediately prior to the thermal death-point.

A record of changes in electrical conductivity on the surface of the adult female louse with temperature increase is shown in Fig. 25. The vibrational effect in the trace was caused by insect movement and increased dramatically above 46°C.

Solubility of synthetic cuticular wax in

potential insecticide solvents

Potential insecticide	Solubility of 1 mg wax in insecticids solvent				
solvent	a الم 100 solvent	b 200 pl solvent	o 300 pl solvent	d 400 pl solvent	
1, 4 - butanediol	0	0	0	0	
didsoyl/sther	+	+	+	+	
dodecanal	0	+	+	+	
dodecanol	0	+	+	+	
2 - dedacanone	0	+	+	+	
lauryl ether	0	o	0	+	
1 - octadecene	+	+	•	+	
n - pentadecane	+	+	+	+	
1, 2 - propanediol	0	0	0	0	
undecanol	0	+	+	+	

0 - insoluble

+ - soluble

a	-	concentration	1.0g wax/100 ml. solvent
Ъ	-	concentration	0.5g wax/100 ml. solvent
ø		concentration	0.33g wax/100 ml. solvent
đ	-	concentration	0.25g wax/100 ml. solvent

Thermal death-points of adult lice

Louse number	Males Thermal death-point ^O C	Famales Thermal death-point ^O C
1	52	51
2	52	51
3	51	50
4	50	52
5	51	51
6	50	52
7	50	52
8	52	51
9	50	52
10	51	51
Average	50.9	51.3



It was not possible to conduct the arperiments with adult male lice since moisture appeared to cover the insects within moments of their being placed in the melting-point tubes thus allowing the passage of electrical current between the wires of the electrode before any increase in temperature.

The traces produced by the female adults (Fig. 25) possessed relatively steady baselines as the temperature was slowly raised to $46 - 47^{\circ}$ C. A marked increase in the recorder response was observed after reaching this temperature range and was followed by a more dramatic increase on reaching $48 - 49^{\circ}$ C. Death of the insect occurred immediately after (i.e. $50 - 52^{\circ}$ C).

3.3.1.4. Comparative volatilities of potential insecticide solvents.

1, 2 - Propanediol was found to be the most volatile of the potential insectioide solvents (Table 20) followed by n - pentadecame which evaporated at a rate 1.4 times mlower than that of the diol. With the exceptions of didecyl ether and lauryl ether, which proved to be virtually non-volatile, the remaining solvents showed similar evaporation rates and were significantly less volatile than 1, 2 - propanediol and n - ventadecame (Table 20).

Comparison of times for complete evaporation of $(5 \ \mu$) potential insecticide solvents (at 30°C and 7% R.H.)

Potential insecticide solvent	Average evaporation rate (30°C / 75% RE) µ1/hour
1, 2 - propanediol	0.250 ± 0.012
n - pentadecane	0.178 ± 0.006
2 - dodecanone	0.052 ± 0.002
dodecanal	0.035 ± 0.001
1, 4 - butanediol	0.030 ± 0.001
undecanol	0.026 + 0.001
1 - octadacene	0,026 + 0.001
dodecanol	0.023 ± 0.001
didecyl ether	non volatile
lauryl ether	non volatile

3.3.2. Toxicity tests with adult lice.

3.3.2.1. Protocol for toxicity tests.

The mortality rates produced by three solvents (dodecanol, 2, - dodecenone and undecenol) among the adult female lice were markedly affected by the type of container in which the lice were separately held for the 24 hour test period (Table 21). For each solvent, the dosage applied per louse was more than sufficient to cause a mortality rate of 100% (see section 5.3.2.2.). Only toxicity tests carried out using glass containers consistently gave mortality rates of 100%, whereas the rates obtained with the plastic containers were significantly lower for all solvents. The most conspicious disparity between results was that observed for 2 - dodecanone (Table 21) where 64 µg of solvent/louse killed 100% of lice held in glass containers, 78.6% of lice held in plastic containers (A) and 3.3% of lice held in plastic containers (B).

Fig. 26 shows the regression lines produced by computer analysis of data from lauryl other toxicity tests in which glass containers and plastic containers (E) had been used. Results with the plastic containers compared to those with the glass (Table 22) indicated that it was necessary to apply 12 times more lauryl other per insect in order to achieve 50% mortality (LD_{ECC}) and 16 times more for 90% mortality (LD_{GC}).

Comparison of mortalities amonget adult female lice held in different container types after topical application

Test solvent	Container	Solvent Mean ^A dose mortality µg/louse %		Standard deviation %	
dodecanol	glass	120	100.0	+ 0.0	
	plastic A	120	66.6	+ 3.0	
	plastic B	120	44.5	+ 2.5	
2-dodecanone	glass	64	100.0	+ 0.0	
	plastic A	64	78.6	+ 6.6	
	plastic B	64	3.3	+ 1.5	
undecanol	glass	66	100.0	+ 0.0	
	plastic A	66	68.3	+ 4.9	
	plastic B	66	41.9	+ 7.7	

a - mean mortalities of 5 replicates



Test solvent	LD ₅₀ µg/louse	95% fiducial limits for LD ₅₀	LD ₉₀ pg/louse	95% fiducial limits for LD90	alops (*) standard error	Р
Lauryl ether (glass containers)	5.3	5.7 4.9	11.1	12.3 10.3	4.0301 ± 0.2789	> 0.05
Lanryl ether (plastic containers B)	63.7	71.3 55.7	179.8	231.5 151.1	2.8422 ± 0.3128	> 0.05

Comparison of results from toxicity tests with lauryl ether in which lice were held in glass or plastic (E) containers Fig. 27 to 36 illustrate the relationships between the solvent masses deposited during topical application and the corresponding number of micrometer divisions used. Linear graphs were obtained for each solvent.

3.3.2.2. Topical application of potential insecticide solvents.

Computer analysed doss/mortality data from the toxicity tests with the potential insecticide solvents are shown in Table 23 and the results were also used to construct the individual regression lines for the solvents on probit-log graph paper (Fig. 37 to 46).

Comparing toxicities of the solvents on the basis of their LD₅₀ and LD₅₀ levels, lauryl ether (5.3 µg/louse and 11.1 µg/louse respectively) was found to be the most potent compound (Table 23) followed by dodecanol (7.2 µg/louse and 13.7 µg/louse respectively). An examination of the collective regression lines of the 10 solvents (Fig.47) revealed that 3 broad groups were formed according to toxicity; with lauryl ether, dodecanol, 2 - dodecanoe and undecanol constituting the most toxic group. A second group, 3 to 4 times less toxic, was formed by 1 - octadecene, dodecanal, n - pentadecane and didecyl ether while a third group which was not appreciably toxic was formed by 1, 2-propanediol and 1, 4-butamediol.











¶ųst es)vent	Alling pac/lours	795 fiducial livite for Lingso	LD ₉₀ µg/10use	igg fiderial inits for 10 ₅₀	2) ape (n.)Standard	,
$1,\ 2-1 \text{ transition}$	-	- 14	-	1	0.6262 • 0.2596	0.05
didecyl ether	38+1	55.a 52.9	63.0	68.5 60.1	4.7660 • 0.2331	0.05
dedacan .	26	24.9 25.4	69.9	79.2 63.2	5.125 • 0.1962	0.04
Lalarme	14	1.6 6.9	13.7	14.2 12.0	4.6491 • 0.2010	0.05
To links prove	10.0	10.5 9.7	16.5	15.2 13.8	8.0522 • 0.4583	0.05
Sampl vomer	-64	5.7 4.9	11.1	12.2 10.5	4.0301 + 0.2779	0.05
1 - Settermen	78.2	25.7 27.8	55.6	61.4 51.2	\$,5561 + 0.206*	0.05
n - pentadecame	29.6	\$1.2 77.9	71,1	79.4	3. 3667 E ^{1. 1}	0.0%
S forgantid	-	-	-	-	1,0661 E ⁻¹⁰⁻¹⁰¹	0.05
an mal	11.0	11,3 10,6	17.5	18.5 16.7	6,215; • 0,3(10	0.05

-

Analysis of tenterity many being beauty and on all prior to-



PAGINATION ERROR







Figure 40. Regression line for dodecanol.



Figure 41. Regression line for 2-dodecanone.

Figure 42. Regression line for lauryl ether.









Figure 45. Regression line for 1,2-propanediol.

Figure 46. Regression line for undecanol.



Figure 47. Combined regression lines for the potential insecticide solvents.



3.3.2.3. Topical application of aqueous insecticide solvent mixture.

The 10% (ν/ν) mixture of lauryl sther in distilled water formed a light gel from which the water readily swaporated if left uncovered. Consequently, it was impossible to determine the mass of the topically applied droplets with the Gahn Electrobalance as described (see section 2.5.3.3.). However, when weighing droplets of the ten potential insecticide solvente (see section 2.3.3.3.), it was found that, on average 95.3% of the volume shown was actually deposited during topical application. Values for the LD₅₀ and LD₅₀ were adjusted accordingly after being calculated as described (see section 2.3.3.2.). The regression line for the 10% lauryl ether mixture (Fig. 48) was plotted as described (see section 2.3.3.2.). while a second graph (Fig. 49) was constructed which compared the lines produced by lauryl ether in the pure form and that actually present in the 10% mixture.

While there was close general agreement between the 2 regression lines (e.g. coincident LD_{90} values), a small difference in slopes was observed which resulted in 18.9% more lauryl other being necessary to achieve 50% mortality when the solvent was used in 10% agusous mixture rather than the pure form.





pure and aqueous forms :

Test solvent	LD 50 pg/louse	97% fidwcial limits for LD ₅₀	LD 90 pg/louse	99% fiducial limite for LD ₉₀
10% (v/v)lauryl ether in water	67.1	70.1 64.2	117.1	127.3 109.3
lauryl ether (present (in 10% mixture)	6.3	6.6 6.1	11.1	12.2 10.3
lauryl ether (applied as pure)	5.3	5.7 4.9	11.1	12.2 10.3

Comparison of lauryl ether toxicity resulting from pure and aqueous forms

3.3.2.4. Topical application of insecticide (dieldrin) in potential insecticide solvent.

After topical application of lice with doses of 0.005 pl/louse, lauryl ether produced an average rate of mortality of 42.6% while 1, 4 - butanedical applied at the same dosage produced no mortalities (Table 25). The average mortality rate increased to 70.6% and 100% when the same volume of lauryl ether per louse (0.005 pl) contained 0.02 µg of dieldrin and 0.2 µg of dieldrin respectively. A similar addition of dieldrin to 1, 4 - butanedical resulting in 0.02 µg and 0.2 µg of dieldrin in 0.005 µl of solvent per louse produced mortality rates of 12.5% and 100% respectively.

Therefore, the increase in mortality attributed to the dieldrin (0.02 $\mu_{\rm g}/$ louse) in the 1, 4 - butanedicl was 12.7% whilst that attributed to the dieldrin at the same level in the lauryl ether was 28%.

Comparison of dieldrin toxicity resulting from solutions in different insecticide solvents

Insecticide solvent	solvent volume per loume (µl)	Dieldrin per louse(µg)	% Mortality ^a
1,4-butanedicl	0.005	0.00	0.0
	0.005	0.02	12.3
	0.005	0,20	100.0
lauryl ether	0.005	0,00	42.6
	0.005	0.02	70.6
	0.005	0.20	100.0

a - average of 3 replicates

3.3.3. Observation of outicular penetration of potential insecticide solvents using fluorescent dys (Rhodamine B).

Cryostat sections from untreated lice and lice treated with solvants exclusive of Rhodsmine B (lauryl ether, dodecanol and 1, 4 - butanediol) displayed no significant fluorescence when examined under fluorescence microscopy with a Rhodsmine system (Plates 48 and 49). Consistent results were produced irrespective of treatment periods in the case of the solvent treated lice.

Lice treated with either lauryl ather or dodecanol were visibly affected after 30 minutes; but examination for Rhodamine fluorescence in sections from lice topically applied with the solutions of Rhodamine B, did not reveal any noticeable increases in fluorescence. After 60 minutes, when most of these lice had died, fluorescence was observed in sections of cuticle containing some degrees of solerotisation (Plates 50 and 52). The effects were more pronounced in sections taken from lice four hours after the initial treatment with either the lauryl ether or dodecanol Rhodamine E solutions (Plates 51, 55, 54 and 55).

In contrast with unsclerotised cuticle, sclerotised outicle was distinguished by a red-orange fluorescence (yellow-orange in micrographs) after the four hour period described above;

Flate 48. Untreated solarotised outicle from paratergal plate (near spiracle). No auto fluorescence is evident in the Rhodemine B region. x 1000.

Plate 49. Outicle at the junction of targum (membranous) and paratorgite (sclerotised) of female adult trasted with lauryl schar only (sectioned after 4 hours). He Bhedamine fluorescence is observed. x 1000.


Plate 50. Postnotum (membranous) separating subcoxae (selerite) of female adult louse (dead) 1 hour after treatment with laury] ether/ Rhodamine B. Faint fluorescence is evident in selerotised outlole. x 1000.

Plate 51. Cuticle from same location as in Plate 50 but sectioned 4 hours after treatment with lauryl ether/Rhodamine B. Fluorescence in selerotised outicle is stronger and pore canals are visible as striations. x 1000.



Plate 52. Cuticle from junction of rear leg and coxa of female adult louse (dead) 1 hour after treatment with lauryl sthere/Rhodamins B. Fluorescence is observed in sclerotised cuticle. x 1000.

246

Plate 53. Cuticle from dormal matathoracic subcoxa (above leg joint) of male adult louse sectioned 4 hours after doleanol/Rhodmine B treatment. Fluorescence is evident throughout outicle (sclerits) and trachess (arrows). x 1000.



Plate 54.

Cutiols from junction of leg and coxa of female adult louas sectioned 4 hours after treatment with lawryl sther/Bhodamine B. Fluorescence is observed mainly in solerotised outicle but also to a lesser degree in unsclerotised rugose outicle. x 1000.

Plate 55. Thorsecio outicle from ventral surface of famale adult louse sectioned 4 hours after dodecanol/ Rhodmaine B treatment. Fluoresence is seen throughout the cuticle (scierite) and pore canals are visible as stristione. x 1000.



examples of which were found in transverse sections showing the dorsal region of the thorax (Plates 51 and 53), the thorax-abdomen junction (Plate 54), the ventral region of the thorax (Plate 55) and the perimeters of the paratergal plates. Rugose cuticle and rubber-like cuticle were not stained by the Rhodamine B (lauryl ether, dodecanol) solutions to the same degree as solerotised outicle despite acting, in many instances, as an effective junction between two sclerotised regions (Plates 51, 53 and 54).

In outicle which was particularly well stained with the dye (Plate 55), pore canals were distinctly wisible as fluorescent, vertical striations in the cuticular cross section. Fluorescence was also observed in the traches of insects dosed with Ehodamine E/ dodecanol solution (Flate 53) which provided evidence that entry of the insect by the solution had been gained through the spiracles and the trachesl system.

Lice applied with 1, 4 - butanedicl (either pure or solution containing Rhodamine E) produced no mortalities during the tests. Under UV examination, sections of the lice dosed with the Rhodamine E/ 1, 4 - butanedicl solution taken 30 minutes and 60 minutes after treatment exhibited no significant increases in fluorescence. A fluorescent layer covering the cuticle was observed, however, in sections taken from lice four hours after the initial treatment (Flate 56).

Plate 56. Outical from junction of tergum and paratergite of female adult louse 4 hours after treatment with 1,4-butanediol/Ebodemine B. Fluorescence is observed on the surface only. X 1000.



4. Discussion

4.1. Physical structure.

4.1.1. External features.

The cuticles of some arthropods are capable of rapid expansion at some stage in their development and examples of such cuticles may be found in caterpillars, Rhodnius prolixus nymphs, queen termites, honey ants and female ticks (Neville, 1975). In each case the expansion occurs in unsclerotised areas of outicle which is usually folded to some extent. While in the unfed state, Rhodnius prolinus (fifth instar) and Boophilus microplus (adult female) have abdominal surfaces which are deeply folded or rugues (Hackman, 1975) and resemble the membranous abdominal cuticle of the human body louse. Although the Rhodnius symph and the Boophilus adult feed only once, whereas the louse feeds frequently, the close similarity of their rugose cuticles suggests that some aspects of abdominal expansion in the three animals are similar. The two forms of rugome outicle found on the body louse are consistent with those previously observed on the abdominal surface of the pubic louse by Ubelaker and coworkers (1973). They described the cuticle on the dorsal abdominal surface of the pubic louse as being rugose, but referred to cuticle on the ventral surface as 'scaled'; a term not wholly correct. The 'scaled' pubic louse cuticle appears exactly the same as the "quilted" cuticle seen in my studies on the body louse. SEM examinations of the abdominal body louse outicle revealed that the two distinct rugose cuticle formations were

closely related. Both consisted of corrugations but those of the quilted type were very much shorter in comparison to the long rugosities of the second type found over regions such as the midgut where expansion occurs during feeding. TEM examination of both types of rugose cuticle showed them to be structurally identical.

Quilted rugose cuticle was not confined to the abdomen but occurred in other regions such as leg joints, where movement caused by the underlying musculature requires a high degree of fluribility. In fact, an e-tensive system of muscles connecting the dorwal and ventral areas of quilted rugose cuticle on the abdomen was meen in the stained light microscope sections of whols lice (Flate 9). The series of abdominal pits found on the body louse in the same areas as the quilted rugose cuticle were previously described on the abdominal surface of the pubic louse (Ubelaker et al. 1973) and probably result from attachment of the underlying musculature to the integument. However, I could ascribe no obvious function to the areas of serrated rugose cuticle on the abdomen of the body louse, also noted on the abdomen of the public louse (Ubelaker et al. 1975).

Externally, the body louse is distinguished by several other unique features. As in all species of Anoplura, the mouthparts of the body louse are among the most highly specialised found in insects and function by piercing and sucking in a manner quite unlike that of other blood-sucking arthropods (Perris, 1951). The

claws on the legs are also characteristic in Anoplura and serve as clasping organs which are perfectly suited to an environment containing numerous hairs and fibres etc. (Burton, 1947). Although eyes are either absent or barely recognisable in most species of Anoplura, a small minority including the body louse possess distinct structures which consist of thickened cuticle (Ferris, 1951). The body louse may also be differentiated from other members of Anoplura by the mushroom-like shape of the spiracles which are characteristic of the family Pediculidae (Ferris, 1935).

4.1.2. The epicuticle.

A review of previous studies shows that some confusion has arisen concerning the terminology used for the various layers comprising the epicuticle. The structure of the epicuticle in sclerite and membranous body louse cuticle closely resembles that described by Filshie and Hadley (1979) in the desert accrpion <u>Hadrurus arizonensis</u> and consequently, I have used this terminology in my study.

Although a wax bloom was apparent in platinum/carbon replicas of outicle surfaces (Locke, 1974), it was almost completely removed during preparation for TEM examination and only remnants of it remained in well-protected areas. The unsaturated lipids which comprise over 25% of the outicular wax are readily fixed by cemium tetroxide and this accounts for the appearance in a few electron micrographs of isolated deposits adjacent to the outicle surface (Flates 13 and 16). This situation has been reported previously with other arthropods such as the spiny rat mite <u>Laclaps ochidnina</u> (Wharton <u>et al.</u> 1968) and the soil mite <u>Oppia colora ensis</u> (Brody, 1970). In fact, Wharton and coworkers demonstrated the presence of wax blooms on the surface opposite pore canals which suggests that cuticular wax traverses the cuticle via the pore canal system in this animal.

In most electron micrographs of louse cuticle the outermost layer of the epicuticle was the outer membrane, which corresponds with the lipid monolayer postulated by Beament (1961, 1964) and apparently demonstrated by Locke (1965, 1966) and Zacharuk (1972). Although the thickness of the layer (about 30 Å) is consistent with that of the outicular hydrocarbons when aligned at 90° to the surface, I tend to agree with Filshie (1970 a, b) and Lockey (1976) who question the existence of the so called oriented lipid monolayer. Studies with <u>Caluedes</u> (Filshie, 1970a), <u>Lucilia</u> (Filshie, 1970b) and H. arizonensis (Hadley and Filshie, 1979) have shown that in each instance, the outer membrane of the epicuticle was completely resistant to severe extraction under conditions more than sufficient to overcome any forces of attraction which might exist between the monolayer and the epicuticle. Lockey (1976) notes that the non-polar (hydrocarbon) contents of cuticular waxes from P. americana and S. gregaria are essentially equal and comprise the bulk of the wares (75-77%), which is also the case

with the body louse, and suggests that if the monolayer does exist, it must be a heterogeneous mixture of non-polar and polar lipid molecules. Both Filshie (1970s) and Lockey (1976), however, envisage that the lipid layer would be in a liquid phase to allow replacement of lost war and if this is so, the lipid monolayer could certainly not survive TEM preparation conditions.

Although the method by which the lipids (or precursors) are transported through the pore canal system is still unclear, there is no doubt that in my studies that the lipids (or precursors) pass through the epicuticle of the louse by way of the epicuticular filaments and are deposited on the surface at the pores or openings. Epicuticular pores are thought to exist in most insects and have been demonstrated in several species including Calpodes (Locks, 1966) and Thermobia (Moble-Mesbitt, 1967) whilst Hadley and Filshis (1979) observed pores in the epicuticle of <u>Hadrurus</u> arizonensis during TEM and SEM studies on the effects of various lipid solvents. The pores, with dismeters about 30 times larger than those found on the body louse, were apparent in the SEM only after prolonged treatment using concentrated potassium hydroxide and Hadley and Filshie (1979) suggest that the surface coating might contain mucopolysacchrides which may be digested by strong alkalis. In contrast, I found organic solvents were more effective in removing surface lipids from the louss, whereas treatment with concentrated potassium hydroxide

merely resulted in a redistribution of the hydrocarbon fraction which does not respond to saponification.

The horizontal tubules which exist at the interface of the procuticle and the dense layer of the epicuticle in the louse have been observed in only one other arthropod, the female cattle tick Boophilus microplus (Filshie, 1976) in which they form a single-layered, interconnecting network. Though it was not possible to produce tangential sections of louse cuticle for TEM examination, it is assumed that the tubules seen below the epicuticle in transverse sections represent a similar network to that in this species. Filshie (1976) was unable to associate any function with the tubular network in the tick, but think it is significant that both the tick and the louse are blood-suckers and in both cases the abdominal cuticle surface is folded to allow expansion when feeding (Hackman, 1975). Whilst inextensible, the epicuticle is still flexible enough to unfold during abdominal expansion (Richards, 1951). In studies with Rhodnius cymphs Maddrell(1966) demonstrated that the nervous system was involved in the plasticisation of abdominal cuticle and suggested that the stimuli resulting in cuticle plasticisation were those sensations which accompany feeding. He recognised that a number of sensory systems which respond to different stimuli have already been detected in certain insects. Perhaps the tubular network found in the cuticles of the tick and the louse performs a similar function to that of the stretch receptors which occur in the pharynx of Locusta

<u>migratoria migratoriodes</u> (Clark and Langley, 1965). Alternatively, the network might simply behave in a pneumatic manner between the sclerotised epicuticle and the unsclerotised exocuticle to lessen the effects of trauma during unfolding of the epicuticle during expansion.

4.1.3. The exocuticle.

Although Neville (1975) has stated that most exocuticles are heliocidal (lamellate) throughout their thickness, there are several instances where this is not the case. The exocuticle in the soft integuments of Rhodnius prolixus (Wigglesworth, 1970) and the rabbit tick Easmaphysalis leporispalustris (Nathanson, 1967) are non-lamellate and Beadle (1974) has suggested that in the non-lamellate exocuticle of the cattle tick Boophilus microplus. the microfibrile have unidirectional orientation. A similar arrangement exists in the exocuticle of louse membranous outicle except that the microfibril crystallites have formed a macrofibrillar structure. Neville (1975) has interpreted cuticular divisions on the basis of staining with Mallorv's triple stain (Baker, 1958) and concludes that aclerotized exocuticle is refractile to staining whereas unsclerotized exoouticle, which is impregnated with proteins and lipids, usually stains red. These observations are in agreement with those on aclerotised and membranous louse cuticle respectively after exposure to Mallory's triple stain. The brittle failure of the exocuticle during the fracture of membranous cuticle indicates

that the exocuticle is sclerotised; but the fixed condition of the outicls caused by the osnium tetroxide treatment and oritical point drying probably account for this anomolous behaviour. During expansion of the abdominal cuticle brought about by feeding, unfolding of the epicuticle is accompanied by a limited deformation of the exocuticle. Given the inextensible nature of the epicuticle and the alignment of the unidirectional exocuticle macrofibrils parallel to the axes of the rugosities, one would expect stretching of the exocuticle to be perpendicular to the axes of the rugosities with little or no extension along the rugosities. In the membranous cuticle the underlying endocuticle is capable of greater stretching and therefore, the exocuticle may be seen as a transitional layer between two extremes.

4.1.4. The endocuticle.

The different action of Mallory's triple stain on the endocuticle layers in sclerite and membranous louse outicle may be interpreted similarly to the exocuticle by applying the rationale of Neville (1975) and Eaker (1958). Where large enough spaces occur in the outicle structure, the larger anions (aniline blus) occupy them to produce a blue stain. In regions of the cuticle where impregnation by proteins and lipids has caused a reduction in the spaces between the molecular chains, only the

smaller anion (acid fuschin) can lodge which regults in a red stain. Where both impregnation and tanning have occurred, neither of the anions can penetrate the matrix and the cuticle does not stain, which is the case with the endocuticle of louse sclerite cuticle. In contrast, the endocuticle in Mallory stained membranous cuticle possessed an overall mauve colouration composed of red and blue. This variable reaction is consistent with the role that the endocuticle plays in the expansion of abdominal cuticle when the louse feeds. The extensibility of the abdominal cuticle regults largely from stretching of the endocuticle layer brought about by a change in cohesion between the chitin-protein components (Nunez, 1963). Because the membrancus abdominal cuticle is subjected to frequent distention through feeding by the louse, spaces of various dimensions are likely to be formed between the molecular chains in the endocuticle matrix on a random and recurring basis and this might account for the dual staining reaction.

The homogeneous, non-lamellate appearance of the sclerite endocuticle in TEM studies and its mechanical properties of strength and stiffness suggest that it is composed throughout of unidirectionally oriented microfibrils (Nevills, 1975) which is quite uncommon. Evidence of a lamellate structure in the endocuticle was revealed in SEM studies of fractured cuticle and this corresponds with the formation of growth layers deposited after blood meals. A similar phenomenon occurs in the tick <u>Ixodes ricinus</u> (Lees, 1952) and the bug <u>Rhodnius</u> (Zwicky and Wigglesworth, 1956).

In membranous louse outicle the structure of the endocuticle which is composed of helicoidally arranged (Bouligand, 1965) 250 - 1000 A macrofibrils closely resembles that observed in the chala tendon of the crab Carcinus maenas (Seville, 1975). Interestingly, the original studies by Bouligand (1965), whereby the helicoidal structure in cuticle was recognised, were performed on Crustacean cuticle which is also constructed of macrofibrils. One of the most interesting theories to explain the formation of unidirectional and helicoidal structure in cuticle involves the self-assembly of liquid crystalline material secreted by the spidermal cells (Neville and Caveney, 1969). It has been suggested (Neville, 1975; Wigglesworth, 1976) that a cholesteric component is responsible for controlling the angle of twist of the belicoid and, in fact, compounds such as cholesteryl esters are frequently found in cuticular lipid extracts. Cholesteryl esters occur in louse cuticular lipids and comprise over 7% of the total (by weight).

Although some points of eimilarity in the membranous abdominal outicles of the louse, <u>Rhodnius</u> larva and <u>Boophilus</u> adult were discussed earlier, the endocuticle structure in the outicles of the latter two arthropods is microfibrillar, unlike the macrofibrillar structure in the endocuticle of the louse. In the <u>Rhodnius</u> larva and the <u>Boophilus</u> adult, cuticle thickness is increased by synthesising additional endocuticle prior to feeding as that stretching of the cuticle can take place during abdominal expansion (Eackman, 1975). Since the louse adult feeds much more frequently than the <u>Rhodnius</u> larva and the <u>Boophilus</u> adult, its abdominal

expansion is not particularly dependent upon increased endocuticle deposition before each meal. Expansion of the abdomsen is limited by the inextensibility of the chitin macrofibrils.

Electron micrographs revealed that the macrofibril layers in the membranous endocuticle of the louse are folded to some extent which, no doubt, plays an important part in permitting maximum abdominal expansion. Whilst the structural form of the chitin within the abdominal outicle of the louse differs from that in Rhodnius and Boophilus, the mechanics of endocuticle plasticisation which permits stretching probably follow the same principles in each. All cuticles have a two-phase composite structure of high modulus polymaccharide fibres of chitin distributed in a low modulus matrix of protein. Bonds exist between the molecules of the two phases but in <u>Rhodnius</u> and presumably other blood-feeders. they are sensitive to pH change (Reynolds, 1975). Maddrell (1966) has shown that plasticisation of <u>Rhodnius</u> abdominal cuticle is under nervous control from the head and Neville (1975) has expressed the view that plasticisation is a highly complex event which is caused by a pH change within the cuticle controlled by the nervous system via the spidermis. Disruption of the pE - labile bonds between chitin and protein molecules allows a degree of sliding to occur between the two phases. Hackman (1975) has suggested that the components of flexible cuticles must not be so highly ordered or closely-packed that crystals are formed and thereby cause a gross change in physical properties. In the light of this statement, the results of Rhodamine E staining of louse cuticle (see section 3.3.4.) are interesting. Under identical experimental conditions,

sclerits outicle was consistently more heavily stained with Ehodamine E than membranous outicle, indicating that the sclerite cuticle contains a higher number of unconjugated protein bonds. Escently, Vincent and Hillsrton (1979) suggested that sclerotisation of cuticle results from highly controlled dehydration within the matrix allowing a large number of sites to become available for conjugation with chemicals involved in tanning, or in this instance, the protein dye Ehodamine B. Maddrell (1966) found that an increase in hydration of the cuticle accompanies plasticisation and this might account for the low uptake of Ehodamine B in the louse membranous cuticle which is, in any case, always hydrated to some extent. In fact, Nevills (1970) stresses that water is an essential component of cuticle, since dehydration increases the bonding between the chitin and protein molecules which results in brittleness.

4.1.5. The epidermis.

Integument examined during the TBM studies was taken from post-ecdysial adult lice, usually 2-3 days old, and in each case the decrease in activity following ecdysis was reflected in the appearance of the epidermal cells which were thin and elongated. In view of the rough endoplasmic reticulum and free ribosomes within the cytoplasm of the cells and their function of protein production (Bloom and Pawcett, 1975), it appears that endocuticle continues to be deposited until at least 2-3 days after ecdysis.

Although it has been shown in <u>Schistocerca grewaria</u> (Biehl, 1973) that the conceptes are responsible for the synthesis of cuticular lipids and that these cells have also been implicated in the production of structural lipids in Rhodnius (Wigglesworth, 1970), the epidermal cells in the louse may, perhape, feature more prominently in lipid production.

Results from the experiments in which adult lice ingested blood labelled with tritiated cleic acid imply that the turnover of cuticular lipids in the louse is rapid and probably dependent to some degree on feeding. In the epidermis the basal extremities of the pore canal filaments appear to maintain a close extracellular relationship with the epidermal cells suggesting that these cells are responsible for synthesising the major proportion of the outicular lipids (or precursors). A deduction supported by the observation that cencoytes were not immediately evident in the electron micrographs.

4.1.6. The pore canal system.

The TGM and SGM examinations of lowes cuticle have established that the surface of the cuticle is linked to the epidemis via the pore canal/epicuticular filament system; also proposed as the pathway for lipids or precursors involved in waterproofing of the epicuticle (Looke, 1961; Gluud, 1968). It has been known for some time that the lowe possesses epicuticular pores (Vigglesworth, 1942) and it has since become evident that the existence of the pore canal system is crucial in the development of the outicle as a whole. As well as being involved in waterproofing of the epicuticle, the pore canals facilitate impregnation of the cuticular layers (Neville, 1975) particularly during sclerotisation (Vigglesworth, 1975b). During ecdysis the moulting fluid gains access to the inner layers of the old cuticle via the newly formed epicuticular channels (or filaments) and the products of digestion then return by the same route (Vigglesworth, 1975b).

Though the platinum/carbon replicas of freeze-fractured membranous cuticle provided some evidence that the pore canals resemble twisted ribbons (Neville and Luke, 1969), the TEM examinations revealed that the structures are similar to those observed in Boonhilus microplus adults (Filshie, 1976) and in Rhodnius prolixus larvas (Wigglesworth, 1975a); where they were described as being roughly cylindrical in cross section. Locks (1961) demonstrated that an esterase is present in the cuticles of Calpodes and Galleria in regions where wax canal filaments (epicuticular filaments) arise and, consequently, he suggested that the final stage of wax synthesis occurs in these regions and involves this esterase. If cuticular lipid synthesis is completed in the epidermis, one might expect to see uniform osmium tetroxide staining throughout the pore canal system, particularly in the case of the louse which has a fairly high unsaturated lipid content. In fact the heaviest staining in

the pore canal system occurred throughout the filamentous network found mainly in the epicuticle and exocuticle. Staining of the pore canals in the endocuticle appeared significantly lighter and was restricted to the walls of the structures (Wigglesworth, 1975a). This would seem to agree with the theory of Locks (1961) that lipid synthesis is completed in the pore canals where the wax canal filaments arise. On the other hand, Filshie (1970a, b) found that the filaments in the epicuticle and exoouticle layers of <u>Calpodes sthlum</u> and <u>Lucilia cuprime</u> are resistant to acid hydrolysis and the action of lipid solvents, which suggests that the filaments commiss of or contain substantial amounts of stabilised compounds (lipids) rather than being composed entirely of free lipids.

An alternative explanation for the variable osmium staining in the pore canal system might also apply. All specimens of cutiols were prepared from unfed lice and if cuticular lipids are produced in response to feeding, we might well expect to observe some visible evidence within the pore canals of the cessation of lipid transport.

The role of the pore canal filaments in the active transport of lipids or precursors to the surface of the outicle could not be determined from these studies, but in view of their welldefined physical structure noted in the SEM, it does seem likely that they serve to keep the pore canals aligned during expansion and contraction of the membranous cuticle (Wigglesworth, 1975s).

4.2. Cuticular lipids of the body louse.

The extremely low level of radioactive contamination in the cuticular lipid extract from lice fed on labelled blood (see section 2.2.2.3.) demonstrates the ability of the polar solvent, hexane, to extract only a minimum of internal lipids while removing most of the surface lipids. Observations on feeding lice have shown that a small volume of clear liquid is usually disoharged from the anus shortly after commencement of bloodfeeding and this, rather than the extraction of internal lipids might account for most of the contamination observed.

The total extractable cuticular wax from either sex of the louse, expressed as a percentage of total fresh body weight (0.016 - 0.019%), ranks among the lowest reported for arthropods. Curiously, similar levels have been observed in scorpions such as <u>Paruroctonus messensis</u> (0.03 - 0.09%) (Hadley and Jackson, 1977) and <u>Hadrurus arizonensis</u> (0.03%) (Toolson and Hadley, 1977) which are noted for their extremely low water loss rates and the desert temsbrionid beetles <u>Centricoters muricats</u> (0.03%), <u>G. varioloss</u> (0.016%) and <u>Palecynhorus</u> adversus (0.02%) (Hadley, 1978). In the light of studies with the scorpions <u>H. arizonensis</u> and <u>Uroctonus suchasmus</u>. Toolson and Hadley (1977) concluded that the highly impermeable nature of the scorpion cuticle resulted from surface lipid composition rather than the presence of an uncommonly thick layer of lipids. Studies with other species of arthropode indicate that the quantity of wax deposited on the

cuticle surface varies with environmental conditions. Deposits of outicular hydrocarbons on pupal and post-ecdysial adult fleshflies Sarcophaga bullata greatly exceed those on larvae up to the third instar which live in a moist environment and prevention of pupariation through contact with water is accompanied by a decrease in the quantity of hydrocarbon synthesized (Armold and Regnier, 1975). Cutioular hydrocarbons on adult desert tenebrionid bestles <u>Eleodes</u> armata collected in summer months are almost three times as abundant as those from adults collected during winter and winter bestles which were subjected to a higher temperature (35°C) for periods of five and ten weeks showed significant increases in the levels of outicular hydrocarbons (Hadley, 1977). Armold at al. (1969) observed that the squatic naiad form of the big stonefly Pteronarovs californica exhibited a quantity of surface was which was half that of the terrestrial adult form. This is of particular interest in the present study, since the amount of extractable surface wax on the louse is similar to that found on the naiad big stonefly (0.0%). The suggestion that as the naied big stonefly inhabits an aquatic environment it requires only minimal amounts of surface lipid to prevent dessication (Armold et al., 1969), is also appropriate in the case of the louse. Transcuticular water losses for the lice used in my studies have been minimized by maintaining a high humidity (75% relative humidity at 30°C) during rearing, and ensuring that the insects were fully fed once each day (Beament, 1976). The low amount of surface wax on the louse is, therefore, in keeping with its low requirement

for water conservation.

Hydrocarbons often constitute the most abundant lipid class in the surface way of arthropods (Jackson and Elonquist, 1976). The cuticular hydrocarbons of the louse are typical in this respect and the percentage (about 74%) of the total surface lipids which the hydrocarbons comprise finds closest agreement with the values from the locust Schistocerca gregaria (76%) (Lockey, 1976) and the fire ants Scienopsis invicts (65-75%) and S. richteri (65-75%) (Lok et al. 1975). Those classes which make up the cuticular hydrocarbons of the louse; n - alkanse, n - alkenes, internally branched monomethylalkanes and dimethylalkanes have not been found in this particular combination in other insect species (Lockey, 1980a). n - Alkane content of the louse cuticular hydrocarbons (51.6%) is smong the higher levels reported for insects whilst the predominance of the odd-numbered n - alkanes; C25, C27 and C29, is common in insect hydrocarbon mixtures. The n - alkane with the longest chain-length contains 35 carbon atoms which is in line with the theory that insects are unable to synthesise n - alkanes larger than 33-35 carbons (Jackson and Blomquist, 1976).

In comparison with the n = alkens fraction in outicular lipids of other insact species, the n = alkenss of the lowse comprise a reasonably high percentage (26.4%) of the total cuticular hydrocarbons, but are not as abundant as those of the male cockroaches

Periplaneta australasias, P. fuliginosa (Jackson, 1970) and P. japonica (Jackson, 1972). The composition of the louse n - alkene series is typical of that observed in other insects possessing cuticular n - alkenes in that odd-numbered alkenes are predominant and one or two constituents of the series are much more abundant than others. 15 - Hentriacontene, the major n - alkene of the louse, has also been identified in an isomeric mixture in the cuticular hydrocarbons of the tenebrionid beetle Tenebrio molitor (Lockey, 1978a) whilst T. obscurus contains 9 - and 11 - hentriscontenes (Lockey, 1978b). A hentriscontene (probably 9 - hentriscontene) was detected as a minor cuticular hydrocarbon component in another tenebrionid bestle Alphitophagus bifasciatus (Lockey, 1979). n - Nonacosene (double bond position unknown) is the second most abundant n - alkene of the louse, and moncunsaturated alkanes containing 29 carbon atoms have been identified in the cuticular hydrocarbons of several other insect species including T. molitor. T. obscurus (Lockey, 1978a), A. bifagciatus (Locksy, 1979) and P. japonica (Jackson, 1972).

The absence of terminally branched monomethylalkanes in the outicular lipids of the louse is most unusual eince the greater majority of arthropods investigated so far possess either 2 methylalkanes or 3 - methylalkanes or both (Lockey, 1960a). It may be possible, however, that terminally branched monomethylalkanes are present in the louse in extremely low concentrations and are not readily resolved during gas chromatography.

Internally branched monomethylalkanes (12.1%) do not comprise an unusually high proportion of the louse hydrocarbons in comparison with those of other insect species, and the composition of the series (C_{24} to C_{54}) is strictly conventional with the 11 - and 13 - methylalkanes predominant (Helson, 1978). Although dimethylalkanes are present in the cuticular hydrocarbons of the louse, their percentage of the total hydrocarbons is low (about 4.7%) compared to that in other insects where they occur.

Hadley (1978, 1981) likens the surface wax layer on arthropod outicle to a plasma membrane or an artificial bilayer and studies involving these two systems indicate that permeability to water is increased by the presence of unsaturated or branched molecules (de Grier et al, 1968; Taylor et al, 1975) and is decreased by the addition of larger molecules (de Grier et al, 1968). It has been noted that among arthropods, a high content of unsaturated or branched chain hydrocarbons in a cuticular was tends to lower the melting temperature of the mixture and increases its mobility (Gilby and Cox, 1963; Neville, 1975; Baker at al, 1978). The predominantly unsaturated hydrocarbon mixture from the cockroach P. americana melts at a temperature 10°C lower than the respective cuticular hydrocarbon mixtures from the cockroaches Leucophasa maderae and <u>Blatts</u> orientalis which contain only saturated constituents (Tartivita and Jackson, 1970). That the differences in melting point temperatures might be associated with the climatic conditions which the cockroaches inhabit, such as temperature and

humidity, is supported by observations on the naiad and adult forms of the big stonefly Pteronarcys californics (Armold et al. 1969) and on larvae and adults of the black carpet beetle Attagenus megatoms (Baker et al. 1979a). In both species, the differences in surface lipid composition between the two developmental stages which are characterised by the presence of unsaturated hydrocarbons in the naiad P. californica and the adult A. megatoma reflect the varying needs for water conservation (Armold et al. 1969; Baker et al. 1979a). In addition, the fact that the cuticular hydrocarbons from many desert species of arthropods such as the scorpions Hadrurus arizonensis (Toolson and Hadley, 1977) and Paruroctomus messensis (Hadley and Jackson, 1977), the locust Schistocerca gregaria (Lockey, 1976) and the five tenebrionid bestles Elecdes armats. Cryptoglosss verrucoss. Centrioptera muricata. C. variolosa and Pelecyphorus adversus (Hadley, 1978) are completely saturated supports the concept that unsaturated cuticular hydrocarbons occur in these arthropods inhabiting environments in which transouticular water loss is not a problem. As was established earlier, the lice used in these studies were reared under conditions where the potential for desiccation was negligible and there appears to be a correlation between this and the high overall content in the cuticular hydrocarbons of unsaturated and branched chain constituents.

The remaining lipids in the outicular way of the louse, cholesteryl esters, triglycerides, fatty acids, 1, 3 - diglycerides and sterols (cholesterol) are present in roughly equivalent proportions and no single class would be expected to have a significant effect on the overall properties of the way, particularly in view of the abundance of the hydrocarbons. The general predominance of palmitic acid and cleic acid in the lipid classes except sterols is common among insect outicular waxes (Jackson and Elemquist, 1976). A cholesteryl ester which constituted the second most abundant non-hydrocarbon component after cholesterol in the louse also contained the longest fatty acid from the lipid classes. The medium chain length (25 carbon atoms) of the cholesteryl ester fatty acid is in keeping with a soft cuticular way in the louse rather than a hard one (Baker et al., 1979s).

Quantitative analyses of the whole body and fascal lipids of the louse were carried out by thin-layer chromatography/densitometry because only small samples were available while the cuticular lipids were analysed using the same method for comparative purposes. Although there was poor numerical agreement between the cuticular lipid composition obtained by this method and that obtained by weighing the lipid classes, the hydrocarbone constituted by far the major fraction in both analyses. The abundance of fatty acids in the whole body lipids corresponds with their importance in

insect mutrition (Gilbert, 1967) and the comparatively high levels of cholesterol and cholesteryl esters in the whole body lipids and the fascal lipids are consistent with the essential requirements of insects for cholesterol and its steady supply in the dist (45 mg/100 ml rmbbit's blood or 0.3 µg in 0.75 µl; Spector, 1956).

Although sarlier studies on the grasshopper <u>Melanopius</u> <u>manguininas</u> have demonstrated that n - alkanes may be obtained directly from the dist (Elemquist and Jackson, 1973), this is unlikely to occur in the louse since long chain hydrocarbons are not present in its dist (rabbit's blood) in appreciable quantities. The extensive incorporation of labelled cleic acid into the hydrocarbon fraction of the cuticular lipids suggests that cleic acid might be a natural precursor of these hydrocarbons and since the louse lipids were collected 24 hours after ingestion of the radiolabelled material, the high level of incorporation implies that the surface wax turnover is quite rapid. In the lipids of the whole body and the faces, the low level of incorporation of radiolabelled material into cholesterol compared with that into the cholesteryl esters is consistent with most of the cholesterol being derived directly from the diet.

The situation in which the hydrocarbons in the faceal and outicular lipids of the louse are qualitatively and quantitatively similar has also been observed in the adult housefly <u>Musca domestica</u>

by Silhacek et al. (1972). Although they proposed that the faecal matter becomes contaminated either in the gut, or, externally through the condensation of hydrocarbons which have evaporated from the cuticle surface, in the louse it probably occurs simply by contact between the fasces and the surface lipid layer.

The quantitative differences between the hydrocarbons in the outicular wax and the (male and female) whole body lipids may have resulted from the incomplete removal of cuticular lipids from the bodies of the lice. However, the fact that n nonaccesses, a major cuticular hydrocarbon, was detected in only trace amounts in the whole body hydrocarbons suggests that efficient extraction of cuticular lipids did occur prior to the analysis of the whole body hydrocarbons. The occurrence of large quantitative differences between the male and female hydrocarbons from whole body extracts of lice appears to be in conflict with the original findings which showed the cuticular hydrocarbons from male and female lice to be virtually identical (see section 3.2.4.1.). Male and female lice used in the analyses of cutioular hydrocarbons were selected from colonies in which there was repeated contact between all insects. A transfer of cuticular waxes during these encounters might explain why the male and female lics possessed similar cuticular hydrocarbon distributions. The higher levels of n - heptacosane, n - octacosane and 15 -

hentriacontene in the male whole body hydrocarbons is ourious, but may be connected with a similar phenomenon observed in the hasmolymph of the male cockroach <u>P</u>. <u>americana</u> whereby the fluctuation of hydrocarbons varies with photoperiod (Turner and Acres, 1967). 4.5. Applied aspects.

4.5.1. Cuticular permeability/transition temperature.

Although some doubts have been raised concerning the existence in terrestrial arthropods of a transition temperature' or a specific temperature at which an abrupt transcuticular water loss takes place (Toolson, 1978), the data from my studies suggest that this phenomenon does occur in the louse. The upper lethal temperature of the louse (50-52°C) and the melting temperatures of its cuticular war (41 - 42°C and 49.5 - 50.5°C) are consistent with observations on the change in electrical potential across the cuticular surface of female lice with increasing temperature. Recorder traces of this change indicate that electrically conductive material, presumably water, is deposited slowly on the cuticular surface above 46-47°C and very rapidly above 48-49°C. This also agrees with my observation that lice of both sexes become covered with a film of moisture immediately prior to the thermal death-point. Wiggslesworth (1942) found that when lice were held under clive oil for long periods, droplets of water were studed from the surface of the outicle and appeared primarily in areas of sclerotisation. This corresponds with the results of my ultrastructural studies which showed that sclerotised louse cuticle contains significantly more epicuticular filaments reaching the surface than does membranous louse outicle.
The upper lathal temperature for lice of 46.5° C reported by Mellamby (1932) was obtained from trials in which the various temperatures were maintained for 1 hour. Under these circumstances, death probably resulted from slow desiccation as opposed to the rapid water loss at the higher lathal temperature range (50 - 52°C) obtained in my experiments. That the outicular surface of the male adult louse is electrically conductive at ambient temperatures (see section 5.3.1.3.) is unusual but might be explained by the fact that the rate of outicular transpiration in these is probably high as a result of being reared under conditions where desiccation is not a problem (see section 4.2.). This explanation is, however, contradicted by the fact that the outicular surface of the female adult louse is electrically nonconductive at ambient temperature.

On the basis of his observations on permeability change with temperature Beament (1945) proposed that a highly oriented monolayer of polar lipids at the interface of the epicuticle and surface lipids controls the passage of water through the cuticle. He conoluded that the sudden increase in permeability associated with the transition temperature results from the disorganisation of the oriented monolayer. In an alternative theory, Looke (1965) suggested that the epicuticular filaments observed in insect cuticle are lipid-water liquid orystals and phase changes within these structures may be responsible for the variable permeability of outicle to water. Filshie (1970 a, b), however, found that the

epicuticular filements of <u>Calpodes ethlius</u> and <u>Lucilia</u> <u>suprime</u> are formed substantially of stabilised lipids since they are unaffected by sold hydrolysis and lipid solvents.

4.3.2. Toxicity test procedure.

The most commonly employed technique for evaluating the toxicities of various compounds to lice involves the exposure of insects to squares of oloth or filter paper impregnated with the toxicant in the form of a dust or an acetone solution (Busvins, 1971). In tests of this type, the insects are usually kept in darkness at a constant temperature (25°C) for a period of 24 hours before determining the rate of mortality. It is obvious that there are many variable factors associated with this procedure which is regarded only as a semi-practical test (Busvine, 1971). In an effort to eliminate as many variables as possible, it was decided to use the topical application technique since it involves giving known doses to individual insects. Attempts to restrain lice for testing by placing them upright on double-sided Sellotape were unsuccessful due to the effects of some of the test compounds upon the Sellctape. With the migration of the organic liquids over the surfaces of the lice, glue material in contact with the lice was soon reached and dissolved which allowed the insects to escape. Placing individual lice in separate containers (see section 2.3.3.2.) was more successful

but as the results indicate (see section 3.3.2.1.), the choice of container is critical. Of the three types of containers tested, the glass container gave the most satisfactory results on the basis of reproducability.

The internal surfaces of the three types of containers were examined under UV radiation after having held lice dosed with Rhodamine B dissolved in 2-dodecanone for 24 hours. In the case of the plastic containers, much of the test solution applied to the lice became deposited on the internal surfaces of the containers whereas the surface of the glass containers was essentially unvettable.

Taking an objective view of the toxicity test procedure used in my studies, the principal advantage lies in the reproducability of results but must be weighted against the fact that the procedure, in its entirety, is particularly time-consuming.

4.3.3. Toxicity of potential insecticide solvents.

In many arthropods, treatment with various toxicants causes a marked increase in the water loss rate by upsetting processes involved with water conservation (Ebeling, 1976). Both Wigglesworth (1942) and Beament (1945) have provided evidence that oils and organic solvents cause significant water loss from insects by

disruption of the epicuticular war layer and this is of particular relevance in the present studies. Of the ten potential insecticide solvents investigated, 1,4-butanedicl and 1, 2-propanedicl were least toxic to lice and were also the only solvents in which the simulated cuticular war mixture did not dissolve. These results are consistent with those from the cuticular penetration studies using solutions of Rhodamine B (see section 3.3.4.). Four hours after topical application of a 0.5% solution of Rhodamine B in 1, 4-butanedicl, weak fluorescence was observed on the surface of the louse cuticle only (Plate 56), whereas a solution of Rhodamine B in the more lipophilic solvent, dodecanol, produced fluorescence throughout the procuticle (Plate 55), indicating extensive penetration by the solution.

The remaining eight molvents appear to form two separate groups based on their levels of toxicity to female adult lice (Figure 47). In general, aclvents of the less toxic group were more efficient in dissolving the simulated outicular war mixture than those of the more toxic group, and the toxicity of the former probably results solely from the disruption of epicuticular war. The higher toxicity of the more toxic group of solvents arises from other properties such as the possession of more favourable phase partition coefficients (Webb and Green, 1945; Burt and Lord, 1968; Burt et al, 1971) in addition to war solubility. Significantly, the most toxic of the ten molvents, lauryl sther, is the only one readily miscible with water and this property allows it to diffuse

from the lipid phase of the outer cuticle to the aqueous phase of the interior of the insect. The penetrative and toxic nature of lauryl ether appears not to be significantly impaired when it is applied as an aqueous solution. Being the less volatile of the two iquids, lauryl ether remains on the cuticular surface as the water evaporates.

In view of the preceding results, indicating the ability to dissolve the epicuticular wax layer and partition from lipid phase into aqueous phase, lauryl ether appears to have the most potential as an insecticide carrier. Limited tests with dieldrin in lauryl ether suggest that this is correct but a more thorough investigation is necessary before firm conclusions may be drawn.

4.3.4. Tracer studies with solutions of Rhodamine B.

Whilst the results of the penetration studies using Rhodsmine B in either lauryl ether or dedecanel are consistent with the theory that insecticides and solvents pass through the cuticle via the pore canals and into the hasmolymph (Flate 55) (Wigglesworth, 1942; Lewis, 1965), there is also some evidence of entry via the tracheal system (Flate 55) (Gerolt, 1969, 1970). As far as the cuticular route is concerned, heavier Rhodsmine B staining in sclerite outicle than in membraneous cuticle does not necessarily indicate that the main path of entry occurs in the former type of cuticle. Nevertheless,

earlier experiments by Wigglesworth (1942) suggest that the solerite cuticle of the louse is more permeable than the membranous cuticle and this is consistent with the results of my ultrastructural studies which indicate that surface pores are more numerous on sclerite louse cuticle than membranous. Furthermore, in some micrographs of Ehodamine E - stained solerite cuticle, a strinted effect is evident between the epidermis and the epicuticle (Plates 51 and 55) which may be attributed to the presence of pore canal structures. Lawis (1963) however, has demonstrated that penetration is possible through cuticle in which pore canals and dermal gland ducts are absent and some degree of lateral diffusion through the louse cuticle might therefore be expected. The role of Ehodamine B as a protein stain (Mairn, 1976) and its differential staining of solerite and membranous cuticles is discussed more fully in section 4.1.4.

Gerolt (1969, 1970) suggests that contact insecticides enter an insect by a lateral migration through the cuticle (endocuticle) into the tracheal system. However, the Rhodamine B fluorescence in the traches in Plate 51 has probably resulted from surface migration of the solution (Lewis, 1962) rather than lateral diffusion in the cuticle as proposed by Gerolt.

The shortcomings of this method of tracing penetration in insect cutiols are fully recognized; but while there is almost certainly a difference in the rates of penetration between Rhodamine B and the

test solvents, it is improbable that the stain could diffuse through the outicle unless sided by a solvent. Autoradiography, which is a more informative method, could not be used in this case however, since the time available was limited.

Acknowledgements.

I would like to express my thanks to Professor W. W. MacDonald, Director of the Department of Entomology for the period of this study, and Nr. J. W. Maunder, who acted as my supervisor, for the opportunity to undertake this project and their continued support during its progress.

My thanks are also due to Mr. W. Gunne from Queen Elizabeth College for his generous co-operation in obtaining the mass spectra; to Mr. J. Rivers for the use of his GLC equipment, and to Dr. R. Bird and his collesgues in the Electron Microscopy Unit for their helpful assistance.

The interest shown by Dr. N. Lehane in this study is greatly appreciated and I am particularly grateful to Dr. C. Leake for his invaluable advice in the latter stages.

Financial support for the work was generously provided by Reed and Carnrick of America and Stafford- Miller of Britain which are both submidiaries of Bloch Incorporated. My personal contacts with these companies have been very rewarding and I would especially like to thank Mr. V. Kally, Mr. J. Green and Dr. A. Singer for their involvement over the past years (not to mention the free lunches!).

Finally, my thanks are due to Miss M. Talati for her extreme patience in typing this thesis.

References.

- AHRENS, W. (1930). Uber die Korpergliederung, die Haut und die Tracheenorgane der Termitenkonigin. Jena Z. Naturw. 64, 449-530.
- ARMOLD, N. T., BLOMQUIST, G. J., JACKSON, L. L. (1969). Cuticular lipids of insects - III. The surface lipids of the squatic and terrestrial life forms of the big stonefly, <u>Pteromarcys</u> californics Newport. Comp. Biochew. Physiol. <u>31</u>, 685-692.
- ARMOLD, M. T., RECNIER, F. E. (1975). A developmental study of the cuticular hydrocarbons of <u>Sarcophaga bullata</u>. J. Insect Physiol.21, 1827-1833.
- BAKER, J. R. (1958). Principles of biological microtechnique. Methuen, London.
- BAKER, G., PEPPER, J. H., JOHNSON, L. H., HASTINGS, E. (1960). Estimation of the composition of the cuticular wax of the Mormon cricket, <u>Anabrus</u> <u>simplex</u> Hald. J. Insect Physiol. 5, 47-60.
- BAKER, G. L., VROMAN, H. E., PADMORE, J. (1963). Hydrocarbons of the American cockroach. Biochem. biophys. Res. Commun. 13, 360-365.
- BAKER, J. E., SUKKESTAD, D. R., WOG, S. H., NELSON, D. R. (1978). Cuticular hydrocarbons of <u>Tribolium castaneum</u>: Effect of food additive tricalcium phosphate. Insect Biochem. <u>9</u>, 159-167.
- BAKER, J. E., NELSON, D. R., FATLAND, C. (1979a). Developmental changes in cuticular lipids of the black carpet beetle, Attagenus megatoma. Insect Biochem. 9, 335-339.

- BAKER, J. E., SUKKESTAD, D. R., NELSON, D. R., FATLAND, C. L. (1979b). Cuticular lipids of larvae and adults of the cigarette beetle, Lasioderma serricorne. Insact Biochem. 9, 603-611.
- BEADLE, D. J. (1974). Fine structure of the integument of ticks, <u>Boophilum decoloratus</u> Kach and <u>B. microplum</u> (Canastriní) (Acarina : Ixodidae). Int. J. Insect. Morph. Embryol. 3, 1-12.
- BEAMENT, J. W. L. (1945). The cuticular lipids of insects. J. exp. Bicl. 21, 115-131.
- BEAMENT, J. W. L. (1961). The water relations of insect cuticle. Biol. Rev. 36, 281-320.
- BEAMENT, J. W. L. (1964). The active transport and passive movement of water in insects. Adv. Insect Physiol. 2, 67-129.
- BEAMENT, J. W. L. (1976). The ecology of cuticle. In: The Insect Integument. (ed. H. R. Hepburn) pp. 359-374. Elsevier, Amsterdam.
- BEATTY, I. M., GILBY, A. R. (1969). The major hydrocarbon of a cockroach cuticular wax. Naturwissenschaften 56, 373.
- BENEZET, H. J., FORGASH, A. J. (1972). Penetration and distribution of topically applied malathion in the house fly. J. econ. Entomol. <u>65</u>, 53-57.
- BLAILOCK, T. T., BLOHQUIST, G. J., JACKSON, L. L. (1976). Biosynthesis of 2-methylalkanes in the crickst, <u>Nemobius</u> <u>Essciatus</u> and <u>Gryllus pennaylvanicus</u>. Biochem. biophys. Res. Commun. 68, 841-849.

- BLOMQUIST, G. J., JACKSON, L. L. (1973). Incorporation of labelled dietary n-alkanes into cuticular lipids of the grasshopper, Nelanoplus sanguinipes. J. Insect Physiol. 19, 1639-1647.
- BLOMQUIST, G. J., BLAILOCK, T. T., SCHEETZ, R. W., JACKSON, L. L. (1976). Cuticular lipida of insects - VII. Cuticular hydrocarbons of the crickets, <u>Acheta domesticus</u>, <u>Gryllus</u> <u>permsylvanicus</u> and <u>Nemobius fascistus</u>. Comp. Biochem. Physicl. 548, 381-386.
- BLOHQUIST, G. J., KEARNEY, G. P. (1976). Biosynthesis of internally branched monomethylaikanss in the cockroach, <u>Periplaneta</u> fuliginosa, Arch. Biochem. Biophys. 173, 546-553.
- BLOMQUIST, G. J., JACKSON, L. L. (1979). Chemistry and biochemistry of insects waxes. Prog. Lipid Res. 17, 319-345.
- BLOMQUIST, G. J., HOWARD, R. W., MCDANIEL, C. A. (1979a). Structures of the cuticular hydrocarbons of the termite <u>Zootermopsis</u> angusticollis (Hagen). Insect Biochem. 9, 365-370.
- BLONQUIST, G. J., HOWARD, R. W., McDANIEL, C. A. (1979b). Bicaynthesis of the cuticular hydrocarbons of the termite <u>Zootermopsis angusticollis</u> (Hagen). Incorporation of propionate into dimethylalkanes. Insect Biochem. 9, 371-374.
- BLOOM, M. D., FAWCETT, D. W. (1975). A textbook of Histology. (10th edition) pp. 1033. W. B. Seunders Company., Philadelphia, London, Toronto.
- BOULIGAND, Y. (1965). Sur une architecture tormade répandue dans de nombreuses cuticles d'arthropodes. C. R. Acad. Sci., Paris 261, 3665-3668.

- BOULIGAND, Y. (1972). Twisted fibrous arrangements in biological materials and cholesteric mesophases. Tissue and Cell 4, 189-217.
- BOWERS, W. S., THOMPSON, M. J. (1965). Identification of the major constituents of the crystalline powder covering the larval cuticle of <u>Samis cynthis ricini</u>. J. Insect Physiol. 11, 1003-1011.
- BRATTSTEN, L. B., WILKINSON, C. F. (1977). Insecticide solvents : Interference with insecticidal action. Science, Wash. 196, 1211-1212.
- BRODY, A. R. (1970). Observations on the fine structure of the developing cuticle of a soil mite <u>Oppin</u> <u>coloradenmis</u> (Acarina : Cryptostigmata). Acarologia <u>12</u>, 421-431.
- BRÜCK, E., KOMNICK, H. (1971). Histochemister Nachweis der Penetrationswege von Salzlösungen durch die isolierte Insektencuticula. J. Insect Physiol. <u>17</u>, 2027-2034.
- BUERGER, A. A., O'BRIEN, R. D. (1965). Penetration of nonelectrolytes through animal integuments. J. Cell. Comp. Physiol. <u>66</u>, 227-234.
- BURT, P. E., LORD, K. A. (1968). The influence of penetration, distribution, sorption and decomposition on the poisoning of the cockrosch <u>Periplaneta smericana</u> treated topically with diazozon. Entomol. exp. Appl. <u>11</u>, 55-67.
- BURT, P. E., LORD, K. A., FORREST, J. M., COODCHILD, R. E. (1971). The spread of topically-applied Pyrethrin I from the cuticle to the central nervous system of the cockroach Periplaneta americana. Entomol. exp. Appl. 14, 255-269.
- BUSVINE, J. R. (1971). A critical review of the techniques for testing insecticides. pp. 345. The Commonwealth Institute of Entomology, London.

- BUXTON, P. A. (1947). The louse. An account of the lice which infest man, their medical importance and control. (2nd edition). Edward Arnold and Co., London.
- CLARK, G. L., SMITH, A. F. (1936). X-ray diffraction of chitin, chitosan and derivatives. J. phys. Chem., Ithaca 40, 863-879.
- CLARKE, K. U., LANGLEY, P. A. (1963). Studies on the initiation of growth and moulting in <u>Lacusta migratoria migratorioides</u> R. & F. - III. The role of the front ganglion. J. Insect Physiol. 9, 411-421.
- CONRAD, C. W., JACKSON, L. L. (1971). Hydrocarbon biosynthesis in Periplaneta americana. J. Insect Physical. <u>17</u>, 1907-1916.

CORSON, B. B., IPATIEFF, V. N. (1955). Org. Syn., Coll. II, 152.

- DE GRIER, J., MANDERSLOOT, J. C., VAN DEENEN, L. L. M. (1968). Lipid composition and permeability of liposcmes. Biochim. Biophys. Acta 150, 666-675.
- DENNELL, R. (1960). Integument and exoskeleton. In: The physiology of Crustacea. (ed. T. H. Waterman) 1, pp. 449-472. Academic Press, London.
- DIEHL, P. A. (1973). Paraffin synthesis in the cenocytes of the desert locust. Nature, Lond. 243, 468-470.
- DOWNING, D. T., CREENE, R. S. (1968). Rapid determination of doublebond positions in monoenoic fatty acids by periodatepermanganate oxidation. Lipids 3, 96-100.
- DRACH, P. (1953). Structure des lamelles cuticulaires chez les Crustaces. C. R. Acad. Sci., Paris 237, 1772-1774.
- DRACH, P. (1939). Mue et cycle d'intermue chez les Crustaces Decapodes. Annls. Inst. oceanogr. Monsco (N. S.) 19, 103-391.

- EBELING, W. (1974). The permeability of insect cuticle. In: The Physiology of Insecta. (ed. M. Rockstein) 6, 271-343. Academic Press, London.
- EBELING, W. (1976). Insect integument: a vulnerable organ system. In: The Insect Integument. (ed. H. R. Hepburn) pp. 383-400. Elsevier, Amsterdam.
- EDNEY, E. B. (1974). Desert arthropods. In: Desert biology. (ed. G. W. Brown) 2, 311-384. Academic Frees, New York.
- FERRIS, G. P. (1935). Contributions toward a monograph of the sucking lice. Stanford Univ. Publ., Univ. Ser., Biol. Sci. II (8), 531-634.
- FERRIS, G. F. (1951). The sucking lice. Mem. Pacific Coast Ent. Soc. 1. New York Lithographing Corporation, New York.
- FILSHIE, B. K. (1970a). The remistance of spicuticular components of an insect to extraction with lipid solvents. Tissue and Cell 2, 181-190.
- FILSHIE, B. K. (1970b). The fine structure and deposition of the larval cuticle of the sheep blowfly, <u>Lucilia cuprina</u>. Tissue and Cell 2, 479-498.
- FILSHIE, B. K. (1976). The structure and deposition of the epicuticle of the adult female cattle tick (<u>Boophilus wicroplus</u>). In: The Insect Integument, (ed. H. R. Repburn) pp. 193-206. Elsevier, Amsterdam.
- FILSHIE, B. K., HADLEY, N. F. (1979). Fine structure of the cuticle of the desert scorpion, <u>Hadrurum arizonensis</u>. Tissue and Cell 11, 249-262.
- FRAENKEL, G., RUDALL, K. M. (1940). A study of the physical and chemical properties of the insect cuticle. Proc. R. Soc. (8) 129, 1-35.

- FRAENKEL, G., RUDALL, K. M. (1947). The structure of insect cuticles. Proc. R. Soc. (B) <u>134</u>, 111-143.
- GEROLT, P. (1969). Mode of entry of contact insecticides. J. Insect Physicl. <u>15</u>, 563-580.
- GEROLT, P. (1970). The mode of entry of contact insecticides. Pestic. Sci. <u>1</u>, 209-212.
- GILBERT, L. I. (1967). Lipid metabolism and function in insects. Adv. Insect Physiol. 4, 69-211.
- GILBY, A. R. (1980). Chemical methods (Lipids). In: Cuticle techniques in arthropods. (ed. T. A. Miller) pp. 217-252. Springer, New York.
- GTLBY, A. R., COX, M. E. (1963). The cuticular lipids of the cockrosch, <u>Periplanets americans</u> (L). J. Insect Physiol. 9, 671-681.
- GLUUD, A. (1968). Zur Feinstruktur der Insektencuticula. Ein Beitrag zur Frage des Eigengiftschutzes der Wanzencuticula. Zool. Jb. Anst. Bd. <u>85</u>, 191-227.
- HACKMAN, R. H. (1971). The integument of Arthropode. In: Chemical zoology. (ed. M. Florkin and B. T. Scheer) pp. 1-62. Academic Press, New York.
- HACKMAN, R. H. (1974). Chemistry of the insect cuticle. In: The Physiology of Insecta. (ed. M. Rockstein) 6, 216-270. Academic Press, New York.
- HACKMAN, R. H. (1975). Expanding abdominal cuticle in the bug <u>Rhodnius</u> and the tick <u>Boophilus</u>. J. Insect Physiol. <u>21</u>, 1613-1623.

JACKSON, L. L., ARNULD, M. T. (1977). Insect Lipid Analysis. In: Analytical biochemistry of insects. (ed. R. B. Turner) pp. 171-206. Elsevier, Amsterdam.

- HADLEY, N. F. (1977). Epicuticular lipids of the desert temebrionid beetle, <u>Eleodes armata</u>: seasonal and acclimatory effects on composition. Insect Biochem. 7, 277-283.
- HADLEY, N. F. (1978). Cuticular permeability of desert tenebrionid beetles: correlation with epicuticular hydrocarbon composition. Insect Biochem. 8, 17-22.
- HADLEY, N. F. (1981). Cuticular lipids of terrestrial plants and arthropods: A comparison of their structure, composition and waterproofing function. Biol. Rev. <u>56</u>, 23-47.
- HADLEY, N. F., FILSHIE, B. K. (1979). Fine structure of the epicuticle of the desert scorpion, <u>Hadrurus arizonensis</u>, with reference to location of lipids. Tissue and Cell <u>11</u>, 263-275.
- HADLEY, N. P., JACKSON, L. L. (1977). Chemical composition of the epicuticular lipids of the scorpion, <u>Paruroctonus</u> messensis. Insect Biochem. 7, 85-89.
- HEPBURN, H. R. (1976). (ed.) The Insect Integument. pp. 571. Elsevier, Amsterdam.
- HOFFMAN, R. A., ROTH, A. R., LINDQUIST, A. W., BUTTS, J. S. (1952). Absorption of DDT in houseflies over an extended period. Science, Wash. 115, 312-313.
- HUTCHINS, R. F. N., MARTIN, N. M. (1968). The lipids of the common house cricket, <u>Achets domesticus</u> L. Lipids <u>3</u>, 247-259.
- JACKSON, L. L. (1970). Cuticular lipida of insecta II. Hydrocarbona of the cockroaches, <u>Periplaneta australasiae</u>, <u>Periplaneta</u> brunnea and Periplaneta fuliginosa. Lipida 5, 18-41.
- JACKSON, L. L. (1972). Cuticular lipids of insects IV. Hydrocarbons of the cockroaches, <u>Periplaneta japonica</u> and <u>Periplaneta</u> <u>americana</u> compared to other cockroach hydrocarbons. Comp. Biochem. Physicl. <u>418</u>, 331-336.

- JACKSON, L. L., BAKER, G. L. (1970). Cuticular lipids of insects. Lipids 5, 239-246.
- JACKSON, L. L., BLOMQUIST, G. J. (1976). Insect waxes. In: Chemistry and biochemistry of natural waxes. (ed. P. E. Kolattukudy) pp. 201-233. Elsevier, Amsterdam.
- JACOB, J. (1977). The cuticular lipids of the beetle, <u>Phyllobius</u> <u>maculicornis</u>. Hoppe-Seyler's Z. Physiol. Chem. <u>358</u>, 1375-1377.
- JENSEN, M., WEIS-FOGH, T. (1962). Biology and physica of locust flight. V. Strength and elasticity of locust cuticls. Phil. Trans. R. Soc. (B) 245, 137-169.
- KOIDSUMI, K. (1957). Antifungal action of cuticular lipids in insects. J. Insect Physiol. 1, 40-51.
- KÜHNELT, w. (1928a). Ein Beitrag zur Histochemie des Insektenskelettes. Zool. Anz. 75, 111-113.
- KÜHNELI, W. (1928b). Studien über den mikrochemischen Machweis des Chitins. Biol. Zbl. 48, 374-382.
- KÜRNELT, W. (1928c). Über den Bau des Insektenskelettes. Zool. Jahrb. Anst. 50, 219-278.
- KURIHARA, N., NAKAJIMA, E., SHINDO, M. (1970). Whole body autoradiographic studies on the distribution of BHC and nicotine in the American cockroach. In: Biochemical Toxicology of Insecticides. (ed. R. D. O'Brien and I. Yamaoto) pp. 41-50. Academic Press, New York.
- LEES, A. D. (1952). The role of cuticle growth in the feeding process of ticks. Proc. Zool. Soc. Lond. <u>121</u>, 759-772.

- LE ROUX, E. J., MORRISON, F. O. (1954). The adsorption, distribution, and site of action of DDT-resistant and DDT-susceptible house flies using carbon¹⁴ labelled DDT. J. econ. Entomol. 47, 1058-1066.
- LEWIS, C. T. (1962). Diffusion of oil films over insects. Nature, Lond. 193, 904.
- LEWIS, C. T. (1963). Some applications of radioisotopes to the study of contamination of insects by insecticide solutions. Proc. Ser. Int. Atomic Energy Agency <u>74</u>, 135-146.
- LEWIS, C. T. (1965). Influence of cuticle structure and hypodermal cells on DDT absorption by <u>Phormia terraenovae</u> R-D. J. Insect Physiol. <u>11</u>, 683-694.
- LEWIS, C. T. (1980). The penetration of cuticle by insecticides. In: Cuticle techniques in arthropods. (ed. T. A. Miller) pp. 367-399. Springer, New York.
- LOCKE, M. (1957). The structure of insect trachese. Quart. J. wicr. Sci. 98, 487-492.
- LOCKE, M. (1960). Cuticle and wax secretion in <u>Calpodes</u> athlius (Lepidopters, Hesperiidae). Quart. J. micr. Sci. <u>101</u>, 333-338.
- LOCKE, M. (1961). Pore canals and related structures in insect cuticle. J. biophys. biochem. Cytol. <u>10</u>, 589-618.
- LOCKE, M. (1965). Permeability of insect cuticle to water and lipids. Science, Wash. <u>147</u>, 295-298.
- LOCKE, M. (1966). The structure and formation of the cuticulin layer in the epicuticle of an insect, <u>Calpodes ethlius</u> (Lepidopters, Kesperiidae). J. Norph. <u>118</u>, 461-494.

- LOCKE, M. (1974). The structure and formation of the integument of insects. In: The Physiology of Insects. (ed. M. Rockstein) 6, 123-213. Academic Press, New York.
- LOCKE, M., SMITH, D. S. (1980). (ed.) Insect biology in the future "VBW". pp. 977. Academic Press, New York.
- LOCKEY, K. H. (1960). The thickness of some insect epicuticular wax layers. J. exp. Biol. 37, 316-329.
- LOCKEY, K. H. (1976). Cuticular hydrocarbons of Locusts, Schistocercs and Periplanets and their role in waterproofing. Insect Biochem. 6, 457-472.
- LOCKEY, K. H. (1978s). The sdult cuticular hydrocarbons of <u>Tenebrio</u> molitor L. and <u>Tenebrio</u> obscurum 7. (Coleopters: Tenebrionidge). Insect Biochem. <u>B</u>, 237-250.
- LOCKEY, K. H. (1978b). Hydrocarbons of adult <u>Tribolium castaneum</u> Hbst. and <u>Tribolium confusum</u> Duv. (Coleopters: Tenebrionidae). Comp. Biochem. Physiol. <u>618</u>, 401-407.
- LOCKEY, K. H. (1979). Cuticular hydrocarbons of adult <u>Alphitophagua</u> <u>bifasciatus</u> (Say) and <u>Alphitobius</u> <u>disperinus</u> (Panz). (Coleopters: Tenebrionidae). Comp. Biochem. Physiol. <u>64B</u>, 47-56.
- LOCKEY, K. H. (1980s). Insect cuticular hydrocarbons. Comp. Biochem. Physiol. 658, 457-462.
- LOCKEY, K. H. (1980b). Cuticular hydrocarbons of adult <u>Blaps</u> <u>mucronata</u> Latreille. (Coleoptera: Tenebrionidae). Comp. Biochem. Physiol. <u>678</u>, 33-40.
- LOK, J. B., CUPP, E. E., BLOMQUIST, G. J. (1975). Cuticular lipids of the imported fire ants, <u>Solenopsis</u> invicts and <u>richteri</u>. Insect Biochem. 5, 821-829.

- LOTMAR, W., PICKEN, L. E. R. (1950). A new crystallographic modification of chitin and its distribution. Experientia 6, 58-59.
- MADDRELL, S. H. P. (1966). Nervous control of the mechanics1 properties of the abdominal wall at feeding in <u>Rhodnius</u>. J. exp. Biol. <u>44</u>, 59-68.
- MAJOR, M. A., BLOMQUIST, G. J. (1978). Biosynthesis of hydrocarbons. Decarboxylation of long chain acids to n-alkanes in <u>Periplanets</u>. Lipids <u>13</u>, 323-328.
- MANGOLD, H.K. (1965). Aliphatic lipids. In: Thin-layer chromatography. (English edition) (ed. E. Stahl) pp. 137-186. Springer-Verlag, Berlin.
- MARTIN, M. M., Mac CONNELL, J. G. (1970). The alkanes of the ant, Atta colombica. Tetrahedron <u>26</u>, 307-319.
- MATSUMURA, F. (1963). The permeability of the cuticle of Periplaneta americana (L) to Malathion. J. Insect Physiol. <u>9</u>, 207-221.
- Mc CARTHY, E. D., HAN, J., CALVIN, M. (1968). Hydrogen atom transfer in mass spectrometric fragmentation patterns of saturated aliphatic hydrocarbons. Anal. Chem. 40, 1475-1480.
- MELLANBY, K. (1932), The influence of atmospheric humidity on the thermal death point of a number of insects. J. exp. Hiol. 9, 222-231.
- MEYER, H. (1842). Über den Bau Hornschale der Käfer. Arch. Anst. Physiol. wiss. Med. pp. 12-16.
- MICHELSON, A. A. (1911). On metallic coloring in birds and insects. Phil. Mag. 21, 554-567.
- NAIRN, R. C. (1976). Fluorescent protein tracing. pp. 648. Churchill Livingstone, Edinburgh, London.

- NATHANSON, M. (1967). Comparative fine structure of sclerotized and unsclerotized integument of the rabbit tick, <u>Haemaphysalis</u> <u>leporispalustris</u> (Acari: Ixodides: Ixodidae). Ann. ent. Soc. Am. 60, 1125-1135.
- NELSON, D. R. (1978). Long-chain methyl-branched hydrocarbons: Occurrence, biosynthesis, and function. Adv. Insect Physiol. 12, 1-33.
- NELSON, D. R., SUKKESTAD, D. R. (1975). Normal and branched alkanes from cast skins of the grasshopper, <u>Schiatocerca vaga</u> (Scudder). J. Lipid Res. <u>16</u>, 12-18.
- NELSON, D. R., SUKKESTAD, D. R., TERRANOVA, A. C. (1971). Hydrocarbon composition of the integument, fat body, hemolymph and diet of the tobacco hornwurm. Life Sci. 10, 411-419.
- NELSON, D. R., SUKKESTAD, D. R., ZAYLSKIE, R. G. (1972). Mass spectra of methyl-branched hydrocarbons from eggs of the tobacco hornworm. J. Lipid Res. 13, 413-421.
- NEVILLE, A. C. (1970). Cuticle ultrastructure in relation to the whole insect. Symp. R. ent. Soc. Lond. <u>5</u>, 17-39.
- NEVILLE, A. C. (1975). Biology of the arthropod cuticle. pp. 448. Springer-Verlag, Berlin, Heidelberg, New York.
- NEVILLE, A. C., CAVENEY, S. (1969). Scarabaeid beetle exocuticle as an optical analogue of cholesteric liquid crystals. Biol. Rev. 44, 531-562.
- NEVILLE, A. C., LUKE, B. M. (1969a). Molecular architecture of adult locust cuticle at the electron microscope level. Tissue and Cell 1, 355-366.
- NEVILLE, A. C., LUKE, B. M. (1969b). A two-system model for chitinprotein complexes in insect cuticles. Tissue and Cell <u>1</u>, 689-707.

- NEVILLE, A. C., PARRY, D. A. D., WOODHEAD-GALLOWAY, J. (1976). The chitin crystallite in arthropod cuticle. J. Cell Sci. <u>21</u>, 73-82.
- NEVILLE, A. C., THOMAS, M. G., ZELAZNY, B. (1969). Pore canal shape related to molecular architecture of arthropod cuticle. Tiesue and Cell <u>1</u>, 183-200.
- NOBLE-NESBITT, J. (1967). Aspects of the structure, formation and function of some insect cuticles. In: Insects and physiology. (ed. J. W. L. Beament and J. E. Treherne) pp. 3-16. Oliver and Royd, Edinburgh.

NOLLER, C. R., DINSMORE, R. (1943). Org. Syn., Coll. 11, 358-362.

- NUMEZ, J. A. (1963). Central nervous control of the mechanical properties of the cuticle in <u>Rhodnius prolinus</u>. Nature, Lond. 199, 621-622.
- O'CONNOR, J. C., BURROW, F. H., NORRIS, M. S. (1962). Determination of normal paraffins in C₂₀ to C₃₂ paraffin waxes by molecular sieve admorption. Anal. Chem. <u>34</u>, 82-85.
- ODIER, A. (1823). Memoire sur la composition chimique des parties cornees des insects. Mém. Soc. Hist. nat. Paris 1, 29-42.
- PARTCH, R. E. (1964). The conversion of alcohols to aldehydes and ketones: A superior method. Tet. Letters <u>41</u>, 3071-3077.
- RAMSAY, J. A. (1935). The evaporation of water from the cockroach. J. exp. Biol. 12, 373-303.
- REYNOLDS, S. E. (1975). The mechanism of plasticisation of the abdominal cuticle in <u>Rhodnius</u>. J. exp. Biol. <u>62</u>, 81-98.

RICHARDS, A. G. (1951). The integument of Arthropods. pp. 411. University of Minnesota Press, Minnespolis.

- RICHARDS, A. G., KORDA, F. H. (1948). Studies on arthropod cuticle.
 2. Electron microscope studies of extracted cuticles.
 Biol. Bull. mar. biol. Lab. 94, 212-235.
- RICHARDS, A. G., KORDA, F. H. (1950). Studies on arthropod cuticle. 4. An electron microscope survey of the intime of arthropod trachese. Ann. ent. Soc. Am. 43, 49-71.
- ROY, D. N., CHOSH, S. M., CHOPRA, R. N. (1943). The mode of action of pyrethrum on the cockroach <u>Periplaneta americana</u> L. Ann. appl. Biol. <u>30</u>, 42-47.
- RUDALL, K. M. (1965). Skeletal structure in insects. In: Aspects of insect biochemistry. Biochem. Soc. Symp. 25, 83-92.
- RUDALL, K. M. (1967). Conformation in chitin-protein complexes. In: Conformation of biopolymers. (ed. C. N. Ramachandran) 2, 751-765. Academic Press, London.
- RUDALL, K. M., KENCHINGTON, W. (1973). The chitin system. Biol. Rev. 49, 597-636.
- SILHACEK, D. L., MAYER, M. S., CARLSON, D. A., JAMES, J. D. (1972). Chemical classification of a male housefly attractant. J. Insect Physiol. 18, 43-51.
- SKIPSKI, V. P., SMOLOWE, A. F., SULLIVAN, R. C., BARCLAY, M. (1965). Separation of lipid classes by thin-layer chromatography. Biochim. biophys. Acta 106, 386-396.
- SPECTOR, W. S. (1956). Handbook of biological data. W. B. Saunders Co., Philadelphia, London.

TORRE-BUNEO, J. R. (1962). A Glossary of Entomology. pp. 336. Brooklyn Ent. Soc.

- TAYLOR, A. R., ROUBAL, W. T., VARANASI. U. (1975). Effects of structural variation in β-monoglycerides and other lipids on ordering in synthetic membranes. Lipids 10, 535-541.
- TOOLSON, E. C., (1978). Diffusion of water through the arthropod cuticle: thermodynamic consideration of the transition temperature phenomenon. J. therm. Biol. <u>J.</u> 69-73.
- TOOLSON, E. C., HADLEY, N. F. (1977). Cuticular permeability and epicuticular lipid composition in two Arizons vejovid acorpions. Physiol. 2001. <u>50</u>, 323-330.
- TURNER, R. B., ACREE, F., Jr. (1967). The effect of photoperiod on the daily fluctuation of haemolymph hydrocarbons in the American cockroach. J. Insect Physiol. <u>13</u>, 519-522.
- UBELAKER, J. E., PAYNE, E., ALLISON, V. F., MDORE, D. V. (1973). Scanning electron microscopy of the human pubic louse, Pthirus pubis (Linnaeus, 1758). J. Parasitol. <u>59</u>, 913-919.
- VINCENT, J. F. V., HILLERTON, J. E. (1979). The tanning of insect cuticle - a critical review and a revised mechanism. J. Insect Physiol. 25, 653-658.
- VROMAN, H. E., KAPLANIS, J. N., ROBBINS, W. E. (1964). Cholesterol turnover in the American cockrosch, <u>Periplaneta americana</u> L. J. Lipid Res. 5, 418-421.
- WEBB, J. E., GREEN, R. A. (1945). On the penetration of insecticides through the insect cuticle. J. exp. Biol. 22, 8-20.

WEIS-FOGH, T. (1970). Structure and formation of insect cuticle. Symp. R. ent. Soc. Lond. 3, 165-185.

- WHARTON, G. W., PARISH, W., JOHNSTON, D. E. (1968). Observations on the fine structure of the cuticle of the spiny rat mite, <u>Laelaps echidnins</u> (Acari - Memostigmeta). Acarologia 10, 206-214.
- WIGGLESWORTH, V. B. (1933). The physiology of the cuticle and of ecdysis in <u>Rhodnius prolixum</u> (Triatomidae, Heniptera); with special reference to the function of the cenceytes and of the dermal glands. Quart. J. micr. Sci. <u>76</u>, 269-318.
- WIGGLESWORTH, V. B. (1942). Some notes on the integument of insects in relation to the entry of contact insecticides. Bull. ent. Res. <u>33</u>, 205-218.
- WIGGLESWORTH, V. B. (1945). Transpir ation through the cuticle of insects. J. exp. Biol. 21, 97-114.
- WIGGLESWORTH, V. B. (1948). The insect cuticle. Biol. Rev. 23, 408-451.
- WIGGLESWORTH, V. B. (1957). The physiology of insect cuticle. A. Rev. Ent. 2, 37-54.
- WIGGLESWORTH, V. B. (1965). The principles of insect physiology. Nethuen, London.
- WIGGLESWORTH, V. B. (1970). Structural lipids in the insect cuticle and the function of the cencoytes. Tissue and Cell <u>2</u>, 155-179.
- WIGGLESWORTH, V. B. (1975s). Distribution of lipid in the lamellate endocuticle of <u>Rhodnius</u> prolixus (Hemipters). J. Cell Sci. 19, 439-457.
- WIGGLESWORTH, V. B. (1975b). Incorporation of lipid into the epicuticle of <u>Rhodnius</u> (Hemipters). J. Cell Sci. <u>19</u>, 459-485.
- WIGGLESWORTH, V. B. (1976). The distribution of lipid in the cuticle of <u>Rhodnius</u>. In: The Insect Integument. (ed. H. R. Hepburn) pp. 89-106. Elsevier, Amsterdam.

ZACHARUK, R. Y. (1972). Fine structure of the cuticle, epidermis, and fat body of larval Elateridae (Coleoptera) and changes associated with molting, Can. J. Zool. <u>50</u>, 1463-1487.

ZWICKY, K., WIGCLESWORTH, V. B. (1956). The course of oxygen consumption during the woulting cycle of <u>Rhodnius prolixus</u> Stål (Hemipters). Proc. R. ent. Soc. Lond. (A) <u>31</u>, 153-160.

Appendix.

 Kraba - Ringer solution (Ca⁺⁺ omitted).

 (Intermediate solutions)

 1.
 0.90% NaCl (0.154M)

 2.
 1.15% KCl (0.154M)

 3.
 2.11% KH2PO4, (0.154M)

 4.
 3.82% MSSO4.7M20 (0.154M)

Intermediate solutions mixed as follows:-

L	100	parts
2.	4	parts
3.	1	part
4.	1	part